

A study of posture shifts during sleep and the effects of supine-avoidance therapy in patients with supine-predominant obstructive sleep apnea

Bу

Matthew Mahdi Rahimi

Bachelor of Medical Science, 2015

Bachelor of Science (Honours), 2016

Thesis Submitted to Flinders University for the degree of

Doctor of Philosophy

College of Medicine and Public Health 01 December 2021

Page 1 of 169

Table of Contents

Table of conte	nts3
LIST OF TABLES	57
LIST OF FIGURI	S 8
PUBLICATIONS	9
DECLARATION	10
ACKNOWLEDG	EMENTS11
GLOSSARY OF	ABREVIATIONS 13
Thesis Summ	ary15
Thesis Aims	
Thesis Outli	ne20
CHAPTER 1.	A review of supine dependent obstructive sleep apnea and an emerging new
	A review of supine dependent obstructive sleep apnea and an emerging new positional therapy22
generation of	
generation of 1.1 Introduc	positional therapy22
generation of 1.1 Introduc 1.2 Obstruct	positional therapy
generation of 1.1 Introduc 1.2 Obstruct 1.3 Literatur	positional therapy
generation of 1.1 Introduc 1.2 Obstruct 1.3 Literatur 1.4 OSA pat	positional therapy 22 tion 23 tive sleep apnea 26 e selections methods 30
generation of 1.1 Introduc 1.2 Obstruct 1.3 Literatur 1.4 OSA pat 1.5 Current	positional therapy 22 tion 23 tive sleep apnea 26 e selections methods 30 nophysiology 31
generation of 1.1 Introduc 1.2 Obstruct 1.3 Literatur 1.4 OSA pat 1.5 Current 1.5.1	positional therapy 22 tion 23 vive sleep apnea 26 e selections methods 30 nophysiology 31 treatments 33
generation of 1.1 Introduc 1.2 Obstruct 1.3 Literatur 1.4 OSA pat 1.5 Current 1.5.1 1.6 CPAP alt	positional therapy 22 tion 23 tive sleep apnea 26 e selections methods 30 nophysiology 31 treatments 33 Continuous positive airway pressure (CPAP) 33

1.6.3	Sleep posture monitoring4	3
1.6.4	Sleep posture - supine-dependent OSA (SDOSA)4	4
1.7 SDOSA	pathophysiology4	5
1.8 Discom	fort based and position modifying treatments4	8
1.8.1	Positional therapy and the new generation of supine-avoidance alarm devices5	0
1.9 Habitual snorers without OSA55		
1.10 Concl	usion5	8
CHAPTER 2.	The clinical effectiveness of simple supine-avoidance versus continuous positive	5
airway press	sure for the treatment of supine-dependent obstructive sleep apnea : The SUPA	
randomised	controlled trial	0
2.1 Introdu	ıction6	3
2.1.1	Supine-predominant OSA6	4
2.2 Metho	d6	6
2.2.1	Study Design	6
2.2.2	Patients and procedures6	7
2.2.3	Study Measurements	0
2.2.4	Statistical Analysis	2
2.3 Results		4
2.4 Discuss	ion8	5
2.5 Conclu	2.5 Conclusion	
2.6 Acknowledgment92		2
2.7 Authors contributions		3

		·
impact of a	supine-avoidance alarm	95
3.1 Introd	uction	
3.2 Metho	ods	
3.2.1	Inclusion and exclusion criteria	
3.2.2	Supine-avoidance treatment	
3.2.3	Protocol	
3.2.4	Measures	
3.2.5	Polysomnography analysis	
3.2.6	Posture analysis	
3.2.7	Statistical Analysis	
3.3 Result	S	
3.4 Discus	sion	
CHAPTER 4.	A comparison of snoring changes with a supine-avoidance alar	m device
compared t	o constant positive airway pressure treatment in patients with sup	ine-predominant
OSA	126	
4.1 Introd	uction	
4.2 Metho	od	
4.2.1	Supine-avoidance treatment	
4.2.2	Sleep and snoring assessments	
4.2.3	Statistical analysis	
4.3 Result	S	
4 4 Discus	sion	

CHAPTER 3. A study of physiological activation responses to posture shifts in sleep and the

CHAPTER 5.	Thesis conclusions146
References	

LIST OF TABLES

Table 2.1 Patient Characteristic. 77
Table 2.2 Device data at baseline (inactive supine-alarm) and from 6-8 weeks of active supine-avoidance versus
CPAP treatment
Table 2.3 Sleep study parameters at baseline and after 6-8 weeks of each treatment. 84
Table 3.1 Basic characteristics of the study population. 110
Table 3.2 The number of PSG device (Embletta) recored posture shifts ≥ 5 -sec.112
<i>Table 3.3</i> The total number of \geq 5-sec posture shifts and alarm events recorded by the supine-avoidance alarm
device
Table 3.4 Distribution of posture shifts as a function of sleep stage immediately prior to the posture shift119
Table 4.1 Baseline characteristics of the study population. 136
Table 4.2 Snoring frequency (snores/hour) of snoring by treatment group and snoring intensity separated by
sleep posture and whole group. Values are median [interquartile range], $N=56$. * indicates p<0.05 vs
baseline, ^ p<0.05 vs CPAP
Table 4.3 Snoring frequency at baseline and on supine-avoidance compared to CPAP treatment in patients with
both supine-predominant OSA and supine-predominant snoring

LIST OF FIGURES

<i>Figure 1.1 Obstructive sleep apnea - cycle of events contributing to frequent apnea/hypopnea during sleep32</i>
Figure 1.2 Two variants of vibration based supine-avoidance alarm devices. BuzzPOD (Gorman Pty Ltd) worn
in the chest (left). Nightshift (Advanced Brain Monitoring, Inc.) worn on the neck (right)
<i>Figure 2.1 Supine-avoidance device (Buzzpod) worn on the chest over the sternum.</i> 70
Figure 2.2 SUPA trial flow of patients throughout the study
Figure 2.3 Epworth sleepiness scale (ESS) changes from baseline over the course of supine-avoidance versus
CPAP treatment
Figure 2.4. Supine-avoidance versus CPAP device usage over the course of the trial
Figure 3.1 Supine-avoidance device (left) showing externalised vibration motor, and a subject wearing the
device (right)
Figure 3.2 Box and whisker plots of the number, duration and total cumulative time spent following each type of
posture shift over the course of the study
Figure 3.3 Kaplan-Meier plots showing the proportion of patients remaining in the supine posture as a function
of time in the 10 min following \geq 5 sec device recorded shifts to the supine posture (left) and \geq 5 sec PSG
recorded shifts to the supine posture (right)
<i>Figure 3.4</i> Kaplan-Meier plots showing the proportion of PSG device recorded posture shifts \geq 5 sec followed
by wake in the 10 minutes after a shift from non-supine to the supine posture from prior sleep (left) versus wake
(right) during the baseline, supine-avoidance and CPAP PSG nights
Figure 4.1 Average hours of use per night

PUBLICATIONS

Conference abstracts

Rahimi M, Heath M, O'Grady A, Mercer J, Antic N, McEvoy RD, Vakulin A, Hnin K, Tickner D, Quinn S, Barnes M, Catcheside P. A preliminary analysis of brief arousal and awakening responses associated with posture shifts in sleep, and the impact of supine-avoidance alarm treatment. Journal of Sleep Research 25(Suppl. 2): 237, 2016.

Rahimi M., Antic N, McEvoy RD, Barnes M, Quinn S, Mercer J, Vakulin A, Heath M, O'Grady A, Tickner D, Catcheside PG. The comparative effectiveness of a simple alarm-based supine-avoidance device versus usual care with continuous positive airway pressure for treating patients with supine-predominant obstructive sleep apnea. World Sleep 2019 Congress, Vancouver Canada.

DECLARATION

I certify that this thesis:

Does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university; and to the best of my knowledge and belief, does not contain any material previously published or written by another person except where due reference is made in the text.

Matthew Rahimi

01/12/2021

ACKNOWLEDGEMENTS

Firstly, I would like to thank my supervisors Professor Peter Catcheside and Associate Professor Andrew Vakulin. Peter has been a tremendous role model for me both as a scientist and as an outstanding character throughout this journey and I will always be grateful for his advice and his insight. My appreciation also goes to Andrew and his continuous support as a great mentor.

Also, many thanks to Dasha Loutchkina, Alison Pinczel, Denzil Paul, Dr Dimitar Sajkov, Rohit Philip, Dr Jason D'Costa, Dr Jeff Bowden, Dr Anand Rose, Dr Karen Latimer, A/Prof Brian Smith, Dr Mark Jurisevic, Dr Andrew Thornton, Dr Aeneas Yeo, Dr Parmjit Singh, Dr Zafar Usmani, Dr Vinod Aiyappan for study, technical and/or clinical support. I am grateful to all the participants and laboratory staff at the Repatriation General Hospital whom without their help the SUPA trial and the works of this thesis would not have been possible. I would like to specifically thank the excellent researchers and research assistants at the Adelaide Institute for Sleep Health whom over the years I learned a great deal from them. In particular I would like to thank Professor Doug McEvoy, Dr Jeremy Mercer, the late Prof Nick Antic my good friend Dr David Stevens, Mandy O'Grady, Alison Pinczel, Hiro Tojo, Melanie Health and Alison Teare.

To my parents: your sacrifices made it possible for me to work toward my goals. Your words and virtues will always be a part of me and I am eternally grateful for your prayers.

To my Wife, Lisa, your love and support kept me going. I would have never been where I am today without you. I love you.

GLOSSARY OF ABREVIATIONS

3% ODI	3% oxygen desaturation indices
AASM	The American Academy of Sleep Medicine
АНІ	Apnea hypopnea index (events hr-1 sleep)
AQoL-8D	Assessment of Quality of Life
BMI	Body Mass Index
СРАР	Continuous Positive Airway Pressure
ECG	Electrocardiogram
EDS	Excessive Daytime Sleepiness
EEG	Electroencephalogram
EMG	Electromyography
ESS	Epworth Sleepiness Scale
HR	Heart Rate
HR ISI	Heart Rate Insomnia Severity Index
ISI	Insomnia Severity Index
ISI N1	Insomnia Severity Index Non-REM Stage 1 Sleep
ISI N1 N2	Insomnia Severity Index Non-REM Stage 1 Sleep Non-REM Stage 2 Sleep
ISI N1 N2 N3	Insomnia Severity Index Non-REM Stage 1 Sleep Non-REM Stage 2 Sleep Non-REM Stage 3 Sleep
ISI N1 N2 N3 NREM	Insomnia Severity Index Non-REM Stage 1 Sleep Non-REM Stage 2 Sleep Non-REM Stage 3 Sleep Non-Rapid Eye Movement
ISI N1 N2 N3 NREM OSA	Insomnia Severity Index Non-REM Stage 1 Sleep Non-REM Stage 2 Sleep Non-REM Stage 3 Sleep Non-Rapid Eye Movement Obstructive Sleep Apnea
ISI N1 N2 N3 NREM OSA PaCO2	Insomnia Severity Index Non-REM Stage 1 Sleep Non-REM Stage 2 Sleep Non-REM Stage 3 Sleep Non-Rapid Eye Movement Obstructive Sleep Apnea Partial Pressure of Carbon Dioxide

REM	Rapid Eye Movement
SaO ₂	Oxygen Saturation
SD	Standard Deviation
SDOSA	Supine-Dependent Obstructive Sleep Apnea
SSS	Snoring Severity Score
TST	Total Sleep Time
WASO	Wake After Sleep Onset
EOG	Electrooculography

Thesis Summary

In the first experimental study reported in Chapter 2, results from the SUpineavoidance for Positional sleep Apnea (SUPA) trial are reported. This study was designed to establish if supine-avoidance is non-inferior to CPAP in reducing sleepiness and to compare treatment compliance between supine-avoidance versus CPAP in patients with obstructive sleep apnoea (OSA) only during supine sleep. 66 patients with supine dependent obstructive sleep apnea (SDOSA) and Epworth scale (ESS) ≥8 completed baseline measurements, sleepiness including questionnaires, inactive supine-avoidance for 1 week for supine-time measurements and in-home full sleep study. Participants were then randomised to receive active supine-avoidance or CPAP treatment, followed by a cross-over to the remaining treatment after 6-8 weeks. Repeat questionnaires, sleep studies and treatment compliance measurements were collected after 6-8 weeks on each treatment. Noninferiority was assessed from the change in ESS with supine-avoidance compared to CPAP using a conservative pre-specified non-inferiority margin of 1.5. Average nightly use over 6-8 weeks of each treatment was also examined and compared between treatments. At baseline, participants had 47.6±20.4% supine sleep and ESS 10.3±3.9. At the end of 6-8 weeks of treatment, the reduction from baseline in ESS with supineavoidance (mean [95%CI] -1.9 [-2.9 to -0.9]) was no worse than CPAP (-2.3 [-3.3 to -1.3], difference -0.2 [-1.3 to 0.9]). Average treatment usage was substantially higher with supine-avoidance compared to CPAP (5.6 [5.0 to 6.3] versus 3.9 [95%CI 3.2 to 4.6] h/night, p< 0.001). These findings support that alarm-based supine-avoidance is non-inferior to CPAP at reducing sleepiness and achieves superior treatment adherence in patients with SDOSA.

In the second study (Chapter 3), physiological responses to posture shifts with and without a supine-avoidance alarm device were examined as a marker of sleep disturbance. From the data collected on polysomnography (PSG) nights of participants in the SUPA trial, posture shifts were systematically investigated to see whether they are preceded by electroencephalogram (EEG) detectable arousal/awakenings and if so, how long it takes for sleep to resume. Key secondary aims were to characterize the temporal relationship between EEG responses to posture shifts during sleep with vs without a supine-avoidance alarm, and to investigate the impact of supine alarms on overall sleep quality.

Posture shift data recorded via PSG and supine-avoidance devices were extracted via custom algorithms to detect posture shifts lasting ≥5-seconds. EEG data 30 secs before and after each posture shift were then examined to assess wake versus sleep and arousals prior to and the time taken to achieve sleep following PSG recorded posture-shifts.

The study population was comprised of 56 participants (61% males) who completed baseline and both treatment PSG nights in the main SUPA trial. These participants exhibited a mean±SD age 53.6±11.2 years, BMI 31.4±8.1 kg/m², AHI 17.2±6.7 /hr (supine AHI=40.3±40.7 /h). From a total of 376 (baseline) vs 249 (supine-avoidance) posture shifts (median [IQR] 6[4-9] vs 6[3-9] per patient), the majority [347/376 vs

222/249; 91[75-100] vs 100[84-100]%) were associated with prior wake or an arousal. The hazard of remaining in the supine posture following posture shifts to supine was around 4-fold lower with active versus inactive supine-avoidance treatment (hazard ratio 4.3 [95% CI 3.7 to 4.9]), with a marked reduction in post-shift supine time (median [IQR] 0.2 [0.1 to 0.3] minutes versus 18.5 [11.0 to 39.6] minutes. Sleep resumed quickly following posture shifts to supine from prior sleep, with no difference with versus without the supine-alarm (0.5 [0.1 to 1.4] versus 0.5 [0.2 to 1.3] minutes). Sleep onset time was considerably longer following shifts to the supine posture from prior wake, but were also not different with versus without the active supine-avoidance alarm (4.9 [1.3 to 8.5] versus 6.1 [2 to 13.9] minutes).

Most posture shifts occur shortly after an arousal or during established wake supporting the presence of higher centre involvement in posture shifts. Thus, supine-alarm device events occurring after supine shifts from sleep typically coincide with a brief posture-shift related return to wake, followed by a rapid return to sleep minimally impacted by the alarm *per se*. This study presents strong evidence to support that supine-avoidance alarms effectively limit supine sleep time with no discernible further sleep disturbance beyond that associated with posture shifts themselves.

In the third study (Chapter 4), snoring was objectively examined in patients with supine-predominant sleep apnea. This study was a sub-study of the SUPA trial that investigated the effect of posture on snoring severity and the effect of a supineavoidance alarm device on snoring. This is the first study to present objective data on snoring frequency and to compare the effects on snoring of a supine-avoidance alarm device compared to CPAP. Snoring frequency was analysed objectively in both supine sleep and non-spine sleep during baseline and supine-avoidance treatment PSG nights in the full study population. Analysis was also performed on a subgroup of participants who mainly snored on their back (defined as supine snorer on the basis of snoring frequency at least 2x higher in supine compared to non-supine sleep). The frequency of discrete snoring events \geq 50 dBA was significantly reduced by CPAP compared to baseline and supine-avoidance therapy in the full study population (p<0.001, baseline= 48.9 [16.7 to 188.7] snores/h, supine-avoidance= 36.8 [6.3 to 233.7] snores/h, CPAP= 4.2 [2.1 to 29.5] snores/h). Both CPAP and supine-avoidance therapy reduced snoring frequency in supine sleep (Baseline= 66.3 [22.9 to 186.7], supine-avoidance= 13.2 [0.7 to 67.1], CPAP= 8.5 [1.6 to 44.3] snores/h). In supine snorers both supine-avoidance therapy and CPAP produced a statistically significant reductions in total frequency of snoring.

This study showed a simple supine-avoidance alarm device in patients with supine dependent sleep apnea does not substantially reduce snoring frequency. In contrast, CPAP was substantially more effective in reducing snoring although even with CPAP some residual snoring remained.

Thesis Aims

The primary objectives of the work presented in this thesis were as follows:

The following aims were addressed in the work presented in Chapter 2;

Aim 1: To investigate whether supine-avoidance therapy via a vibratory alarm device is non-inferior compared to CPAP therapy in reducing sleepiness in patients with supine-predominant OSA.

Aim 2: To examine the efficacy of supine-avoidance therapy via a vibratory alarm device compared to CPAP therapy in reducing the apnea/hypopnea index in patients with supine-predominant OSA.

Aim 3: To examine participant adherence to treatment with supine-avoidance therapy via a vibratory alarm device versus CPAP therapy over 6-8 weeks of each therapy in patients with supine-predominant OSA.

The work presented in Chapter 3 addressed the following further aims;

Aim 4: To investigate the temporal relationship between EEG and posture shifts during sleep.

Aim 5: To examine the effect of a vibratory supine-avoidance alarm device on the time taken to return to sleep following attempts to shift to the supine posture during the sleep period.

The work presented in Chapter 4 was designed to address the final thesis aims; Aim 6: To examine the effect of a vibratory supine-avoidance alarm device on snoring event frequency during sleep in patients with supine-predominant OSA. Aim 7: To examine the effect of a vibratory supine-avoidance alarm device on snoring frequency during sleep in the sub-group of patients with both supine-predominant OSA and supine-predominant snoring.

Thesis Outline

Chapter 1 Describes the current literature surrounding supine-avoidance therapy and the main existing gaps in knowledge for which the remaining thesis work was designed to help address.

Chapter 2 presents results of a non-inferiority randomised controlled clinical crossover trial (SUPA trial) to compare the effectiveness of 6-8 weeks of supine-avoidance therapy via a simple vibratory alarm device compared to 6-8 weeks of CPAP therapy in reducing sleepiness in patients with supine-predominant OSA.

Chapter 3 presents a detailed analysis of posture shifts in patients with supinepredominant obstructive sleep apnea specifically seeking to examine if posture-shifts occur primarily from wakefulness or following a brief arousal from sleep and if a supineavoidance alarm device negatively impacts the time-taken to resume sleep following overnight supine-posture shifts.

Chapter 4 presents an in-depth analysis of supine-avoidance therapy via a vibratory alarm device compared to CPAP therapy on snoring frequency in supine-predominant

obstructive sleep apnea patients and a sub-group of patients who also demonstrated supine-predominant snoring.

Chapter 5 summarises the overall contribution of this thesis work to advancing the field of sleep medicine and research specific to supine-avoidance therapy for supine-predominant sleep problems, outlines the key limitations of the work and highlights future directions for research in this area.

CHAPTER 1. A review of supine dependent obstructive sleep apnea and an emerging new generation of positional therapy

Key points

Question: This Chapter explores the current evidence from the existing literature on supine-avoidance therapy and CPAP therapy in OSA patients and highlights key gaps in knowledge with a particular focus on patients with supine-dependent OSA.

Key Findings: There is limited available evidence to support the treatment efficacy, compliance and the degree of symptom reduction with supine-avoidance therapy in patients with supine-dependent OSA.

Meaning: Objective data remain needed to determine if supine-avoidance therapy can be used as a viable treatment option for a sub-group of OSA patients who are diagnosed with SDOSA in clinical settings.

1.1 Introduction

Around ~30%-65% of obstructive sleep apnea (OSA) patients can be classified as supine dependent (SDOSA), which is traditionally defined based on an apnoea/hypopnea index (AHI) during supine sleep at least twice that of non-supine AHI (Mador et al., 2005a; Marshall et al., 2007; Senaratna et al., 2017; Adams et al., 2017), indicating a substantially higher proportion of breathing interruptions in supine compared to non-supine sleep. For this group, one of the simplest forms of treatment could potentially be to simply avoid supine sleep.

Traditionally, supine-avoidance has been achieved using tennis ball treatment (TBT) (Oksenberg et al., 2006; Bignold et al., 2009), or a similar approach, which involves strapping an object (e.g. tennis ball or another hard object or airbags or similar approaches) to the back to discourage supine sleep through discomfort or a physical impediment to lying supine. TBT is reasonably effective at reducing supine sleep and overall AHI and is low cost (Skinner et al., 2008). However, the inherent discomfort of this approach is problematic for long-term use and around 90% of SDOSA patients report abandoning TBT due to discomfort after only 3 months (Bignold et al., 2009).

Patient acceptance and compliance with continuous positive airway pressure (CPAP), the main first-line treatment recommended for OSA, is also problematic. Although not specifically designed to target SDOSA patients, CPAP therapy is very effective at reducing AHI and improving sleep quality and is most often recommended when patients report excessive sleepiness during the day and exhibit a moderate to severe AHI (Kribbs et al., 1993; Weaver et al., 2007). However, similar to TBT, treatment compliance remains a significant issue as around 50% of patients recommended CPAP find the mask and pressure uncomfortable and fail to use it for at least 4 hours per night (the cut-off generally accepted as minimum usage for therapeutic benefit), or do not even commence CPAP due to reasons such as high cost of acquiring CPAP, or psychological factors around claustrophobia (Weaver and Grunstein, 2008).

A recent new generation of "smart" supine-avoidance devices show significant promise for effective treatment for patients with SDOSA. These small devices are typically strapped around the chest (Bignold et al., 2011; Dieltjens et al., 2015) or sometimes the neck (Levendowski et al., 2014) and are designed to vibrate each time the user moves on their back during sleep to alert and discourage the wearer from remaining supine. Randomised control trials have shown that these devices can be effective in reducing supine sleep and supine AHI (Van Maanen and De Vries, 2014; Bignold et al., 2011; Berry et al., 2019). Nevertheless, these trials have only been conducted in small samples with little focus on treatment compliance. Consequently compliance relative to CPAP, the current recommended best-practice treatment for OSA, remains unclear. Given very poor acceptance and usage of traditional discomfort-based supine-avoidance treatment, both adequate effectiveness and treatment acceptance and usage compared to established mainstream treatments remain critically important to demonstrate before new generation supine-avoidance treatments can be recommended as a viable treatment option in patients with SDOSA. Multiple criteria have been used to define and categorise the SDOSA, and this warrants consideration in the context of supine-avoidance treatments given potential impacts on treatment selection, effectiveness and ultimately patient acceptance and adherence to treatment for improving sleep and health outcomes.

A further relevant area for which available literature is currently lacking is the current understanding of the physiological sequence of events associated with posture shifts during sleep. Knowledge regarding these events is likely to importantly guide how positional therapy may impact the physiological propensity for upper airway obstruction and the severity of obstructive breathing events and arousals in SDOSA patients. A leading hypothesis, that either wake or an arousal may be prerequisites for higher cortical co-ordination of body movements necessary to achieve a posture shift during the sleep period remains to be systematically examined. Thus, objective data and a systematic analysis of the electroencephalographic (EEG) sequence of events around posture shifts remain warranted in the context of supine-avoidance treatments. Other physiological changes likely to be temporally associated with posture shifts, such as heart rate and respiratory rate changes also remain to be investigated. Establishing the temporal relationships between physiological changes and posture shifts remains important for evaluating the impacts of a vibration alarm, a key characteristic of new generation supine-avoidance devices that aim to discourage supine sleep but with minimal sleep disruption.

This chapter outlines the range of criteria used to define SDOSA for guiding supineavoidance treatments, and the available evidence regarding the effectiveness of SDOSA treatments. A further focus is on trials comparing traditional supine-avoidance therapy with CPAP and new generation positional therapy devices, and on studies examining the timing and nature of physiological changes around the time of posture shifts during sleep.

1.2 Obstructive sleep apnea

Obstructive sleep apnea (OSA) is the most common form of sleep disordered breathing and is characterised by recurrent narrowing (hypopnea) or complete closure (apnea) of the upper airway during sleep. OSA results in intermittent oxygen desaturations, along with frequent arousals that fragment sleep. OSA severity is conventionally determined by the apnea/hypopnea index (AHI), which reflects the number of apneas and/or hypopneas that occur during each hour of sleep. Using current scoring criteria, OSA severity is often categorised as mild with an AHI of 5-15, moderate with AHI 15-30, and severe with AHI greater than 30 (Berry et al., 2012).

Based on an AHI>5, the prevalence of symptomatic OSA ranges from 9% to 38% and is generally higher in men (Senaratna et al., 2017). With an AHI >15, the estimated prevalence of symptomatic OSA in the general population ranges from 6% to 17%, but reaches as high as 49% in older age groups and in obese men and women (Senaratna et al., 2017). Other community cohort studies (Heinzer et al., 2015; Benjafield et al., 2019) suggest that asymptomatic OSA affects roughly 50% of the adult population and that symptomatic OSA impacts around 1 billion people globally (Benjafield et al., 2019; Benjafield et al., 2018).

In a randomised controlled CPAP withdrawal cross-over study OSA related frequent awakenings and hypoxemia have been found to increase stress hormones such as cortisol and to increase blood glucose and fatty acid levels (Chopra et al., 2017) that may ultimately contribute to long-term autonomic, inflammatory, hemodynamic and metabolic disturbances. In cross-sectional studies, OSA has consistently been found to be associated with cardiometabolic problems such as hypertension, stroke, heart failure, cardiac arrhythmias and type 2 diabetes mellitus; and ultimately increased allcause mortality (Guilleminault et al., 2005; Stein and Pu, 2012; Parish and Somers, 2004; Marshall et al., 2008; Heinzer et al., 2015; Dredla and Castillo, 2019). OSA is also independently associated with increased incidence of mental health problems, including depression and anxiety (Ejaz et al., 2011).

Furthermore, and likely as a result of chronic sleep fragmentation, OSA often results in excessive daytime sleepiness (EDS) associated with increased risk of motor vehicle and workplace accidents, memory loss and reduced workplace productivity (Maspero et al., 2015; White and Younes, 2012). In the United States, undiagnosed OSA has been estimated to contribute an annual economic burden to the community of \$149.6 billion a year (Sulivan, 2015). In Australia, the estimated cost of untreated OSA to the economy through direct and indirect costs such as through accidents and lost workplace productivity is around \$21 billion a year (Economics, 2017). Thus, it is clearly important that OSA is not only identified and comprehensively diagnosed, but effectively treated via appropriately targeted approaches that can ultimately be demonstrated to effectively and cost-effectively improve individual and public health. Whilst there is growing research into different endotypes or phenotypes of OSA and their variable aetiology, the predominant cause of OSA clearly remains failure of upper airway musculature to successfully maintain upper airway patency through a combination of anatomical and non-anatomical compromise (Eckert, 2016; Edwards et al., 2017; Edwards et al., 2014). OSA patients show some morphological abnormalities in their craniofacial and pharyngeal measurements, including increased soft palate length and thickness and reduced cross-sectional area compared to healthy adults (Johal et al., 2007).

Upper airway collapse is likely to be further exacerbated by obesity and aging effects. For example, increased fat mass around the neck may also contribute to reduced pharyngeal diameter and mass loading effects on pharyngeal muscles to promote an increased propensity for upper airway collapse (Ahbab et al., 2013; Kim et al., 2014). Abdominal obesity may also promote cranial displacement of the diaphragm and increased collapsibility of the pharyngeal airway through a combination of reduced lung volume and axial tension on the airway (Stadler et al., 2009). When a person transitions from wake to sleep, there is also an abrupt reduction in neural drive to the pharyngeal muscles in the upper airway (Wilkinson et al., 2008), likely directly causing increased upper airway resistance and reduced patency in patients with OSA vulnerable to airway collapse (Stadler et al., 2010). OSA patients also show evidence of poorly coordinated activation of the upper airway muscles and unfavourable biomechanical properties, leaving them susceptible to frequent upper airway collapse (Bilston and Gandevia, 2014).

Recent trends in sleep medicine have been pushing for more targeted therapies for OSA towards a more 'precision medicine' approach (Eckert, 2016; Deacon et al., 2016; Zinchuk et al., 2016). This recognizes that various phenotypes/endotypes exist in OSA, where targeted therapies delivered based on underlying causal mechanisms and/or types of clinical presentation are more likely to achieve maximal treatment efficacy, adherence and thus overall effectiveness. This strategy is likely to achieve improved outcomes for patients compared to the current approach of one-size-fits-all treatment, predominantly via CPAP, which is typically followed by further trial and error treatments which ultimately fail in many patients who do not accept or effectively use CPAP and other alternatives before being lost to follow-up with uncertain outcomes (Eckert, 2016).

The remainder of this narrative review firstly evaluates the existing literature on OSA pathophysiology and current treatment options for SDOSA patients and supine-predominant snorers, and then focusses on sleep position effects in SDOSA patients. Finally, available literature regarding a new generation of non-discomfort-based supine-avoidance devices are examined.

1.3 Literature selections methods

PubMed and Google Scholar were searched for relevant peer reviewed journal articles on OSA, its treatments and physiology. Search strategies aimed firstly to identify relevant publications on obstructive sleep apnea, CPAP effectiveness, patient adherence to treatment and secondly to identify articles regarding the role of physiological responses to position and posture shifts in sleep and how an alarm device may impact OSA severity. These included searches for randomized control trials comparing CPAP to positional therapy, as well as literature surrounding new generation supine-avoidance devices. Given an apparent gap in knowledge regarding the impact of supine-avoidance on OSA and very few relevant articles on the physiology of posture shifts during sleep, searches included articles from 1980 to 2019 selected on the basis that supine dependent OSA was first formally recognised in the mid 1980's (Cartwright et al., 1985). The reference section for each article found was also searched to identify further relevant articles. Keywords used in online searches along with Boolean terms where appropriate included sleep disordered breathing, obstructive sleep apnea, BuzzPod, Nightshift, Night Balance, positional therapy, CPAP therapy, compliance, adherence, supine dependant OSA and supineavoidance.

1.4 OSA pathophysiology

Traditionally, OSA was considered to reflect anatomical abnormalities, such as a small oropharynx and an abnormally collapsible upper airway that is mostly compensated by increased pharyngeal dilator muscle activity when awake but inadequately compensated during sleep (Figure 1.1) (White and Younes, 2012). However, in line with current concepts of differing endotypes/phenotypes, the aetiology and mechanisms underpinning OSA remain poorly understood and likely differ between individual patients.

The pathophysiology of OSA is clearly more complex than simple anatomical deficits and/or neuromuscular factors alone, with growing evidence to support that more complex and inter-dependent respiratory control, low arousal threshold and ventilatory control instability also play important roles (White and Younes, 2012). During sleep, reduced neural drive to the pharyngeal muscles in the upper airway associated with loss of wakefulness inputs to postural and respiratory muscle tone likely contribute to reduced patency (Wilkinson et al., 2008; Kubin, 2016), along with a greater reliance on autonomic respiratory control compared to wake.

It is increasingly evident that OSA is a heterogenous disorder with multiple physiological mechanisms influencing its severity (Figure 1.1).



Figure 1.1 Obstructive sleep apnea - cycle of events contributing to frequent apnea/hypopnea during sleep.

1.5 Current treatments

Therapeutic and cost-effective treatments for OSA already exist, but patient acceptance, adherence and access to treatment are all major ongoing problems. A large community survey of 10,000 people aged between 18 to 64 suggested around 6% of the Australian population seek medical help for problem snoring and/or OSA, with 2% reporting subsequent treatment (Marshall et al., 2007); implying that many who seek specialist help do not receive or accept treatment. The Sleep Health Foundation (2017) in Australia reported that 17% of people fall asleep on the job and 29% report making errors at work due to sleepiness (Adams et al., 2017). In combination with estimates that over 80% of adults with OSA remain undiagnosed (Young et al., 1997), this suggests that the burden of treatable OSA and problematic snoring in the community is very large and under-managed. There are several treatment strategies available for patients discussed below.

1.5.1 Continuous positive airway pressure (CPAP)

Continuous positive airway pressure (CPAP) therapy is the current gold-standard clinical treatment for OSA. CPAP uses an air pump and well-sealed facial mask to create a pneumatic splint to maintain airway patency. This approach is highly effective in normalising breathing in sleep and improving sleep quality and daytime symptoms (Giles et al., 2006). However, CPAP is cumbersome and many patients find it uncomfortable. Side effects are also common with patients reporting claustrophobia, mask leaks and dry eyes and throat (Abdelghani et al., 2009; Bahammam et al., 2015;

Krieger, 1992). Individuals also often remove their CPAP during the night and fail to put it back on (Fan et al., 2019) thus leaving them untreated, potentially especially during the latter part of the night where REM-related OSA and impacts may be most severe (Appleton et al., 2016). Adequate adherence to CPAP therapy, somewhat arbitrarily defined as at least 4 hours use per night for more than 4 nights a week, is highly effective in reducing AHI and improving sleep quality and daytime sleepiness in compliant users (Dempsey et al., 2010; Greneche et al., 2013; Mcevoy et al., 2016). However, CPAP adherence is often poor, and several large randomised controlled trials demonstrate that average CPAP usage is generally below the 4 hour cut-off widely accepted as minimum level of compliance needed for effective long-term treatment (Mcevoy et al., 2016; Fan et al., 2019). CPAP only controls obstructed breathing while an appropriately fitted mask remains in place, and daytime deficits return with a single night of missed treatment (Kribbs et al., 1993). Around 50% of patients recommended CPAP reject it at the outset (Lee et al., 2017), with a further 12-25% of patients who commence treatment abandoning it within 3 years (Engleman and Wild, 2003a). Even patients who continue on CPAP long-term have been shown to demonstrate poor treatment adherence (Weaver and Grunstein, 2008).

On the other hand, CPAP adherence in more recent very large real-world CPAP manufacturer data sets appears to be much higher. Cistulli et al (2019) found in a database of over 2.6 million CPAP users that 75% of patients used CPAP \geq 4 h/night on \geq 70% of nights in the first 90 days of use and with average daily use per night round 5.5 hours. Similarly, in a larger and longer-term analysis in over 4 million CPAP users, Drager et al (2021) reported that over 80% of patients met conventional criteria of

acceptable CPAP adherence at 3 months, with 75% remaining adherent at 1-year. These apparently discrepant findings with earlier studies may well reflect improved patient engagement tools and changes in testing and insurance criteria for the provision of CPAP, particularly in the USA (Drager et al., 2021). However, higher adherence rates should also be expected in patients who have already committed to using CPAP compared to earlier studies that predominantly investigated CPAP uptake and usage in clinic patients recommended therapy for OSA.

Behavioural interventions may improve CPAP uptake and adherence to some degree, but ultimately the intrusive nature of mask treatments will likely continue to limit widespread acceptance and usage. CPAP is particularly poorly tolerated in milder and asymptomatic cases of OSA (Engleman and Wild, 2003a) and may not be considered a practical primary treatment for mild and relatively asymptomatic OSA or simple snoring. In addition, the burden of wearing a CPAP mask all night for supine dependent OSA may be considered excessive to physicians and patients alike. Although CPAP generally improves bed partner sleep quality (Parish and Lyng, 2003), treatment related disturbance of bed partner sleep can also be a barrier to treatment adherence in some patients (Weaver et al., 2003).

CPAP can also pose a significant financial burden and barrier for some patients, depending on country-specific reimbursements offered through health systems and individual insurance status (Hillman et al., 2006; Weaver, 2006). In Japan, all CPAP therapy costs are paid for by the health system, which has led to high rates of patients who initiate CPAP treatment but low long-term adherence (Tanahashi et al., 2012). In

Israel, CPAP acceptance is significantly greater (~50%) when patients receive financial incentives amongst low socioeconomic echelons (Tarasiuk et al., 2012). A recent study in India found that 60% of patients choose not to buy CPAP citing financial constraints (Goyal et al., 2017). CPAP non-buyers may choose to rent a CPAP machine, but are more likely to have significantly worse compliance when compared to CPAP buyers (Goyal et al., 2017). For the many patients who refuse or choose not to start CPAP therapy, there may be several factors besides financial cost involved in their decision. These could include a family member or acquaintance reporting prior difficulties with CPAP, patients lack of education or perceived need for treatment, living/not living with a partner or simply a fear of or dislike due to the intrusive nature or perceived burden of long-term use of CPAP (Bahammam et al., 2015; Engleman and Wild, 2003b; Goyal et al., 2017; Lee et al., 2017; Weaver, 2006).

As effective as CPAP therapy may be in normalising OSA and symptoms, significant adherence issues and its economic burden has prompted calls for non-CPAP treatment options that are sufficiently efficacious, but more acceptable to patients. Thus, superior treatment outcomes appear likely to be possible through more effectively targeted treatments more specifically targeted to OSA phenotypes or endotypes, and that are more acceptable to patients for more regularly and effective long-term use.
1.6 CPAP alternatives

The main non-CPAP treatments currently in use for OSA include weight-loss (through diet or bariatric surgery), mandibular advancement splint (MAS) devices designed to pull the lower jaw forward to open and stiffen the airway, and upper airway surgery such as modified uvulopalatopharyngoplasty, coblation channelling of the tongue, hyoid suspension and maxilla-mandibular advancement (Kneisley, 1998; Marshall et al., 2007; Basyuni et al., 2018; Carvalho et al., 2012).

1.6.1 Weight loss

There is a strong association between OSA and obesity so weight loss if often strongly recommended as a desirable treatment option to treat and reduce OSA symptoms (Dempsey et al., 2010; Kim et al., 2014; Stadler et al., 2010; Stadler et al., 2009; Supriyatno et al., 2010; Roche et al., 2020). A 10% increase in excess body weight relative to normal body weight is associated with a 32% increase in AHI (Peppard et al., 2000) and a 6-fold increase in the odds of developing moderate to severe OSA (Peppard et al., 2000). Weight loss of around 20 kilograms can successfully reduce AHI by almost 20 events per hour (Smoots and Pisani, 2004). Additionally, leptin, a peptide produced by fatty tissue that regulates homeostatic inflammation, metabolism and sympathetic nerve activity, has been found to influence upper airway patency and resistance along with sleep structure (Imayama and Prasad, 2017). Leptin appears to play a complex role in OSA and its severity as it can be both a bi-product of OSA (via leptin resistance) and a marker of cardiovascular diseases and obesity (Martin et al.,

2008; Wannamethee et al., 2007). Patients diagnosed with OSA show a higher level of leptin that is positively correlated with more severe OSA. Imayama et al. (2017) investigated leptin levels in 32 obese men compared to 32 matched obese men without OSA. Leptin levels were higher in obese men with OSA, suggesting a potential role of OSA in leptin resistance. Leptin resistance in OSA patients could hinder weight loss efforts and may contribute to worsening of OSA.

Although weight loss is strongly recommended for obese or overweight patients with OSA, effective and sustained weight loss is generally very difficult to achieve and maintain for most patients. Patients who choose to undergo weight loss surgery, such as Roux-en-Y gastric bypass (RYGB), often show effective treatment outcomes in the short term, but may return to preoperative BMI in the longer-term (Hawkins et al., 2017).

1.6.2 Mandibular advancement splint (MAS) device and surgical treatments for OSA

The effectiveness of MAS and surgery is more variable and difficult to predict compared to CPAP (Mackay et al., 2013). MAS devices greatly differ in types and design with many variants available on the market. MAS devices generally increase the diameter of the airway through soft tissue displacement by protruding the mandible forward which in turn changes the jaw and tongue position to help open and stiffen the upper airway (Casey, 2015). MAS efficacy is variable, and although AHI reductions can range from around 30% to 70% in different groups of patients (Basyuni et al.,

2018), and some individuals show good treatment outcomes, in others MAS can be totally ineffectual.

Most MAS studies have focussed on AHI as the primary outcome. However, AHI shows only weak relationships with symptoms for which patients seek treatment so may not be the best outcome measurement from which to gauge effective treatment outcomes for patients. Side effects of MAS devices such as hypersalivation, dry mouth, dental pain, gingival irritation, myofascial pain and temporomandibular joint (TMJ) discomfort may also impact both treatment effectiveness and tolerance (Basyuni et al., 2018).

Surgical treatments for OSA aim to increase airway patency by targeting specific sites of obstruction, which often differ between individual patients (Carvalho et al., 2012). There are several surgical methods available and each have different success rates and variable outcomes depending on the selected patients and procedures. For example, tongue surgeries including posterior midline glossectomy or radiofrequency ablation show success rates ranging from around 25% to 80% (Kezirian and Goldberg, 2006; Nelson and Barrera, 2007). Maxillomandibular advancement (MMA) has perhaps the highest rate of success (80%-90%) with approximately equivalent effectiveness compared to nasal CPAP (Caples et al., 2010; Holty and Guilleminault, 2010). In addition to the general risks involved with any surgery, sleep apnea increases surgical anaesthesia risks (Carvalho et al., 2012; Casey and Teodorescu, 2015; Franklin et al., 2009). Patients often require endotracheal intubation given that anaesthesia poses an airway risk in OSA patients, which in turn may lead to a longer

hospital stay. Other surgical risks include excessive bleeding, infection, deep vein thrombosis, additional breathing problems and allergic reaction to anaesthesia (Carvalho et al., 2012).

One other important factor to consider, especially in clinical settings, is adherence to therapy. There are strong interactions between perceived symptoms, adherence to treatment and treatment outcomes in patients with OSA. For example, asymptomatic patients are less likely to perceive treatment benefit and thus to adhere to treatment compared to patients who perceive symptom improvements and are therefore more likely to be treatment adherent (Weaver and Grunstein, 2008). This phenomenon is not restricted to OSA and a World Health Organization report concluded that 50% of patients living with chronic conditions do not adhere to treatment (De Geest and Sabate, 2003), quite similar to adherence rates observed in patients who frequently discontinue CPAP treatment after 6 months (Kribbs et al., 1993). Adherence to therapy is thus an important focus of new and emerging treatments for OSA where treatments typically aim to maximise effectiveness through both high efficacy and adherence. Hypoglossal nerve stimulation (HNS) is a relatively new treatment where upper airway neuromuscular stimulation is used to reduce airway collapsibility. HNS has been shown to substantially reduce AHI (Eastwood et al., 2020; Friedman et al., 2016; Steffen et al., 2018; Strollo et al., 2014) and to be well tolerated by most patients, but is generally less efficacious compared to CPAP (Hofauer et al., 2019; Kent et al., 2016; Kezirian et al., 2014). Given that HNS devices are implanted adherence rates are typically high. However, adverse events such as pain, tongue abrasions or internal/external device malfunction can also limit treatment adherence in same cases (Kompelli et al., 2019).

Given that OSA is a heterogeneous disorder with multiple anatomical and nonanatomical factors contributing to aetiology and severity, pharmacological agents targeted to a range of relevant physiological factors have the potential to usefully treat OSA. For example, sedative drugs such as Zopiclone, have the potential to help treat patients with a low arousal threshold who arouse to relatively mild obstruction that could potentially be compensated through increased breathing effort and upper airway muscle recruitment with a higher arousal threshold. Zopiclone has been studied in short-term trials typically involving only one night, so long-term adherence data remain lacking (Carberry et al., 2017; Eckert et al., 2011; Roth et al., 2005). In the largest and longest duration randomised double-blind parallel group study to date of Zopiclone for the treatment of OSA, Carter and colleagues (2018) found treatment adherence over one month to be high in both treatment and placebo groups (mean±SD 93.4%±11.0% and 93.7%±9.7%, respectively). However, only one third of patients reported to be willing to continue to take medication if it were available beyond the trial.

There is also evidence to support that ineffective upper airway dilator muscle recruitment can be improved with targeted pharmacotherapy using serotonergic, noradrenergic and antimuscarinic medications such as atomoxetine and oxybutynin. These drugs have also been mostly studied in single night randomised controlled conditions using relatively small sample sizes and no adherence data reported (Taranto-Montemurro et al., 2019).

A hypersensitive ventilatory control system (high loop gain) has also been recognised to likely play an important role in promoting unstable breathing in many individuals with OSA. Consequently, medications such Acetazolamide, or O₂ therapy, targeted to reduce loop gain could potentially be useful in some patients. Acetazolamide has been shown to be effective in reducing AHI in OSA (Schmickl et al., 2020), but as with other recent pharmacotherapies, long-term adherence data are relatively lacking as most studies have only examined the effectiveness of Acetazolamide at high doses given only for one night (Edwards et al., 2012; Javaheri, 2006). In the few longer-term studies of Acetazolamide, many patients reported discontinuation of treatment due to adverse drug effects (Debacker et al., 1995; Ulrich et al., 2015; Sakamoto et al., 1995; Latshang et al., 2012).

Non-CPAP treatments typically show poorer and less predictable efficacy outcomes and thus more variable success rates compared to CPAP and, much like CPAP, this includes a component of variable adherence. More accessible and acceptable treatments than those currently available are clearly still needed to more efficiently and effectively treat patients with OSA long-term. One particularly promising form of treatment, and the primary focus of the work described in this thesis, is supine avoidance in patients with supine-predominant respiratory disturbances during sleep.

1.6.3 Sleep posture monitoring

There are several proposed methods for measuring and monitoring posture in sleep, although there is no general consensus on how posture measurements should be conducted or classified for the assessment of posture-related sleep disorders (Dingli et al., 2003; Liu et al., 2013; Jeng et al., 2021; Clemente et al., 2020; Ren et al., 2016; Ferrer-Lluis et al., 2021). Traditionally, in clinical settings, posture is assessed using a combination of periodic manual assessment of video monitoring during a polysomnography study, along with position sensor signal acquisition across the night. With the advancement of PSG recording devices, most systems use tilt-sensing devices or an accelerometer to assess the angle of the position sensor device, usually worn on the chest. PSG systems then typically apply custom algorithms to classify posture, often into 5 basic body positions including left and right lateral, supine, front or prone and upright positions (Dingli et al., 2003). Depending on sensor type and position, this approach is not necessarily entirely reliable and warrants validation against video confirmed posture monitoring of the PSG. The accuracy of PSG posture recordings with ambulatory devices, particularly in home set-up sleep studies remains largely unknown. Other methods such as a dense pressure-sensitive bedsheet (Liu et al., 2013), a wrist worn sensor (Jeng et al., 2021), bed frame mounted sensors (Clemente et al., 2020), a depth sensing camera (Ren et al., 2016), and smart phone tri-axial accelerometer (Ferrer-Lluis et al., 2021) have also to some extent been tested and used for posture monitoring. In general, these appear to achieve greater than 80% accuracy, but have only been tested in relatively limited settings with variable posture classification approaches and small sample sizes. For the most part posture

monitoring has almost exclusively been directed at body position, so very little remains known regarding the effects of head and neck position relative to body position.

1.6.4 Sleep posture - supine-dependent OSA (SDOSA)

Multiple definitions have been used to describe supine-dependent OSA (SDOSA) such as supine isolated (Sutherland et al., 2015) or supine-predominant OSA (Kim et al., 2016). However, there are currently no universal clinical criteria for the diagnosis of SDOSA. Cartwright's classic definition of supine-predominant OSA was defined on the basis of a supine AHI at least twice that of the non-supine AHI (Cartwright et al., 1985). This definition was subsequently adapted by Mador et al. (2005) to include a normal non-supine AHI (<5 /h based on AHI scoring criteria at that time) determined from a minimum of 30 minutes of sleep in each posture and a >50% reduction in AHI between supine and non-supine postures. This definition has advantages over Cartwright's definition for use in clinical practice since, by definition, patients who meet these criteria have clinically significant OSA that would be expected to normalise with successful supine-avoidance treatment given the combination of sufficient accumulated supine-sleep time and elevated AHI contribute to the OSA classification itself. However, further definitions have also been advanced to include somewhat different criteria, such as the Amsterdam positional OSA classification (APOC)(Frank et al., 2015) and a 4:1 supine: non-supine AHI ratio to define supine-isolated OSA which appears to be a more stable measure of supine dependence night to night (Joosten et al., 2014a).

Irrespective of classification methodology, it has also been known for some time that the upper airway is more susceptible to collapse when a person is sleeping supine compared to non-supine positions (Benumof, 2016; Cartwright et al., 1991; Cartwright et al., 1985). Although mechanisms and the degree of influence sleep position on the upper airway remain incompletely understood, gravitational effects on the tongue, upper airway and potentially caudal tracheal traction and lung volume dependent effects may all play a role. Numerous studies show that supine AHI is significantly higher compared to non-supine AHI in a substantial proportion of OSA patients (Berry et al., 2019; Bignold et al., 2009; Bignold et al., 2011; Cartwright et al., 1991; Frank et al., 2015; Ha et al., 2014; Joosten et al., 2014a; Kavey et al., 1985).

Depending on how supine-predominant OSA is defined, between 35%-60% of OSA patients suffer from SDOSA (Mador et al., 2005a; Adams et al., 2017). For sufferers of SDOSA, supine sleep avoidance could be a very simple, low cost and potentially more effective treatment than other forms of OSA treatment (Engleman and Wild, 2003a; Ha et al., 2014). Patients in this group have been shown to be thinner, younger, with lower neck circumference and lower AHI compared to non-positional OSA (Mador et al., 2005a; Oksenberg and Gadoth, 2019).

1.7 SDOSA pathophysiology

Several observations have been made involving anatomical and physiological factors that may play a role in explaining strong associations between posture and OSA. Gravity has major effects on upper airway tissues such as the tongue and soft palate (Walsh et al., 2008) and on diaphragm position and lung volume that may further influence the propensity for upper airway collapse via airway stiffening effects of tension on the upper airway from the trachea and other intra-thoracic structures (Van De Graaff, 1991). OSA is virtually abolished in the microgravity of space and most patients show more severe OSA when supine (Prisk, 1998).

Airway lumen shape, size and tendency to collapse are determined by factors such as bony structures (e.g. maxilla, mandible) and size of soft tissue (e.g. tongue) that may ultimately define airway behaviour during sleep when neuromuscular compensation responses may be diminished. Pharyngeal area has been shown to decrease in OSA patients when changing posture from sitting to supine, followed by an increase in uvular width and narrowing of the retroglossal airway when patients adopt the supine posture (Walsh et al., 2008). Most of these anatomical changes have been investigated in patients who were awake (Miyamoto et al., 1997; Walsh et al., 2008). In sleeping patients, particularly when supine, lung volume decreases are associated with increased upper airway resistance, although this might just be a normal response to supine sleep (Dempsey et al., 2010; Joosten et al., 2015; Margues et al., 2017). Upper airway muscle function may also play an important role in determining the severity of OSA. During non-rapid eye movement sleep, genioglossal muscle responsiveness to negative pressure pulses has been shown to decrease in the supine compared to lateral decubitus position (Malhotra et al., 2004). Genioglossal EMG activity is also significantly reduced by head/neck rotation in both upright and supine postures suggestive of simultaneous changes in airway stiffness, patency and muscle activity (Otsuka et al., 2000).

However, in most OSA patients AHI is generally higher in the supine position than when sleeping laterally, especially in mild OSA cases, even when the formal definition of SDOSA is not necessarily met (Cartwright et al., 1991; Kavey et al., 1985; Mador et al., 2005b). In a small study of 13 male patients with OSA examined during an all-night polysomnography while inclined at an angle of 60 degrees versus lying supine, AHI was decreased (inclined 19.6 \pm 6.9/h versus supine 48.9 \pm 5.4/h, p<0.0005) and both mean and minimum arterial oxyhaemoglobin saturation (SaO₂) were increased (nREM; mean inclined 92.1 \pm 0.5% versus supine 90.6 \pm 0.8%, p<0.005; minimum inclined 80.8 \pm 2.1% versus supine 64.8 \pm 3.2%, p<0.005) (Mcevoy et al., 1986). Similar findings were observed in another study where 8 patients with severe OSA who slept in supine, lateral and 30-degree elevated positions, showed less collapsible airway pressures and a 50% reduction in upper airway opening pressures in lateral and elevated compared to supine positions (Neill et al., 1997).

In addition to an increased AHI, respiratory events have been shown to be more severe and more prolonged in the supine compared to lateral postures (Oksenberg and Silverberg, 1998; Poyares et al., 2002; Edwards et al., 2017). AHI also appears to be less dependent on position in more severe OSA. However, recent evidence suggests that almost one third of patients with severe OSA may also have supinedependent OSA and hence could also benefit at least to some extent from supineavoidance therapy (Oksenberg and Gadoth, 2019).

1.8 Discomfort based and position modifying treatments

CPAP, which is designed to treat OSA regardless of underlying causes, is inherently somewhat intrusive and uncomfortable, and despite high efficacy only around 50% of patients recommended CPAP ultimately accept and adequately use it long-term (Catcheside, 2010). MAS devices are also somewhat intrusive and uncomfortable. Consequently, even conventional treatments for OSA are associated with a degree of discomfort.

Supine sleep avoidance has been traditionally achieved by strapping a bulky object, such as a tennis ball, to the back to counteract supine posture shifts through discomfort (Skinner et al., 2008). When appropriately applied to patients with supine-predominant OSA, the 'tennis ball treatment' (TBT) is more targeted, simpler and cheaper compared to CPAP therapy and can be effective in lowering AHI by decreased time spent in supine sleep (Skinner et al., 2008; Jokic et al., 1999). However, in a survey follow-up of patients recommended TBT in clinical practice, almost 90% of patients reported abandoning TBT within a median time of 1 month due primarily to discomfort (Bignold et al., 2009). Thus, discomfort with this traditional approach is clearly highly problematic to the extent that TBT cannot be recommended as a viable long-term treatment for positional OSA. Other supine-avoidance approaches, such as position restricting devices (e.g. supine prevention vest), anti-snoring pillows (Zuberi et al., 2004) and auditory alarms (Cartwright et al., 1991) have been reported, but data to support clinical effectiveness and patient comfort, acceptance and compliance with long-term use remain remarkably limited. An ongoing lack of evidence to support

effective treatment outcomes with these forms of treatment remains problematic for guiding evidence-based treatment recommendations, particularly given strong evidence to support that comfort is fundamentally important to achieve effective treatments that patients will also accept and use long-term.

Although some patients claim to learn to avoid the supine posture during sleep (Bignold et al., 2009; Oksenberg and Silverberg, 1998), there are very few studies to support long-term supine-avoidance conditioning effects with positional therapies. Two studies (Chaudhary et al., 1986; Cartwright et al., 1991) have shown that there might a proportion of patients who learn to avoid supine sleep with or without ongoing use of a supine-avoidance device. There is also some anecdotal evidence that patients claim to have learnt to avoid supine sleep after using a supine-avoidance device (Bignold et al., 2009; Van Maanen et al., 2013). However, there is also some evidence from clinical experience that patients revert to sleeping supine (Oksenberg et al., 2006). Evidence to support the hypothesis that patients may become effectively trained or conditioned to avoid supine sleep remains lacking.

Considering the significant role of posture in influencing OSA propensity and severity and the large number of patients with position-dependent OSA, it is surprising that fundamental knowledge gaps remain regarding posture shifts during sleep and potential positional effects on clinical outcomes. For example, agreed standards for defining positional OSA remain unclear. Perhaps more importantly, knowledge regarding the physiological responses before and after posture shifts during sleep remain remarkably lacking. There is also a dearth of literature comparing usual care to supine-avoidance therapy, especially when it comes to vibration based alarm devices and their effect on OSA patients or snorers. A better understanding of the pathophysiology of position dependent OSA and treatment outcomes is needed to better manage positional OSA and its symptoms.

1.8.1 Positional therapy and the new generation of supine-avoidance alarm devices

Low adherence to traditional supine-avoidance strategies has prompted exploration of new non-discomfort based supine alarm devices designed to discourage supine sleep with minimal discomfort. These new generation of devices include a sleep position trainer (SPT) (Berry et al., 2019; Laub et al., 2017) and other devices worn on the chest (Bignold et al., 2011) or on the neck (Levendowski et al., 2014; Van Maanen and De Vries, 2014) via a strap, and use a position triggered alarm to vibrate shortly after patients shift to the supine position (Figure 1.2). A vibration stimulus is generally used in preference to an auditory alarm more likely to disturb bed-partner sleep. The concept of using smart devices to avoid supine sleep is relatively new, such that clinical trials investigating the efficacy of these devices and whether they can be used as a successful treatment option for positional OSA remain limited.

Six small randomised clinical trials to date have examined several of these new smart devices and in general showed positive results regarding reduced supine sleep time (Bignold et al., 2011; Levendowski et al., 2014; Van Maanen et al., 2013; Dieltjens et al., 2015; Mok et al., 2020; Hidalgo Armas et al., 2019). In a small proof of concept

study, Bignold (2011) showed that a new generation of supine-avoidance alarm device is effective in reducing supine sleep time from (mean ± SEM) 19.3% ± 4.3% to 0.4% \pm 0.3% of sleep (p < 0.001) and in reducing AHI (25.0 \pm 1.7 to 13.7 \pm 1.1 events/h, p = 0.030). Dieltjens (2015) compared the addition of a vibration alarm SPT device to MAS therapy in 20 SDOSA patients and found that individually both MAS and SPT were effective in reducing AHI from 20.9 (95% CI 17.0 to 34.0) /h at baseline to 11.0 (6.6 to 14.0) /h and to 12.8 (3.9; 17.9) /h with MAS or SPT, respectively. However, a combination of both therapies had an even greater effect on reducing AHI to 5.5 (3.4 to 7.2) /h. Levendowski (2014) assessed the effectiveness of a neck worn vibration alarm device in reducing supine sleep in 30 SDOSA patients and found that 83% of participants showed an overall reduction of at least 50% in supine sleep time when using the SPT device with supine sleep reduced from 44% of total sleep time at baseline to just 2% at the end of 4 weeks of treatment. In a recent randomised controlled trial, Mok (2020) compared the effectiveness of CPAP vs positional therapy using the NightShift device in patients with excessive daytime sleepiness (ESS>10) to test the hypothesis that supine-avoidance is non-inferior to CPAP at reducing ESS. Although ESS and AHI were reduced with both treatments, the difference in ESS between treatments exceeded the author's pre-specified non-inferiority margin of 1.5, so non-inferiority could not be concluded (Mok et al. 2020).

Hidalgo Armas et al. (2019) also found improvements in AHI and oxygen saturation with 4 weeks of supine-avoidance therapy using a vibrating device; from a median (IQR) baseline AHI of 30.7 (23.2–38.2) to 21.5 (12.4–24.3) /h at the end of treatment. Time spent supine was also significantly reduced from (mean \pm SD) 51.5 \pm 14.8% to

 $25.2 \pm 21.0\%$ (Hidalgo Armas et al., 2019). However, further studies to determine if time spent supine can be reduced further by more effective devices and if time spent supine is reduced during sleep versus wake remain warranted.

In a larger randomized controlled trial, Laub (2017) and colleagues randomised 101 SDOSA patients to an SPT group (n=52) or a control group (i.e. no treatment, n=49) and found AHI after 6 months of SPT reduced from (mean \pm SD) 35 \pm 18 /h to 10 \pm 9 /h (p < 0.001) compared to 18 \pm 10/h (p > 0.05) in the control group. In addition, in the SPT group the average supine sleep time decreased from 47 \pm 22% to 17 \pm 18% (p < 0.001) compared to 39 \pm 21% (p > 0.05) after 2 months in the control group (Laub et al., 2017).

A Cochrane Review in 2019 ultimately included only three studies that compared positional therapy with CPAP and concluded that ESS was not different between the two forms of treatment. However, CPAP produced a consistently greater AHI reduction, but with lower adherence to treatment compared to positional therapy (Srijithesh et al., 2019).



Figure 1.2 Two variants of vibration based supine-avoidance alarm devices. BuzzPOD (Gorman Pty Ltd) worn in the chest (left). Nightshift (Advanced Brain Monitoring, Inc.) worn on the neck (right).

Other vibration-based alarm devices have also been shown to be effective in reducing supine sleep time and overall AHI (Eijsvogel et al., 2015; Van Maanen et al., 2013). Assessment of a neck worn alarm device showed a reduction to <3% supine sleep (compared to ~47% in baseline) and a 50% reduction in overall AHI after several weeks of using the device including 1 night of polysomnography (Levendowski et al., 2014). Two other devices with similar functionality, but strapped around the chest rather than the neck, also showed a significant decrease in supine sleep time (Eijsvogel et al., 2015; Van Maanen and De Vries, 2014; Van Maanen et al., 2013).

Although a range of new generation supine-avoidance devices show promising results for reducing supine sleep time and thus overall AHI in patients with supinepredominant OSA (Laub et al., 2017; Eijsvogel et al., 2015), results to support reduced daytime sleepiness or improved quality of life remain lacking and somewhat unconvincing. A fundamental issue of existing studies is the use of variable definitions of SDOSA, which likely influences the target populations selected for study and may influence the apparent clinical effects of these devices (Dieltjens et al., 2015; Eijsvogel et al., 2015; Laub et al., 2017; Van Maanen and De Vries, 2014; Van Maanen et al., 2013). In addition, whether standard CPAP therapy is still the better option for SDOSA patients compared to positional therapy remains unknown due to an ongoing lack of objective data regarding how new positional therapies compare to CPAP in terms of efficacy and compliance and thus overall effectiveness to reduce AHI, as well as cost, symptom relief and overall quality of life (Oksenberg et al., 2019).

Ongoing uncertainty with all supine-avoidance treatments, including the new vibrationbased alarm devices, arises from a considerable lack of clinical evidence regarding the impact of supine-avoidance therapies on sleep quality. In the context of new generation supine-avoidance devices, it remains unclear if vibration alarms triggered by supine posture shifts might be disruptive to sleep through exaggerated or potentially more prolonged awakenings. Alternatively, posture-shifts themselves may well require either wakefulness or brief arousal to engage higher motor control functions potentially needed for co-ordinated muscle activity necessary to shift body position, in which case an additional alarm may have relatively little impact on the process of returning to sleep following body movements. However, remarkably little is known regarding the sequence of events preceding posture shifts in sleep, with no objective evidence available in the literature to guide if a brief return to consciousness is required to engage muscle activity to change posture, followed by a return to sleep with little or no conscious perception or lasting memory of the posture change itself (Tassi and Muzet, 2001).

Combination therapies that combine positional and another form of therapy to improve overall treatment efficacy are a further option that has received relatively little attention to date in the literature. At the time of writing, only one study appears to have investigated the effects of combining a supine-vibration alarm with a mandibular advanced device, where combination therapy led to higher therapeutic efficacy when compared to either therapy used alone (Dieltjens et al., 2015). An earlier study to investigate the combination of a tongue retaining device (TRD) and an auditory posture alarm to avoid supine position was reported by Cartwright et al. (1991). This study found that 11/15 (73%) participants achieved an AHI<5 events /h (i.e. a successful treatment response) with combined TRD and posture alarm treatment compared to fewer achieving treatment success with supine avoidance auditory alarm (60%) or TRD (53%) treatment alone. In addition, weight loss may also be usefully combined with positional therapy and obese patients with OSA have been shown to transition into mainly supine-predominant OSA following significant weight loss (Joosten et al., 2017). Thus, supine avoidance therapy combined with weight loss could potentially be curative for some patients.

1.9 Habitual snorers without OSA

Snoring and OSA are part of a continuum of obstructed breathing in sleep. Around 95% of OSA patients snore heavily most or every night (Viner et al., 1991) and there

are strong correlations between snoring frequency and intensity and respiratory and arousal disturbances in sleep (Wilson et al., 1999). Habitual loud snoring is at least as common as OSA, affecting somewhere between 15-54% of all middle-aged adults (Enright et al., 1996; Young et al., 1993).

However, there are no standardised or gold-standard methods for quantifying and classifying snoring, and assessment methods often vary widely between studies. Individual snorer and bed-partner perceptions regarding problem snoring are also highly variable (Hoffstein et al., 1996). The American Academy of Sleep Medicine (AASM) recommends three measures of snoring: an over-head acoustic sensor, a piezoelectric sensor and a nasal cannula pressure transducer (Arnardottir et al., 2016). Comparisons against a microphone attached to the chest show sensitivity and positive predictive values of scoring snore events of 0.78 and 0.98 respectively for an overhead acoustic sensor, 0.55 and 0.67 for a nasal cannula and 0.78 and 0.92 for a piezoelectric sensor (Arnardottir et al., 2016). However, there is also no unified agreement regarding optimal placement of microphones in either laboratory of home PSG settings (Arnardottir et al., 2016; Dafna et al., 2014). This lack of standardisation and agreed measurement guidelines for snoring assessment remains problematic for both research and clinical decision making relevant to snoring.

Snoring is often trivialised, but negative effects on snorer and bed partner sleep quality can be substantial. Snoring frequently disrupts patient *and* bed partner sleep contributing to insomnia, psychosocial problems and reduced quality of life (Mcardle et al., 2001). Snoring has also been linked to marital problems (Jones and Swift, 2000),

occasional acts of violence including homicide (Pelausa and Tarshis, 1989) and is a leading reason patients seek specialist ear, nose and throat (ENT), and sleep treatments (Mcardle et al., 2001; Marshall et al., 2007). Increased cardiovascular risk from snoring alone is also plausible, via vibration-induced vascular injury and effects of chronic cardiac exposure to large negative intrathoracic pressures thought to cause right ventricular hypertrophy, as has been demonstrated in rats (Salejee et al., 1993). Further animal data show that snoring vibrations are transmitted to the carotid arteries (Amatoury et al., 2006), and that vibration *in vitro* induces arterial endothelial cell inflammation and damage (Curry et al., 2002; Puig et al., 2005). A dose-dependent relationship has also been shown between snoring severity and carotid, but not femoral artery, atherosclerosis (Lee et al., 2008). Available data are therefore consistent with the hypotheses that chronic obstructed breathing in sleep increases adverse cardiovascular outcome risk and that snoring vibrations may contribute to carotid arterial damage, plaque rupture and stroke risk.

Like OSA, snoring propensity and severity can be strongly influenced by posture. Limited epidemiological data are available to guide how many people suffer from only supine-predominant snoring without OSA. However, the American Sleep Association estimates that 90 million Americans suffer from chronic snoring, with around half that number potentially also exhibiting OSA. Snoring, besides being an early sign of possible future OSA, is disturbing to bed partners and may reduce their sleep quality.

Supine-avoidance has been found to be effective in reducing habitual snoring (Cartwright et al., 1985; Jones and Swift, 2000), but there is currently a lack of

evidence to support the effectiveness of new generation supine-avoidance devices for reducing habitual snoring.

1.10 Conclusion

Despite the growing body of knowledge in sleep medicine, further research is needed to better understand the role of supine sleep posture in the pathophysiology of OSA and the efficacy, adherence to treatment and overall effectiveness of non-discomfort supine-avoidance treatment compared to CPAP. There is also limited knowledge regarding the sequence of events associated with posture shifts in sleep and the impact of supine-avoidance alarms on sleep in the post-posture shift period. A better understanding of physiological events accompanying posture shifts, and the importance of supine posture in OSA is needed to more effectively guide improvements of supine-avoidance targeted therapies for OSA.

The work described in this thesis aimed to fill some of the key existing gaps in knowledge regarding the treatment of supine-predominant OSA by means of a simple supine-avoidance alarm device compared head-to-head with CPAP. In the first experimental chapter (Chapter 2) the primary aim was to investigate the efficacy of a vibration-based supine-avoidance alarm device against usual care (CPAP) in reducing sleepiness, the primary symptom complaint of clinical concern in patients with supine-predominant OSA. A secondary aim was to examine the comparative treatment efficacy, adherence and overall effectiveness over 6-8 weeks of treatment with each

therapy. Chapter 3 reports on a detailed analysis of temporal relationships between posture shifts during the sleep period and key relevant physiological responses including wake versus sleep and arousals immediately prior to posture shifts and the time taken for sleep to resume following posture shifts with versus without supineavoidance alarms. In the final experimental chapter (Chapter 4), the effects of a supine-avoidance alarm device compared to CPAP on snoring in supine-predominant OSA patients was evaluated. The final chapter (Chapter 5) briefly summarises the overall thesis findings and makes recommendations for future research towards further improvements in supine-avoidance therapies for supine-predominant OSA and snoring. CHAPTER 2. The clinical effectiveness of simple supineavoidance versus continuous positive airway pressure for the treatment of supine-dependent obstructive sleep apnea : The SUPA randomised controlled trial

Key points

Question: This study sought to test the hypothesis that supine-avoidance via a vibration alarm device is non-inferior to CPAP therapy in reducing daytime sleepiness in patients with supine-predominant OSA and achieves greater treatment adherence over 6-8 weeks of each therapy.

Findings: In 66 patients with supine-predominant OSA supine-avoidance therapy was non-inferior to CPAP in reducing excessive daytime sleepiness and achieved superior treatment adherence.

Meaning: Supine-avoidance is an effective treatment option for patients with supinepredominant OSA

Abstract

Around 50% of OSA patients show supine-predominant OSA, where simply avoiding supine sleep could potentially be a highly effective therapy. However, traditional methods of supine-avoidance, such as strapping a tennis ball to the back, are inherently uncomfortable and most patients self-report abandoning discomfort-based treatments within around one month. Modern non-discomfort supine-alarm devices are now widely available, but evidence remains lacking to support sufficient effectiveness and adherence to treatment to reduce excessive daytime sleepiness compared to standard treatment with continuous positive airway pressure (CPAP).

This study was designed to test the hypothesis that in patients with supinepredominant OSA, alarm-based supine-avoidance is non-inferior to CPAP for reducing daytime sleepiness and achieves superior treatment adherence compared to CPAP over 6-8 weeks follow-up.

66 patients with supine-predominant OSA and Epworth sleepiness scale (ESS) score ≥8 completed baseline measurements including questionnaires, inactive supineavoidance for 1 week for pre-treatment supine-time measurements, and in-home full polysomnographic evaluation of sleep. Patients were then randomised to active supine-avoidance or CPAP treatment, followed by cross-over to the remaining treatment after 6-8 weeks. Repeat questionnaires, sleep studies and treatment adherence measurements were collected after 6-8 weeks on each treatment. Noninferiority was assessed from the change in ESS with supine-avoidance compared to CPAP using a pre-specified non-inferiority margin of 1.5. Average nightly use over all nights, and treatment efficacy in repeat home sleep studies were also compared between treatments.

Patients were predominantly males (62%) aged (mean±SD) 52.8±11.9 years, body mass index 31.9 ± 7.8 kg/m², with a baseline sleep study showing supine-predominant OSA (total, supine and non-supine AHI 18.0±8.9 and 40.2±38.8, 5.0±3.2 /h respectively), a moderate degree of daytime sleepiness (ESS 10.3±3.9) and substantial supine sleep time (47.6±20.4%). After 6-8 weeks of treatment a reduction in ESS with supine-avoidance (mean [95%CI] -1.9 [-2.9 to -0.9]) was no worse than with CPAP (-2.3 [-3.3 to -1.3], difference -0.2 [-1.3 to 0.9]). Average treatment usage was substantially higher with supine-avoidance compared to CPAP (5.6 [5.0 to 6.3] versus 3.9 [95%CI 3.2 to 4.6] h/night, p< 0.001), although efficacy and effectiveness for reducing AHI was lower.

This study supports that alarm-based supine-avoidance is non-inferior to CPAP for reducing sleepiness and achieves superior treatment adherence over 6-8 weeks of treatment in patients with supine-predominant OSA.

2.1 Introduction

Obstructive sleep apnoea (OSA) is the most common form of sleep disordered breathing and affects around 25% of men and 10% of women (Heinzer et al., 2015; Young et al., 2002; Adams et al., 2016). Largely untreated OSA is estimated to cost around \$150 billion in direct and indirect financial costs in the USA alone (Economics, 2017; Hillman et al., 2006; Sulivan, 2015). OSA is characterised by frequent upper airway collapse, with periods of absent or severely reduced ventilation and recurrent brief arousals that cause severe sleep disruption and marked repetitive oxidative and cardiovascular system stress. OSA is an established risk factor for excessive daytime sleepiness, cognitive impairment, motor vehicle and other accidents and cardiometabolic disorders (Tregear et al., 2009; Dredla and Castillo, 2019).

Continuous positive airway pressure (CPAP) is the first-line treatment for OSA and is highly effective in normalising breathing and oxygenation during sleep and improving sleep quality and daytime symptoms. However, CPAP usage is poor and side effects, such as claustrophobia and discomfort, as well as problems associated with mask leaks are common. Between 5% and 50% of patients recommended CPAP reject it up-front (Abdelghani et al., 2009; Bakker et al., 2019; Engleman and Wild, 2003a) and many patients who initiate CPAP treatment and continue to use it long-term show poor treatment adherence with <4 hours usage per night, the most commonly used cut-off for defining acceptable use (Engleman and Wild, 2003b). CPAP is particularly poorly tolerated in milder cases of OSA and thus CPAP is not widely used to treat mild OSA and snoring (Weaver et al., 2003). Alternative treatments include weight loss, oral

devices and surgery to help open and stiffen the airway. However, all have more limited and variable effectiveness compared to CPAP and patient acceptance and adherence are often also poor. For example, weight loss is rarely achieved and maintained long-term, and effectiveness of surgery and oral devices are variable and difficult to predict (Franklin et al., 2009). Consequently, there remains an ongoing need for simple efficacious alternative treatments that patients will accept and use longterm.

2.1.1 Supine-predominant OSA

Around 50% of patients diagnosed with OSA show at least twice as many obstructed breathing events when they sleep supine (Mador et al., 2005a; Omobomi and Quan, 2018), and around 30% exhibit OSA only when supine (Laub et al., 2015; Mador et al., 2005a). This effect is most likely dominated by gravitational effects on the tongue and pharyngeal airway, combined with tracheal traction, lung volume and fluid shift dependent effects. Thus, avoiding the supine sleep posture could potentially be a very simple but clinically useful therapeutic approach in many patients, particularly those with supine-only or supine-predominant OSA. When appropriately titrated and in the absence of significant leaks, CPAP can effectively eliminate OSA regardless of sleep posture. However, poor patient acceptance and adherence remain a significant problem with CPAP, particularly when perceived symptom benefits are low. Thus, alternative strategies with superior patient acceptance and treatment adherence, despite potentially lower efficacy in reducing the AHI compared to CPAP, remain warranted.

Various devices mimicking the traditional "tennis ball treatment" to discourage supine sleep have been shown to decrease OSA severity and to improve quality of life in patients with supine-predominant OSA (Casey, 2015; Skinner et al., 2008; Van Maanen and De Vries, 2014). However, to date there is only one recently reported head-to-head comparative effectiveness trial of CPAP versus supine-avoidance, which showed non-inferior treatment efficacy to reduce AHI and greater treatment adherence with supine-avoidance compared to CPAP, but was not designed to test for non-inferior symptom relief (Berry et al., 2019). Given very low self-reported acceptance of traditional discomfort based supine-avoidance (Bignold et al., 2009), supine-avoidance clearly requires sufficient evidence to support effective long-term treatment and use. Furthermore, given sub-optimal patient acceptance and adherence with CPAP (Abdelghani et al., 2009), and the importance of both treatment efficacy and adherence in determining treatment effectiveness for symptom relief, head-tohead studies of supine-avoidance compared to CPAP are clearly also warranted. Consequently, this study was specifically designed to directly compare sleepiness symptom relief and treatment adherence, efficacy and effectiveness (the product of efficacy and adherence) of a simple supine-avoidance alarm-based device versus usual care with CPAP for treating patients with supine-predominant OSA over 6-8 weeks of each treatment. The primary hypothesis was that supine-avoidance therapy would be non-inferior to CPAP in reducing daytime sleepiness. A key secondary hypothesis was that supine-avoidance therapy would achieve superior treatment adherence when compared to CPAP.

2.2 Method

2.2.1 Study Design

This supine-avoidance for OSA (SUPA) study was National Health and Medical Research Council (NHMRC) funded randomised controlled comparative-effectiveness cross-over study of 6-8 weeks treatment with a supine-avoidance device versus CPAP in patients with supine-predominant OSA. The study was designed as a non-inferiority trial to test the primary hypothesis that the improvement in ESS with supine-avoidance would be non-inferior to that with CPAP. ESS was chosen as the most widely used metric of daytime sleepiness, which is the most concerning clinical issue relevant to OSA. Key secondary outcomes were objective device-measured treatment adherence over 6-8 weeks of treatment, treatment efficacy to reduce the apnea-hypopnea index (AHI, the number of apnea and hypopnea events per hour of sleep) measured via repeat in-home sleep studies on each treatment, and the overall effectiveness to reduce average AHI (the product of efficacy and adherence) over the full course of each treatment.

Randomisation of treatment allocation order was performed by a local pharmacy department at Repatriation General Hospital independent of the investigators, using a minimisation program (MinimPy)(Saghaei and Saghaei, 2011) to avoid imbalance of potentially important confounders between randomised groups; including recruitment site, sex, age (<50 vs \geq 50 years), body mass index (BMI <30 vs \geq 30 kg/m²), supine apnoea hypopnoea index (AHI <35 vs \geq 35 /h), ESS score (<12 vs \geq 12). Cut-offs were based on historical clinic median data using the same eligibility criteria. Patients

commenced 6-8 weeks of treatment according to allocation order with no washout period between the two treatments. This reduced participant burden from additional visits and was considered appropriate given extended outcome measures over time up to the end of each treatment and that usual clinical practice does not delay changeover to alternative treatments.

The Adelaide Institute for Sleep Health (AISH) at Flinders University was the lead site for patient recruitment, overall trial management, data collection and analysis. Patients with supine-predominant OSA were screened and recruited from several South Australian hospitals including Repatriation General Hospital, Royal Adelaide Hospital, Ashford Hospital, Memorial Hospital, Queen Elizabeth Hospital and Flinders Medical Centre based on polysomnography (PSG) findings from each patient's initial diagnostic sleep study at the recruitment site. Patients who met the study criteria were then invited to participate in the trial following review and approval from their sleep specialist. All subsequent study procedures were conducted at AISH.

The project was approved by the Southern Adelaide Clinical Human Research Ethics Committee (HREC protocol number 431.13) and was registered on the Australian and New Zealand Clinical Trials Registry (ANZCTR registration number 12613001242718). All participants gave written informed consent.

2.2.2 Patients and procedures

Eligibility criteria were: 1) age \geq 18 and <75 years; 2) total AHI \geq 10 /h (based on American Academy of Sleep Medicine "alternate" scoring criteria (lber et al., 2007)

used across sites); 3) Epworth sleepiness scale (ESS) score \geq 8; 4) Supinepredominant OSA defined as supine AHI \geq twice non-supine AHI and non-supine AHI <10 /h; and 5) total sleep time \geq 4 hours with \geq 30 min supine and \geq 30 min non-supine sleep. Criteria 2-4 were based partly on Mador et al (Mador et al., 2005a), to ensure that only patients with supine-predominant OSA associated with at least mild sleepiness were selected where effective supine-avoidance would be expected to successfully treat their OSA. Criteria 5 was selected to ensure adequate time was available to assess supine- vs non-supine OSA severity and sufficient supine-time to warrant supine-avoidance treatment. Patients with any of the following criteria were excluded: 1) prior treatment with CPAP or supine-avoidance therapy; 2) severe sleepiness (history of falling asleep while driving or ESS \geq 16) requiring urgent treatment; 3) commercial driver; 4) co-morbidities that might preclude supineavoidance treatment (e.g. arthritis or mobility problems or a pacemaker where a chestworn device might interfere); and 5) unwilling to cease current alternative OSA therapies (e.g. mandibular advancement splint).

Posture monitoring and supine-avoidance treatment utilised a chest-worn device (BuzzPOD Model 2, Gorman ProMed Pty. Ltd, Melbourne Australia) to monitor and record body position (at 1 Hz over the full 6-8 week recording period) and discourage supine sleep via a strong vibration stimulus. The vibration stimulus was set to activate after \geq 5 consecutive seconds in the supine posture and remained continuous until either a non-supine position was registered, or 2 minutes elapsed. If supine positioning continued beyond 2 minutes a pulsed mode was activated (2-sec periods of alternating vibration off and on) for a further 3-minutes. In the event of failure to discourage

supine-positioning within 5 minutes the device alarm timed-out and de-activated to conserve battery. This device was selected based on previously demonstrated accuracy and reliability for recording body position and established effectiveness for discouraging supine sleep (Bignold et al., 2011).

Following patient consent, participants were provided with an inactivated supineavoidance alarm device for 7 days to record baseline sleep postures in the home setting. Participants then returned to the laboratory for setup of a baseline home polysomnography study for assessing pre-treatment OSA severity in the home-setting. Patients were then randomized to commence either CPAP or active supine-avoidance therapy for 6-8 weeks before crossing over to the remaining treatment for a further 6-8 weeks with a repeat home PSG within the last week on each treatment.

Patients who were allocated to CPAP were issued an auto-PAP device (AirLiquide Healthcare) to use for 3 nights before transitioning to a fixed pressure device set at the 95th centile pressure, with ongoing telephone and in-person support in the first week and as required for the remainder of CPAP treatment. Patients were instructed in the use of both the supine-avoidance device and CPAP by trained research staff and an experienced CPAP provider. Patients had follow-up appointments after 4 weeks on each treatment to download treatment usage data from each device and to help trouble-shoot any ongoing issues regarding their allocated treatment.



Figure 2.1 Supine-avoidance device (Buzzpod) worn on the chest over the sternum.

2.2.3 Study Measurements

Anthropometric measurements (age, gender, height, weight, waist circumference and BMI), questionnaires and in-home sleep studies were conducted at baseline and at the end of each treatment. Questionnaires included the ESS (Johns, 1992), general health questionnaire (GHQ) (Lundin et al., 2016), assessment of quality of life questionnaire (AQoI-8D) (Richardson et al., 2014), Pittsburgh sleep quality index (PSQI) (Mollayeva et al., 2016), functional outcome of sleep questionnaire (FOSQ), insomnia severity index (ISI) (Gagnon et al., 2013), snoring scale score (SSS) (Lim and Curry, 1999), and a simple symptoms, complaints and treatment satisfaction questionnaire.

In-home sleep studies were recorded using an Embletta MPR (REMLogic[™], Pleasanton, California, United States) home-PSG device to record EEG (C3-M2, C4-M1), left and right electro-occulograms (EOG), chin EMG (all sampled at 500 Hz),

nasal cannula pressure and thermistor (sampled at 250 Hz), thoracic and abdominal motion (sampled at 100 Hz), oximetry (sampled at 3 Hz) and body position (sampled at 20 Hz).

All PSGs were scored by a single experienced sleep technician who participated in a regular PSG scoring concordance quality assurance program and remained blind to treatment allocation. Objective CPAP compliance was calculated as the average hours of usage per night, including zero hours when not used, and the percentage of nights used for more than 4 hours. Objective supine-avoidance device usage was calculated on the same basis using device recorded position changes between manual button pushes with confirmatory sleep diary entries recording bed- and rise-times.

Treatment efficacy was assessed from total sleep AHI on each in-home sleep study treatment night. The overall effectiveness of each treatment to reduce average AHI over the full 6-8 weeks of treatment was estimated from the time-weighted product of baseline versus on-treatment supine and non-supine AHI, the estimated fraction of each night spent supine (F_{Supine}) versus non-supine (F_{Non-supine}=1-F_{Supine}; using device measured values for supine-avoidance and baseline week values for CPAP), and the average adherence (h/night) with each treatment relative to average time in bed (h/night; from the baseline week) using the following equation;

Estimated Average AHI = Treatment (AHIsupine x Fsupine + AHINon-supine x FNon-supine) x average treatment hours per night/average time in bed +

Page 71 of 169

Baseline night (AHI_{Supine} x F_{Supine} + AHI_{Non-supine} x F_{Non-supine}) x (1-average treatment hours per night/average time in bed)

This expression ranges between total AHI at baseline (re-weighted to baseline week supine vs non-supine times) without treatment, through to total AHI on treatment when average treatment hours per night equals total time in bed.

2.2.4 Statistical Analysis

The primary null hypothesis was that the difference in ESS between supine-avoidance and CPAP would be >1.5 ESS units. When designing the study it was considered that expert clinicians in the field would agree that a difference of <1.5 is unlikely to be clinically important. This non-inferiority margin is conservative and was selected before Patel et al (Patel et al., 2018) reported a minimum clinically important reduction in ESS of between 2 and 3. It was estimated that if the true difference between treatments was zero, then 58 patients would be required to reject the null hypothesis to support non-inferiority with a type I error rate of 2.5% (one-sided) in a cross-over trial design.

Statistical analysis was performed on an intention-to-treat treatment basis using IBM SPSS Statistics Version 25 and all available data. Normality was assessed using Shapiro-Wilk Tests. The primary outcome was examined by testing if the lower 95% confidence interval bound of the difference in ESS reduction with CPAP minus supine-avoidance did not cross the *a-priori* non-inferiority margin of -1.5. Secondary outcomes were examined using linear mixed effects model analysis with time and treatment as
repeated factors within-subjects using an auto-regressive co-variance structure, treatment order as a fixed effect and subjects as a random effect each with their own intercept. Data are presented as mean±SD or [95% confidence interval] unless otherwise specified.

2.3 Results

The flow of patients and trial procedures employed are shown in Figure 2.2. A total of 917 patients were assessed for eligibility across all sites. 273 met the initial screening criteria and were approached to participate in the study and further screening. Sixty-four patients declined to participate and a further 61 were excluded for other reasons, the majority due to excessive distance from the study site. Overall, 68 patients met the study criteria and 66 consented to participate. Recruitment began in January 2014 with final visits in October 2017. A total of 10 patients discontinued treatment (5 during each arm of the study) before their final visit, such that fifty-six patients completed the full study.

Figure 2.2 SUPA trial flow of patients throughout the study.



Baseline characteristics of study participants are shown in Table 2.1. Participants were predominantly middle-aged, overweight or obese males with mild-to-moderate OSA, but with severe OSA when sleeping supine, which accounted for around 50% of total sleep time, consistent with supine-predominant OSA and the study selection criteria.

Parameter	Whole group	Supine- avoidance treatment 1 st Group	CPAP treatment 1 st Group
N (% Male)	66 (62.1%)	33 (63.6%)	33 (60.6%)
Age (yr)	52.8 ± 11.9	54.1 ± 12.7	51.5 ± 11.0
Height (cm)	171.9 ± 9.1	173.2 ± 8.8	170.7 ± 9.4
Weight (kg)	93.8 ± 21.4	96.3 ± 24.4	91.4 ± 18.0
Body Mass Index (kg/m²)*	31.9 ± 7.8	32.2 ± 8.4	31.6 ± 7.2
Diagnostic sleep study			
Total sleep time (min)	357.0 ± 61.1	351.3 ± 59.7	362.7 ± 62.7
Non-supine sleep time (min)	186.2 ± 79.8	183.0 ± 79.2	189.3 ± 81.5
Supine sleep time (min)	170.2 ± 81.4	169.3 ± 80.6	171.2 ± 83.5
Supine sleep time (%sleep)	47.6 ± 20.4	48.0 ± 20.6	47.1 ± 20.5
Total AHI (/h) †	18.0 ± 8.9	19.1 ± 10.1	17.0 ± 7.4
Non-supine AHI (/h)	5.0 ± 3.2	4.6 ± 2.7	5.4 ± 3.5
Supine AHI (/h)	40.2 ± 38.8	46.1 ± 50.9	34.3 ± 19.7

Table 2.1 Patient Characteristic.

Data are presented as mean ± SD or number (%) of patients.

* The body-mass index is the weight in kilograms divided by the square of the height in meters.

[†] The apnea-hypopnea index is the number of apnea and hypopnea events per hour of sleep

There were no significant treatment order effects on any outcome. Both CPAP and supine-avoidance reduced the ESS compared to baseline (Figure 2.3 and Table 2.2). Furthermore, the reduction in ESS with supine-avoidance was non-inferior to that with CPAP, as the lower limit of the one-sided 95% confidence interval for the mean difference did not cross -1.5, the pre-specified margin of non-inferiority.

Table 2.2 shows device recorded usage data for supine-avoidance and CPAP. During the baseline period prior to commencing active treatment patients used the posture device (with inactive alarm for baseline monitoring of overnight posture) for around 8 hours almost every night in the home setting. Overnight supine time was consistently lower in the home compared to in-laboratory diagnostic and in-home baseline sleep study nights (around 25% of time in bed compared to around 50% and 30% of sleep time respectively). The device estimated percentage of time in bed spent supine was markedly reduced from around 25% over the baseline week to around 3% on active supine-avoidance treatment, with a reduction in average supine period duration from around 20 minutes to less than 2 minutes from an average of 5 alarm events per night. Most alarms resulted in a rapid shift to a non-supine posture, including 7917 and 8176 of 9552 alarms (82.9% and 85.6%) within 30 and 60 sec respectively. However, 730 of 9382 (7.8%) alarm events during non-PSG treatment nights, and a lower proportion of alarm events during PSG treatment nights (5 of 170 or 2.9%, Fisher's p=0.013), failed to discourage supine positioning before the alarm time-out at 5 minutes. Across all nights of 6-8 weeks of each treatment, patients used supine-avoidance therapy over 2 hours per night more than CPAP, and with a greater percentage of nights used for 4 hours or more hours per night.

				CPAP-supine-	
	Baseline	Supine-avoidance	CPAP	avoidance	Р
				difference	Value
Ν	62	59	62	57	
Nights available for use	9.7 [8.8 to 10.7]	52.6 [48.3 to 57.0] *	45.9 [41.9 to 50.0]	-6.0 [-12.2 to 0.2]	0.172
Average use (h/night)	7.9 [7.7 to 8.2]	5.7 [5.0 to 6.3] *	3.9 [3.2 to 4.6]	-1.7 [-2.5 to -0.9]	<0.001
Average use on nights used (h/night)	8.1 [7.8 to 8.3]	7.4 [7.0 to 7.7] *	5.2 [4.6 to 5.7]	-2.2 [-2.7 to -1.7]	<0.001
% nights used	98.3 [97.0 to 99.6]	75.4 [68.7 to 82.2]*	67.5 [58.8 to 76.2]	-7.1 [-16.6 to 2.3]	0.541
% nights used >4 h/night	97.3 [95.7 to 98.8]	71.8 [64.4 to 79.1]*	51.3 [42.3 to 60.3]	-19.4 [-29.1 to - 9.6]	<0.001
Supine time (%time in bed)	23.9 [20.1 to 27.7]	2.9 [1.2 to 4.5] *	-	-	
Supine periods (/night)	7.3 [6.6 to 8.0]	5.1 [4.3 to 6.0] *			
Supine period duration (min)	17.7 [14.7 to 20.6]	1.9 [1.1 to 2.7] *			
Alarms (/night)		4.7 [4.0 to 5.5]			

Table 2.2 Device data at baseline (inactive supine-alarm) and from 6-8 weeks of active supine-avoidance versus CPAP treatment.

Values are mean [95% confidence interval].*Indicates p<0.05 compared to baseline.

Figure 2.3 Epworth sleepiness scale (ESS) changes from baseline over the course of supine-avoidance versus CPAP treatment.



A) ESS (Epworth Sleepiness Scale) at baseline, 3-4 weeks of each treatment in each randomisation group and B) both groups combined. C) Change in ESS from baseline in each randomisation group and D) both groups combined. E CPAP-supines avoidance ESS difference between treatments. N=66 patients (33 in each group) at baseline. Values are mean±95% confidence intervals. A-D symbols in grey indicate values at baseline, open symbols indicate values during CPAP treatment and black filled symbols indicate values during supine-avoidance treatment.

The effects of CPAP versus supine-avoidance therapy on objective home sleep study measures of sleep, sleep posture and questionnaire outcomes are shown in Table 2.3. Both treatments reduced the amount of N1 sleep, total AHI, supine AHI, arousal index and 3% ODI, but with greater reductions in total AHI, arousal index and 3% ODI with CPAP compared to supine-avoidance. More patients achieved an AHI <10 /h during the home sleep study on CPAP compared to supine-avoidance (32% vs 12% Fisher's test p=0.017). CPAP also reduced non-supine AHI, but there was a substantially greater reduction in supine sleep time with supine-avoidance than with CPAP. Other indices of sleep quality, such as sleep efficiency, were not altered by either treatment. Both treatments also reduced the supine vs non-supine and on- versus off-treatment time-weighted estimated average AHI to a similar degree over the full-course of each treatment.

The complaints and treatment satisfaction questionnaire at the end of each treatment arm showed that 68% (34/50) of study participants would continue to use supine avoidance therapy compared to 55% (29/53, p=0.225) who said they would continue to use CPAP. At the end of the full-study, participants were also asked about their treatment preference, although only 22 patients answered the last treatment preference question. A significantly higher proportion of participants [72% (16/22)] reported that they preferred the supine avoidance therapy and were willing to continue with treatment compared to a preference for CPAP treatment [28% (6/22), p=0.006]. Two patients reported that they felt they had learned to avoid supine sleep and therefore did not require ongoing treatment. There were no significant changes with treatment or differences between treatments in PSQI, AQoL-8D or SSS. However, both treatments reduced the ISI, with a greater reduction following CPAP than with supine-avoidance (Table 2.3).

Figure 2.4. Supine-avoidance versus CPAP device usage over the course of the trial.



Values are group means±95% confidence intervals of average nightly device use with supine-avoidance (filled symbols) and CPAP (open symbols) device treatment at baseline and after 3-4 and 6-8 weeks of treatment. During the baseline period participants wore the supine-alarm device with the alarm inactive, followed by randomisation to active supine-avoidance or CPAP for 6-8 weeks before cross-over to the remaining treatment; with device data downloads at 3-4 and 6-8 weeks of each treatment. Grey indicates device recorded hours of use on nights with device use. Black indicates data from all nights, including zero hours when not used.

		Change from baseline at 6-8 weeks of treatment					
		CPAP-supine-					
	Baseline	Supine-avoidance	СРАР	avoidance difference	p-value		
Ν	66	53	52	49			
Sleep study parameters							
Total sleep time (TST) (min)	391.3 [371.8 to 410.7]	-3.3 [-26.7 to 20.0]	-1.5 [-30.6 to 27.6]	-2.6 [-40.5 to 35.2]	0.634		
Sleep onset latency (min)	22.9 [17.1 to 28.7]	-0.5 [-11.1 to 10.0]	2.9 [-7.1 to 12.9]	-0.8 [-12.6 to 11.0]	0.933		
Wake after sleep onset (min)	51.3 [42.6 to 60.0]	-7.0 [-17.5 to 3.6]	-7.3 [-18.4 to 3.9]	-2.9 [-15.1 to 9.3]	0.996		
Sleep efficiency (%)	81.2 [78.4 to 84.0]	2.5 [-1.8 to 6.9]	0.5 [-2.8 to 3.8]	-1.5 [-6.4 to 3.3]	0.998		
Supine sleep time (%sleep)	34.6 [28.8 to 40.4]	-20.3 [-27.1 to -13.4]*	12.8 [3.1 to 22.5]*	30.9 [22.2 to 39.5]	<0.001*		
N1 (%TST)	16.2 [13.8 to 18.7]	-3.7 [-6.4 to -1.0]*	-3.9 [-6.5 to -1.2]*	-0.7 [-2.6 to 1.1]	0.760		
N2 (%TST)	51.0 [48.3 to 53.7]	3.7 [0.3 to 7.2]*	2.9 [-0.8 to 6.6]	-1.6 [-5.2 to 1.9]	0.995		
N3 (%TST)	14.4 [12.1 to 16.6]	-0.2 [-2.8 to 2.4]	0.6 [-2.2 to 3.3]	0.4 [-2.2 to 2.9]	0.998		
REM (%TST)	18.4 [16.5 to 20.2]	0.2 [-2.2 to 2.5]	0.2 [-2.4 to 2.7]	-0.1 [-2.9 to 2.7]	0.986		
Total AHI (/h)	14.0 [10.8 to 17.2]	-4.2 [-7.8 to -0.6]*	-9.1 [-12.9 to -5.4]*	-5.3 [-9.3 to -1.3]	0.005*		
Supine AHI (/h)	27.4 [21.7 to 33.2]	-14.6 [-21.3 to -7.9]*	-17.6 [-24.4 to -10.9]*	-3.2 [-9.2 to 2.9]	0.994		
Non-Supine AHI (/h)	7.6 [4.9 to 10.3]	0.4 [-2.4 to 3.1]	-4.0 [-7.7 to -0.3]*	-5.2 [-8.5 to -2.0]	<0.001*		
Arousal Index (/h)	20.3 [18.1 to 22.4]	-2.9 [-5.5 to -0.2]*	-7.0 [-9.6 to -4.5]*	-3.6 [-5.9 to -1.2]	0.006*		
3% ODI (/h)	10.2 [7.6 to 12.9]	-3.0 [-5.6 to -0.4]*	-6.1 [-9.0 to -3.3]*	-3.3 [-5.5 to -1.1]	0.005*		
%Sleep Time SaO ₂ <90%	5.1 [3.3 to 6.8]	-0.9 [-3.8 to 2.0]*	-1.4 [-4.2 to 1.3]*	-1.7 [-3.7 to 0.4]	0.047*		
Treatment effectiveness							
Estimated average AHI (/h)	11.7 [8.5 to 14.9]	-3.1 [-4.9 to -1.2]*	-4.0 [-6.9 to -1.1]*	-1.3 [-3.3 to 0.6]	0.547		
Questionnaires							
ESS	10.3 [9.4 to 11.3]	-1.9 [-2.9 to -0.9]*	-2.3 [-3.3 to -1.3]*	-0.2 [-1.3 to 0.9]	0.956		
PSQI	8.3 [7.1 to 9.6]	-0.7 [-1.7 to 0.3]	-0.9 [-1.9 to 0.1]*	0.0 [-0.9 to 0.9]	0.894		
AQoL-8D	69.6 [66.2 to 73.0]	0.2 [-1.9 to 2.3]	1.9 [-0.6 to 4.4]	0.5 [-1.2 to 2.1]	0.900		
ISI	12.3 [10.9 to 13.7]	-1.4 [-2.6 to -0.2]	-3.6 [-4.7 to -2.4]*	-1.9 [-3.1 to -0.8]	0.011*		
SSS	6.1 [5.4 to 6.8]	-0.5 [-1.1 to 0.1]	0.2 [-0.8 to 1.3]	0.5 [-0.6 to 1.5]	0.698		

Table 2.3 Sleep study parameters at baseline and after 6-8 weeks of each treatment.

Values are mean [95% confidence interval], N=56. AHI = apnea-hypopnea index; ODI = oxygen desaturation index; ODI 3% = ODI, 3% desaturation; TST = total sleep time; VAS = visual analogue scale. AHI is calculated in supine position. N1 refers to sleep stage 1;

2.4 Discussion

This is amongst the first and largest well-powered randomised controlled trial to evaluate if alarm-based supine-avoidance treatment is non-inferior to CPAP for reducing daytime sleepiness, the primary symptom complaint of clinical concern in OSA, and the second study to directly compare treatment adherence and efficacy for reducing AHI versus CPAP in patients with supine-predominant OSA. The main findings support that simple supine-avoidance alarm device treatment to discourage supine sleep is non-inferior to CPAP in reducing sleepiness in patients with at least mild sleepiness (ESS>8). Furthermore, patient usage over 6-8 weeks of treatment was around 1.5-2 hours per night greater compared to CPAP, through both more prolonged overnight use when used and fewer skipped nights of use. These are important findings given that CPAP acceptance and use are often problematic, particularly in patients with milder OSA typical in supine-predominant OSA (Popescu et al., 2001). These findings are similar to a recent study showing non-inferior treatment efficacy in reducing overnight AHI and greater treatment adherence with a vibration based supine-avoidance alarm device compared to APAP in positional OSA patients (Berry et al., 2019), but where patients were not selected on the basis of sleepiness as in the current study. The findings from this study differ from those of Mok et al. (2020), where non-inferiority of supine-avoidance versus CPAP to reduce sleepiness could not be concluded from a smaller group of more symptomatic patients (ESS>10) (Mok et al. 2020). These somewhat divergent findings could indicate severity-dependent effects, where CPAP may be superior to supine-avoidance for reducing more severe OSA and sleepiness, or an inadequate sample size in the study of Mok et al. (2020) from which non-inferiority could not be confidently concluded or ruled out.

The efficacy of CPAP for reducing daytime sleepiness and AHI, and problems with treatment adherence are well established (Chen et al., 2015; Mcevoy et al., 2016; Weaver et al., 2007) and formed the basis for selecting CPAP as the optimal treatment comparator for this study. Although comparable and non-inferior to CPAP, the reduction in ESS with supine-avoidance was small, and below the reported minimally important clinical difference for treatment reductions in ESS of 2-3 (Patel et al., 2018). These small changes are likely reflective of relatively mild OSA and symptoms in patients with supine-predominant OSA who typically may not be recommended CPAP. Nevertheless, by study design, all patients had been referred for investigation and treatment, and selected on the basis of at least mildly symptomatic supinepredominant OSA, for which CPAP would be expected to be the most efficacious treatment. Whilst below the usual cut-off of 14 suggestive of insomnia (Gagnon et al., 2013), ISI improvements with both treatments, with larger improvements with CPAP, also support treatment benefits and a degree of OSA and insomnia symptom overlap (Cho Yong et al., 2018). No changes in self-reported sleep quality, quality of life or snoring scale score with treatment even with CPAP are perhaps surprising. This most likely reflects a combination of relatively mild OSA with greater efficacy, but poorer adherence with CPAP, compared to supine-avoidance, such that overall treatment effectiveness for reducing the estimated average overnight AHI and symptoms were more comparable. Given that patients with supine-predominant OSA constitute 30-50% of patients attending sleep clinics, these are important data to support the use of simple low-cost minimally intrusive supine-avoidance approaches for achieving comparable treatment outcomes with superior adherence compared to CPAP.

Whilst it might appear self-evident that patients with supine-predominant OSA should benefit from supine-avoidance treatment, favourable comparative-effectiveness evidence to support supine-avoidance therapy has previously been lacking. Long-term treatment effectiveness is the product of treatment efficacy, patient acceptance and ongoing adherence to treatment, where all must be sufficiently favourable to support effective outcomes long-term. Given that around 90% of patients commencing traditional discomfort based supine-avoidance with tennis-ball-treatment (TBT) selfreport abandoning treatment, most within one month from discomfort (Bignold et al., 2009), TBT clearly cannot be recommended long-term. Similarly, whilst CPAP is highly efficacious, sub-optimal treatment acceptance and adherence also limit long-term treatment effectiveness with CPAP (Weaver, 2006; Weaver and Grunstein, 2008). Thus, more effective treatment outcomes for patients with OSA requires more selective and sufficiently comfortable and efficacious treatment recommendations that patients will accept and use, and with favourable evidence to support their use. In this study supine-avoidance therapy was well accepted and used by participants, with higher use for most of the sleep period and on more nights compared to CPAP. Although efficacy for reducing AHI was reduced, greater treatment usage may help to explain a non-inferior reduction in sleepiness. Furthermore, a higher proportion of participants reported a preference for supine avoidance therapy over CPAP. In combination, these findings support that supine-avoidance alarm-treatment should be considered in appropriately selected patients with supine-predominant OSA, unlike

traditional discomfort based supine-avoidance approaches which clearly cannot be recommended given poor self-reported acceptance and use (Bignold et al., 2009).

Supine sleep avoidance is potentially the simplest and most effective way to treat OSA patients who exhibit sleep apnea mostly or exclusively during supine sleep. Similar to several previous smaller or shorter-term studies, our data strongly support the efficacy of supine-avoidance alarm-based devices for reducing supine sleep time and, when combined with high treatment adherence, to reduce AHI (Bignold et al., 2009; Bignold et al., 2011; Dieltjens et al., 2015; Ha et al., 2014; Levendowski et al., 2014; Van Maanen and De Vries, 2014). However, it should be noted that supine sleep was not completely abolished during supine-avoidance treatment, as indicated by residual supine periods with device measurements over the full treatment period, and independent posture measurements during the overnight sleep study on treatment. This likely reflects a combination of some degree of variability in posture classification with different posture measurement devices (Danker-Hopfe et al., 2009), more restricted movement with additional leads and sensors during sleep study nights, and failure of some alarm events to successfully discourage ongoing supine sleep. We have previously shown good posture classification agreement with supine-alarm device posture classification against in-laboratory video confirmed body position (Bignold et al., 2011), therefore posture classification disagreement is unlikely to have played a major role.

Similar to previous studies (Bignold et al., 2011; Levendowski et al., 2014), we found substantially greater supine time in the in-laboratory diagnostic (around 50% of total

sleep time) and baseline in-home sleep study (around 35% of total sleep time) compared to device estimated supine time (around 25% of time in bed) during the baseline week of home recording with an inactive supine-alarm. These findings support a substantial bias towards more supine-sleep and a higher AHI in most patients with a degree of supine-predominant events with conventional sleep monitoring, especially in a laboratory environment. This warrants careful consideration and potentially assessment of supine sleep habits in the home environment before selecting supine-avoidance over other treatments.

Also of note was that despite a strong vibratory stimulus, achieving comparable or perhaps greater supine-time reduction (to around 3 vs 8% of total sleep time) to a recent study using a different device (Berry et al., 2019), around 8% of all alarm events over the 6-8 week treatment period and 3% on sleep study nights failed to discourage supine positioning before the alarm time-out at 5 minutes. A lower failure rate on sleep study nights might indicate more rapid sleep resumption without more intrusive sleep measurements. Nevertheless, residual supine time from alarm events failing to shift away from the supine posture suggests that alarm improvements remain warranted. Given similar reductions in N1 sleep time with both treatments, with no differences in total sleep time, sleep efficiency or wake after sleep, another somewhat related concern that supine-alarms could potentially interfere with sleep was not supported. Nevertheless, further studies to examine the impact of supine-alarms on temporal relationships between arousal events, posture shifts and the resumption of sleep are likely to be useful.

Limitations

Several potential limitations warrant consideration. The study selection process for identifying patients with supine-predominant OSA reflects somewhat arbitrary criteria conservatively selected as most likely to achieve efficacious supine-avoidance and AHI reduction benefits. However, multiple criteria exist for defining supine-predominance (Joosten et al., 2014b; Frank et al., 2015; Mador et al., 2005a), OSA severity (Hudgel, 2016; Berry et al., 2012) and treatment success, with no clear consensus or robust clinical evidence to support which criteria may be best. Thus, results likely differ with application of alternative patient selection and clinical criteria.

As this was an open label study, self-reported outcomes such as ESS are inevitably prone to potential bias. Patient blinding to different forms of treatment is not possible and all self-reported outcomes with any treatment in usual clinical practice are subject to similar potential bias. Although AHI is clearly more objective, relationships between symptom complaints and AHI are week and patients do not specifically seek treatment for an elevated AHI. Hence, direct comparisons of sleepiness against best available care with CPAP in a cross-over design where each participant serves as their own control was considered the optimal and most pragmatic study design choice for testing the primary non-inferiority outcome. Moreover, substantial reductions in objective measures of supine time, with reduced N1 and arousal index, and similar timeweighted AHI over the full-course of treatment are consistent with comparable treatment effectiveness and symptom relief with both treatments. Instructions to participants could also be an important element of posture modification therapy success. For example, Neill et al (1997) found different upper airway closing pressure and upper airway opening pressure in a study where patients with OSA were instructed to sleep in either supine, lateral, or 30° elevation postures. Thus, to some extent participant self-selection of posture at sleep onset and the return to sleep following awakenings overnight may importantly influence sleep outcomes. However, although individuals may be able to successfully avoid falling asleep when supine, avoiding shifts to the supine following brief arousals without full awakenings during the sleep period may be resistant to conscious supine-avoidance and learning effects. In this study the intent of supine-avoidance therapy was clear to study participants, so participants may to some extent altered their sleep posture behaviours to help avoid supine-sleep. A more detailed examination of potential changes in posture behaviours and the number of alarm events over the course of treatment would clearly be useful to help evaluate potential behavioural and learning effects.

The estimated average AHI was calculated based on time-weighted averages to adjust for differences in both AHI and times spent in supine versus non-supine postures during sleep. This approach inevitably requires some simplifying assumptions and is partly dependent on PSG scorer assessments of time in bed based on sleep diary entries and signal characteristics suggestive of gross body movements associated with getting into and out of bed. Thus, the validity of these assessments is uncertain. It has also previously been shown that part of a night on CPAP can significantly change AHI for the second half of the night off CPAP (Stöberl et al., 2017), so potential carry-over effects also cannot be discounted. However, temporal changes may also be confounded by sleep stage distribution and other temporal influences across the night not necessarily related to treatment.

2.5 Conclusion

In summary, this study supports that supine-avoidance therapy via an alarm-based device worn on the chest is non-inferior to CPAP at reducing daytime sleepiness, and achieves substantially greater treatment adherence, with similar overall effectiveness and improvements in AHI and other objective sleep outcomes compared to CPAP. These findings contrast with previous data showing remarkably poor patient acceptance and adherence to traditional discomfort-based supine-avoidance treatment and support the use of non-discomfort alarm based supine-avoidance for treating appropriately selected patients with supine-predominant OSA.

2.6 Acknowledgment

The contributions from the following people are gratefully acknowledged; Dasha Loutchkina for study randomisation assistance; Mandy O'Grady. Alison Pinczel, Denzil Paul and Rohit Philip for technical assistance during the conduct of the trial; Dr Dimitar Sajkov, Dr Jason D'Costa, Dr Jeff Bowden, Dr Anand Rose, Dr Karen Latimer, A/Prof Brian Smith, Dr Mark Jurisevic, Dr Andrew Thornton, Dr Aeneas Yeo, Parmjit Singh, Dr Zafar Usmani and Dr Vinod Aiyappan for their assistance with patient recruitment. This study was funded by a National Health and Medical Research Council (NHMRC) Project Grant (1020892).

2.7 Authors contributions

MR undertook the primary data collation, analysis, interpretation and write-up of this work. PC, RDM, NA, MB, SQ, JM were involved in the conception, design and funding application. MO assisted with data acquisition.

CHAPTER 3. A study of physiological activation responses to posture shifts in sleep and the impact of a supineavoidance alarm

Key points

Question: This study sought to examine temporal relationships between EEG-defined wake and arousal responses to overnight posture shifts and to examine the effects of a supine-avoidance alarm device and CPAP therapy on overnight posture shift responses.

Findings: The majority of posture shifts during the sleep period were either preceded by wake or an arousal from sleep. Supine-avoidance therapy was effective in reducing supine sleep and was not associated with more prolonged awakenings following supine-alarm events.

Meaning: Supine-avoidance therapy via a vibro-tactile device stimulus worn on the chest is effective for reducing supine sleep without additional sleep disturbance in patients with supine-predominant OSA.

Abstract

Supine-avoidance alarm devices may be useful for treating patients with supine dependent obstructive sleep apnoea (SDOSA). However, little is known regarding temporal relationships between brief arousals/awakening responses and posture shifts during sleep, or the subsequent impact of supine-avoidance alarms on the time taken to return to sleep. This study aimed to assess the frequency and duration of arousal and awakening responses associated with posture shifts during sleep, and to examine how quickly sleep resumes following posture shifts with and without active supine-avoidance alarm treatment. This study was part of a larger randomised controlled trial comparing supine-avoidance versus CPAP therapy in patients with SDOSA. Following a baseline full in-home sleep study (Embletta MPR) with an inactive supine-avoidance alarm device (BuzzPOD, Gorman Promed Pty Ltd), patients were randomised to active supine-avoidance treatment vs CPAP for 6-8 weeks before cross-over to the remaining treatment for a further 6-8 weeks, with a repeat in-home sleep study at the end of each treatment. A technician blinded to supine-treatment data scored arousal and awakening responses (3 to <15-sec and ≥15-sec EEG changes respectively). All posture shifts ≥5-sec in duration commencing from established sleep during baseline and supine-avoidance nights were assessed for arousal/ awakening responses (within -30 to +3 sec). Posture shifts from wake, and sleep onset latency following each posture shift were also examined and compared between baseline and treatment sleep study nights.

Supine sleep was markedly diminished with supine-avoidance device treatment compared to baseline. Around 90% of posture shifts were preceded by either wake or

an arousal with no differences between treatment versus baseline nights. There were statistically significant effects of night (p<0.001) and state (wake vs sleep, p<0.001) preceding the posture shift but no significant night x state interaction effect on the time spent supine after posture shifts to supine. Almost all posture shifts to supine on PSG nights with active alarm treatment were followed by an alarm, but there were no differences in the propensity or time-course of achieving sleep between baseline and treatment nights.

The findings of this study support that most posture shifts during sleep are preceded by either wake or EEG arousal / awakening such that most supine alarm events are likely to occur during a period of brief cortical activation potentially needed for coordinated body movements with minimal effect on the time taken for sleep to resume. Thus, supine alarms themselves do not appear to interfere with the process of returning to sleep following overnight posture-shifts.

3.1 Introduction

Obstructive sleep apnea (OSA) is characterised by frequent partial or complete collapse of the upper airway during sleep. OSA has been associated with poor health outcomes such excessive daytime sleepiness (EDS), high blood pressure, depression and has been linked to a 2-7 fold increased risk of motor vehicle accidents that may lead to heightened risk of fatality (Dempsey et al., 2010; Tregear et al., 2009; Dopp et al., 2007). OSA is a heterogeneous disease with complex anatomical and physiological factors contributing to the initiation and severity of obstructive breathing events. Consequently, patients with OSA likely exhibit a range of phenotypes or endotypes for which improved identification of different phenotypes and causal mechanisms may facilitate better personalised and targeted treatments more likely to effectively and cost-effectively treat individual patients earlier and better than with current usual care. Following diagnosis, most patients with OSA undergo a protracted trial-and-error treatment approach until they either find and accept an effective treatment option, or are lost to ongoing medical care with uncertain but most likely unsatisfactory treatment outcomes.

In current clinical practice continuous positive airway pressure (CPAP) therapy is the standard first-line treatment for OSA patients regardless of phenotypic differences for which current diagnostic testing largely ignores. CPAP provides pressurised air through a mask during the night and is an effective treatment for reducing the apnea/hypopnea Index (AHI) (Sullivan et al., 1981) and normalising symptoms such as excessive daytime sleepiness. However, patient acceptance and regular ongoing

nightly use of CPAP is often limited such that the number of hours of CPAP usage per night are directly associated with CPAP effectiveness for reducing OSA symptoms (Weaver et al., 2007).

Several studies have shown that for CPAP to be optimally effective, patients are required to use the device at least 4 hours/night (Weaver et al., 2007; Wolkove et al., 2008; Bakker et al., 2019). Despite its effectiveness, CPAP is inherently an intrusive treatment and patients' adherence is generally low. Almost 50% of patients who are recommended to use CPAP stop using it after a median of 3 months resulting in their sleep apnea remaining untreated. Alternative treatments to CPAP do exist and include weight loss, upper airway surgery and mandibular advancement splint (MAS) devices. However, all are problematic. Weight loss is hard to achieve and difficult to maintain, MAS and surgery have unpredictable outcomes and surgery carries both surgical and anaesthetic risks, especially in patients with OSA more vulnerable to airway collapse during anaesthesia.

One major factor with significant potential to provide for an efficacious and affordable clinical treatment option for OSA is supine posture effects on OSA severity, which could potentially be markedly reduced via supine-avoidance therapy. OSA is often, although not always, more severe in the supine position so patients can be usefully divided into two groups of "supine dependent OSA" where sleep-related breathing abnormalities predominantly occur while sleeping supine, and "non-positional OSA" where respiratory events occur in both lateral and supine postures during sleep. There is no agreed standard definition of supine-dependent OSA, and depending how it is

defined an estimated 50%-60% of patients with OSA may have supine-dependent OSA with even higher prevalence of around 65% in mild to moderate OSA (i.e. AHI ranging from 5 to 30) (Oksenberg and Silverberg, 2009) and particularly in mild OSA (AHI 5-15 /h), reaching as high as around 87% (Mo et al., 2011). A recent study of 1719 community participants in Switzerland found OSA (AHI≥ 5 /h) to be highly prevalent and present in 71% of the study sample, within which 53% had position dependent OSA (Heinzer et al., 2018). This is an important phenotypic distinction as supine predominant OSA patients may substantially benefit from simply avoiding supine sleep. Thus, positional therapy could potentially be amongst the simplest and most effective treatment options for appropriately selected patients with positional OSA, particularly when treatment adherence with CPAP is often poor and treatment efficacy and patient acceptance and adherence to alternative treatments is more variable.

Given a high prevalence of positional OSA and that patients could potentially be treated simply by avoiding the supine position during sleep, supine-avoidance is an appealing treatment option. The work presented in Chapter 2 supports that 6-8 weeks of supine-avoidance alarm treatment is not inferior to CPAP in reducing daytime sleepiness, the primary OSA symptom complaint of greatest clinical concern in OSA. Furthermore patients also showed superior adherence to supine-avoidance therapy supporting the clinical utility of this simple minimally intrusive therapy. However, suitable caution remains warranted given evidence of poorer supine-avoidance alarm device efficacy at reducing AHI, and the combined importance of both efficacy and adherence in determining long-term treatment acceptance and adherence of any treatment. In the context of supine-avoidance treatment it is important to evaluate the impact of treatment on sleep itself, including posture-shift behaviours and the influence of supine-avoidance alarms on body movements and sleep.

Low adherence to traditional supine-avoidance strategies such as the tennis ball treatment has prompted exploration of new non-discomfort supine alarm devices to discourage supine sleep with minimal discomfort. These new generation devices are worn on the chest (Bignold et al., 2011) or on the neck (Levendowski et al., 2014; Van Maanen and De Vries, 2014) via a strap and use a position triggered alarm to vibrate whenever patients attempt to shift to the supine position, rather than an auditory alarm more likely to disturb bed-partner sleep. In addition to supine alarm treatment, these devices simultaneously record posture during sleep allowing for long-term monitoring of effectiveness to reduce supine time and treatment adherence, which has not previously been possible with traditional supine-avoidance approaches.

Despite effectiveness of vibratory alarm devices there is still ongoing uncertainty with all supine-avoidance treatments, including the new vibration-based alarm devices, regarding the impact of supine-avoidance alarms on sleep quality. In the context of new generation supine-avoidance devices, it remains unclear if vibration alarms triggered by attempts to shift to the supine posture in themselves might cause sleep disturbance. Supine alarms could promote extended wakefulness after a posture shift due to alerting effects of the strong vibration stimulus. Alternatively, posture-shifts themselves may require either wakefulness or brief arousal to engage co-ordinated muscle activity likely necessary to shift body position, in which case an additional vibration alarm may have relatively little impact on the process of returning to sleep following body movements. Remarkably little is known regarding the sequence of events preceding posture shifts in sleep, with no objective evidence available in the literature to guide if a brief return to consciousness is required to engage muscle activity to change posture, followed by a return to sleep with little or no conscious perception or lasting memory of the posture change itself (Tassi and Muzet, 2001). There have been no studies that have systematically analysed alarms generated by a supine-avoidance device in relation to physiological responses coincident with posture shifts with or without an alarm. In particular, the sequence of electroencephalogram (EEG) events occurring around the time of posture shifts have not been systematically investigated. This study aimed to characterise temporal relationships between EEG and arousal responses to posture shifts during sleep with or without a supineavoidance alarm to investigate potential alarm-related sleep disturbance. We hypothesised that the majority of overnight posture shifts would occur during wake or following a brief arousal and that sleep would quickly resume following posture shifts irrespective of an active supine-avoidance alarm.

3.2 Methods

This study was a sub-study of a larger comparative effectiveness trial (SUPA trial – Chapter 2) of 6-8 weeks of supine-avoidance treatment versus 6-8 weeks of CPAP for reducing sleepiness in patients with supine-predominant OSA. However, this study specifically reports only on data collected on polysomnography nights at baseline and at the end of each treatment period, with a primary focus on active versus inactive supine-avoidance alarm treatment.

3.2.1 Inclusion and exclusion criteria

Patients with supine-predominant OSA were recruited through sleep clinics at Repatriation General Hospital, Royal Adelaide Hospital, Ashford Hospital, Memorial Hospital, Queen Elizabeth Hospital and Flinders Medical Centre based on polysomnography (PSG) findings from each patient's initial diagnostic sleep study. Patients who met the study criteria were then invited to participate in the trial following review and approval from their sleep specialist.

The inclusion criteria used to select patients and to define supine-predominant OSA were; age 18 and over and less than 75 years, total AHI \geq 10 /hr, supine AHI \geq twice non-supine AHI, non-supine AHI <10/h, total sleep time \geq 4 hours with \geq 30 min supine and \geq 30 min non-supine sleep and Epworth Sleepiness Scale (ESS) \geq 8. Patients with any of the following criteria were excluded; prior treatment with CPAP or supine-avoidance therapy, potentially dangerous sleepiness requiring urgent treatment (e.g.

history of falling asleep while driving or ESS \geq 16), commercial driver, co-morbidities that may preclude supine-avoidance treatment (e.g. arthritis, pacemaker, mobility problems preventing non-supine sleep), unwilling to cease current alternative OSA therapies (e.g. mandibular advancement splint).

3.2.2 Supine-avoidance treatment

Supine-avoidance was achieved via a posture recording and alarm based vibration device (BuzzPOD, Gorman ProMed Pty Ltd, Melbourne, Australia, Figure 3.1) used in a previous study that demonstrated supine-avoidance efficacy and posture recording accuracy but without direct sleep measurements (Bignold et al. 2011). The device is worn on the chest during sleep using a Velcro strap and pouch, records body position via internal tilt switches sampled at 1 Hz, and records supine alarm events. A strong vibration alarm stimulus, externalised from the device to avoid interference with the internal tilt-switches, can be pre-configured to remain inactive or to activate when \geq 5 consecutive samples (seconds) are detected in the supine position. Once activated the device alarm remains continuously active for 2 minutes before switching to a pulsed 5-sec on/5-sec off mode for a further 3 min before the alarm times-out to conserve power when the ongoing alarm has failed to illicit a posture shift. The device has sufficient internal memory and typical battery life to support 4 weeks of continuous use prior to download via a personal computer and universal serial bus (USB) and device software interface.



Figure 3.1 Supine-avoidance device (left) showing externalised vibration motor, and a subject wearing the device (right).

3.2.3 Protocol

After screening and consent study participants were instructed in the use of the device and provided with an inactivated supine-avoidance alarm device for 1 week before returning to the laboratory on the last night to be setup for a baseline home PSG. Subsequently, participants were randomized to receive CPAP therapy or supineavoidance therapy for 6-8 weeks before crossing over to the remaining treatment for a further 6-8 weeks in order to complete the main trial, which included a repeat home PSG on allocated treatment at the end of each treatment arm.

3.2.4 Measures

Anthropometric measurements including age, gender, height, weight, waist circumference and body mass index (BMI) were determined at baseline and weight and BMI were also determined at the end of each treatment.

PSG signals were recorded using an Embletta MPR (REMLogic[™], Pleasanton, California, United States) home-PSG device and included EEG (C3-M2, C4-M1), left and right electro-occulograms (EOG), chin EMG (all sampled at 500 Hz), nasal cannula pressure and thermistor (sampled at 250 Hz), thoracic and abdominal motion (sampled at 100 Hz), oximetry (sampled at 3 Hz) and body position (sampled at 20 Hz). The supine-avoidance device independently recorded sleep posture at a sample rate of 1 Hz.

3.2.5 Polysomnography analysis

An experienced sleep technologist blinded to treatment allocation scored sleep stages and respiratory and arousal events according to American Academy of Sleep Medicine (AASM) alternate criteria (Iber et al., 2007) as described in more detail in Chapter 2. OSA was defined on the basis of an AHI \geq 10 /h (Ruehland et al., 2011). Total sleep time, sleep onset latency and wake after sleep onset were measured along with arousal index, number of awakenings and arousal durations.

3.2.6 Posture analysis

In order to assess temporal relationships between posture shifts and arousal events with the highest available temporal accuracy, posture data collected on the PSG device (sampled at 20 Hz) and supine-avoidance device (sampled at 1 Hz) were extracted via custom algorithms to detect all posture shifts lasting ≥5-seconds for further analyses. For supine-avoidance device recorded data the duration of each posture shift lasting ≥5-sec was determined and classified on the basis of posture shifts from and to non-supine versus supine postures. The presence and duration of any alarm event associated with shifts to the supine posture were also determined. The same posture duration and shift from and to analysis was applied to PSG recorded posture data using the default device angle cut-offs for classifying postures. The frequency of occurrence, duration and total cumulative time spent in supine versus non-supine postures following each type of posture shift (non-supine to supine, supine to non-supine and non-supine to a different non-supine posture) from both recording devices were also examined. EEG data 30 secs before and after each PSG recorded posture shift were also examined to assess wake versus sleep and arousals prior to and the time taken to achieve sleep following each posture shift. Prior sleep versus wake was determined from the epoch of the posture shift when posture shifts occurred in the last 15-sec of the epoch, or the prior epoch when the posture shift occurred in the first 15-sec of an epoch. Whilst imperfect, this classification strategy maximises the available temporal resolution of traditional 30-sec epoch sleep scoring. Following identification of all \geq 5 sec posture shifts, sleep vs wake stage before each posture shift and the onset time of the nearest arousal within 30 sec before to +3 sec after each

posture shift were used to classify the presence of arousal associated with the posture shift. -30 and +3 seconds were chosen on the basis of epoch length used to define sleep vs wake state preceding posture shifts, and to allow for variability in scorer marking of arousal onsets. Sleep onset times following each posture shift were determined on the basis of the number of epochs taken for sleep to resume (i.e. with 30 sec temporal resolution, as typically applies for sleep onset latency calculations). Given no internal clock or synchronisation channel options available to help more accurately temporally align supine-avoidance device posture and alarm recordings to PSG recorded events, it was not possible to more accurately synchronise PSG versus supine-avoidance device posture recording devices. Thus, it was only possible to undertake a meaningful analysis of temporal relationships between posture shifts and EEG defined sleep stage, arousals and sleep onset times using PSG-recorded posture shift, sleep and arousal data independent of the supine-avoidance device recordings.

3.2.7 Statistical Analysis

Statistical analyses were performed using IBM SPSS Statistics version 23. Normality was assessed using Kolmogorov-Smirnov tests. Normally distributed data are reported as mean ± SD (or 95% confidence intervals as indicated) and were compared between baseline and treatment nights using mixed effects model analysis. The total number of posture shifts under each condition were compared between nights using Fisher's exact tests. Non-normal data are reported as median [interquartile range] and were compared between baseline and treatment nights using mixed effects using Wilcoxon sign-rank
tests. Kaplan-Meier and Cox regression analyses were used to examine the effects of treatment (vs baseline) and sleep vs wake state prior to supine and non-supine posture shifts on the time taken to shift to another posture, and the time taken for sleep to resume following supine and non-supine posture shifts (reported as mean [95% confidence interval]). P values < 0.05 were considered statistically significant.

3.3 Results

56 patients out of 66 enrolled in the main trial successfully completed baseline and both treatment nights for this analysis. Table 3.1 shows the baseline characteristics of the patients included in analysis. Patients were mostly middle-aged men (34 males and 22 female) and obese (BMI>30 kg/m²), and by study inclusion criteria, all had supine-predominant OSA.

Parameter	Value
N (% Male)	56 (60.7%)
Age (yr)	53.6 ± 11.2
Body mass index (kg/m²)	31.4 ± 8.1
Diagnostic sleep study	
Supine sleep (%total)	44.6 [33.3 to 64.8]
Total AHI (/h)	14.5 [12.2 to 20.7]
Non-supine AHI (/h)	5.1 [2.3 to 7.5]
Supine AHI (/h)	31.5 [21.4 to 45.0]
Epworth Sleepiness Score (ESS)	10.0 [8.0 to 13.0]

Table 3.1 Basic characteristics of the study population.

N, Age and Body Mass Index are mean \pm SD. Supine sleep, Total AHI, Non-supine AHI Supine AHI, Epworth Sleepiness Score are median [IQR].

There were several technical problems with the acquisition of posture data from PSG files that reduced the number of participants with data available for detailed PSG recorded posture analysis to 54, 38 and 45 baseline, supine-avoidance and CPAP PSG nights respectively. The number posture shifts \geq 5 seconds during the sleep period recorded by the PSG device (i.e. Embletta 100x) on each PSG night are shown in Table 3.2. There was no significant reduction in the total number of posture shifts on treatment nights compared to the baseline night. Around 60% of all posture shifts were preceded by wake and most posture shifts from sleep on all three PSG nights were preceded by an arousal such that the majority of all posture shifts were preceded by either wake or arousal from sleep (P<0.001, Table 3.2).

Table 3.3 summarises the total number of ≥5-sec posture shifts and alarm events recorded by the supine-avoidance device during the baseline week, PSG nights and subsequent 6-8 weeks of active supine-avoidance treatment use. Figure 3.2 shows the average number of posture shifts per night, their mean duration and total cumulative hours spent following posture shifts from non-supine to supine, supine to non-supine and non-supine to non-supine postures as recorded by the supine-avoidance and PSG devices.

Table 3.2 The number of PSG device (Embletta) recored posture shifts ≥5-sec.

	To Supine	To Non-Supine	Total
	Baseline PSG (N=54)		
N shifts	6.0 [4.0 to 9.3]	9.0 [5.0 to 12.0]	16.0 [10.0 to 21.5]
N shifts during sleep	3.0 [2.0 to 5.0]	3.0 [2.0 to 6.0]	6.0 [3.0 to 10.0]
N shifts during sleep preceded by arousal	3.0 [1.0 to 4.5]	2.0 [1.0 to 5.0]	5.0 [2.0 to 9.0]
% Sleep shifts preceded by arousal	100 [75 to 100]	100 [78 to 100]	94 [78 to 100]
% Total shifts preceded by wake or an arousal	100 [90 to 100]	100 [89 to 100]	100 [91 to 100]
Arousal duration (sec)	15.7 [12.5 to 17.4]	15.8 [13.4 to 19.1]	15.9 [13.0 to 17.4]
SOL following shift from sleep (min)	0.5 [0.2 to 1.1]	0.6 [0.3 to 1.5]	0.8 [0.4 to 1.5]
	Supine avoidance PSG (N=	=38)	
N shifts	6.0 [3.0 to 9.0]	9.0 [6.0 to 12.0]	15.0 [8.3 to 20.8]
N during sleep	2.5 [1.0 to 5.0]	3.0 [2.0 to 5.0]	6.0 [3.0 to 10.0]
N shifts during sleep preceded by arousal	2.0 [1.0 to 4.8]	3.0 [2.0 to 5.0]	6.0 [3.0 to 8.3]
% Sleep shifts preceded by arousal	100 [60 to 100]	100 [100 to 100]	100 [84 to 100]
% Total shifts preceded by wake or an arousal	100 [86 to 100]	100 [100 to 100]	100 [93 to 100]
Arousal duration (sec)	15.1 [13.5 to 17.1]	15.3 [13.0 to 17.1]	14.9 [13.3 to 16.7]
SOL following shift from sleep (min)	0.6 [0.1 to 1.4]	0.5 [0.3 to 1.0]	0.5 [0.3 to 1.2]
	CPAP PSG (N=45)		
N shifts	5.0 [3.0 to 9.0]	7.0 [5.0 to 12.0]	13.0 [8.0 to 20.0]
N during sleep	2.0 [1.0 to 4.0]	3.0 [2.0 to 5.0]	5.0 [3.0 to 8.3]
N shifts during sleep preceded by arousal	2.0 [1.0 to 3.0]	3.0 [1.0 to 4.0]	4.5 [2.8 to 7.0]
% Sleep shifts preceded by arousal	100 [83 to 100]	100 [83 to 100]	97 [79 to 100]
% Total shifts preceded by wake or an arousal	100 [100 to 100]	100 [94 to 100]	100 [90 to 100]
Arousal duration (sec)	15.7 [13.4 to 17.9]	16.9 [14.2 to 19.5]	16.3 [14.5 to 18.6]
SOL following shift from sleep (min)	0.5 [0.2 to 1.2]	0.5 [0.3 to 1.1]	0.8 [0.4 to 1.4]

All values are presented as median [interquartile range].Sleep onset latency (SOL).

-		• •	
	To Supine	To Non-Supine	Total
Ba	seline (N=56)		
N total	3262	7662	10924
N /night	6.9 [4.8 to 8.7]	15.0 [10.5 to 18.2]	21.6 [15.8 to 26.9]
Ва	seline PSG (N=50)		
N total	338	657	995
N /night	8.0 [4.8 to 11.0]	13.0 [8.3 to 16.0]	19.0 [13.5 to 26.0]
Su	pine-avoidance PSG (N=45)		
N total	161*	611	772*
N /night	4.0 [2.0 to 6.3]*	12.0 [9.0 to 16.0]	16.0 [10.0 to 20.0]
N alarms total	158		
N alarms/night	4.0 [2.0 to 6.3]		
Mean alarm duration (sec)	7.0 [3.4 to 11.7]^		
Su	pine-avoidance treatment p	eriod (N=56)	
N total	9030	37642	46672
N /night	3.4 [1.7 to 5.8]*	14.9 [11.9 to 20.3]	18.6 [14.3 to 24.8]
N alarms total	8830		
N alarms/night	3.4 [1.6 to 5.4]		
Mean alarm duration (sec)	12.3 [6.6 to 23.7]		

Table 3.3 The total number of \geq 5-sec posture shifts and alarm events recorded by the supine-avoidance alarm device.

Values are total numbers or median [interquartile range]. *indicates p<0.05 vs baseline PSG. ^ vs supine-avoidance treatment period.

During PSG nights with an active supine-avoidance alarm the number of shifts to supine were significantly reduced in both supine-avoidance and PSG device recorded posture data compared to the baseline PSG night with an inactive alarm. In addition, almost all supine-avoidance device recorded shifts to supine produced an alarm event (Table 3.3).

Active supine-avoidance treatment produced a total of 158 alarm events from 161 \geq 5sec shifts to supine on the PSG night (Table 3.3) and markedly reduced the average duration and total cumulative time spent supine after posture shifts to supine (Figure 3.2). However, on the PSG night there were 5 (3.2%) alarm events associated with ongoing supine sleep time of more than 5 minutes (ranging from 5.6 to 99.3 minutes in duration) most likely indicating sleep-through alarm events. However, it was not possible to accurately align device alarm events with PSG recorded sleep and wake data to clarify if these were sleep-through events. Furthermore, whilst supineavoidance device recorded posture shifts showed marked reductions in supine time during both PSG nights and the 6-8 week supine-avoidance treatment period (P=0.006, P=0.001 respectively) with corresponding increases in non-supine time (P=0.001, P=0.007 respectively) (Table 3.3 and Figure 3.2), PSG device recordings showed no statistically significant reductions in supine time with supine-avoidance treatment (Table 3.2, Figure 3.2 left versus right panels).



Figure 3.2 Box and whisker plots of the number, duration and total cumulative time spent following each type of posture shift over the course of the study.

Boxes indicate median, interquartile range and mean (x) number (top), duration (middle) and total cumulative time (bottom) of non-supine to supine (grey boxes), supine to non-supine (open boxes) and non-supine to non-supine (patterned boxes) posture shifts per night. The left panels show supine-avoidance device recorded posture shifts lasting \geq 5-sec during the baseline period (7 days), baseline PSG night, supine-avoidance (SUPA) PSG night and subsequent 6-8 week active treatment period. The right panel shows PSG device recorded posture shifts lasting \geq 5-sec on the baseline, supine-avoidance (SUPA) and CPAP PSG nights. * indicates p<0.05 vs non-supine to non-supine shifts and † p<0.05 vs non-supine to supine shifts within the same treatment condition.

Kaplan-Meier curves showing the time spent supine following \geq 5-sec supineavoidance device recorded and PSG recorded \geq 5-sec shifts to the supine-posture under each condition are shown in Figure 3.3. There were statistically significant differences in the hazard of remaining supine in the 10-min following shifts to the supine posture with an inactive compared to an active supine-alarm (p<0.001). However, there were no differences between the baseline period and baseline PSG nights (p=0.180) or between the supine-avoidance PSG versus the remaining all 6-8 weeks of supine-avoidance treatment nights (p= 0.638, Figure 3.3). Compared to baseline, the hazard or remaining supine was significantly reduced with active supineavoidance alarms in both supine-avoidance (p<0.001) and PSG (p<0.001) device recorded data but was significantly higher with CPAP therapy (Figure 3.2).

Kaplan-Meier curves showing the proportion of PSG recorded posture shift events to supine from sleep or wake associated with ongoing wake are shown in Figure 3.4. The hazard of remaining awake following supine posture shifts from sleep was markedly reduced compared to posture shifts from prior wake (p<0.001, Figure 3.4). However, there were no significant differences between nights in the hazard of remaining awake following posture shifts to supine from either sleep (p=0.941) or wake (p=0.306). Thus, sleep resumed rapidly following supine-posture shifts, particularly from sleep, despite the presence of supine-avoidance device alarms.

Figure 3.3 Kaplan-Meier plots showing the proportion of patients remaining in the supine posture as a function of time in the 10 min following \geq 5 sec device recorded shifts to the supine posture (left) and \geq 5 sec PSG recorded shifts to the supine posture (right).



Baseline (~5-7 nights, grey dashed line) and baseline PSG (black dashed line) nights were recorded with the device supine alarm inactive. Supine-avoidance treatment nights (6-8 weeks, grey solid line) and a supine-avoidance PSG recording night (black solid line) were recorded with the supine alarm activated. HR indicates hazard ratios [95% confidence intervals] versus the baseline PSG night (ref). Tabulated numbers indicate the total number of posture shifts and individual participants 'at risk' of remaining supine over the 10 minutes after the onset of each posture shift.

Figure 3.4 Kaplan-Meier plots showing the proportion of PSG device recorded posture shifts ≥ 5 sec followed by wake in the 10 minutes after a shift from non-supine to the supine posture from prior sleep (left) versus wake (right) during the baseline, supine-avoidance and CPAP PSG nights.



Prior sleep versus wake was determined from the epoch of the posture shift when posture shifts occurred in the last 15-sec of the epoch, or the prior epoch when the posture shift occurred in the first 15-sec of an epoch. Tabulated numbers indicate the total number of posture shifts and individual participants remaining 'at risk' of remaining supine over the 10 minutes after the onset of each posture shift.

Further analysis of posture distributions in different sleep stages during PSG nights showed that the majority (~60%) of shifts occurred during wake, but with around 17% of posture shifts in N2 sleep and very few posture shifts in deep sleep. However, there were no differences between baseline or treatment nights in posture shift distributions across sleep.

Table 3.4 Distribution of posture shifts as a function of sleep stage immediately prior to the posture shift.

	Wake	N1	N2	N3	REM
Baseline	60.0 [45.6 to	3.1 [0.0 to	16.7 [4.7 to	0.0 [0.0 to	5.0 [0.0 to
	80.0]%	11.1]%	20.7]%	3.3]%	25.0]%
Supine-	59.4 [48.0 to	4.8 [0.0 to	17.4 [9.6 to	0.0 [0.0 to	2.1 [0.0 to
avoidance	76.7]%	11.3]%	27.5]%	10.0]%	14.0]%
CPAP	66.7 [50.0 to	0.0 [0.0 to	16.7 [7.7 to	0.0 [0.0 to	6.1 [0.0 to
	80.0]%	10.0]%	25.7]%	6.8]%	12.5]%

Values are median [interquartile range] of all posture shifts ≥5-sec of any type (nonsupine to supine, supine to non-supine or non-supine to supine. N1, N2, N3 indicate the different stages of non-REM sleep. REM: Rapid eye movement sleep.

3.4 Discussion

This is the first study to objectively examine the sequence of EEG events occurring before and after overnight posture shifts and also the first to investigate the effects of a supine-avoidance alarm device and CPAP therapy on posture shifts in patients with supine-predominant OSA. The main findings support that most overnight posture shifts during the sleep period are preceded either by wake or by an EEG discernible arousal or awakening event. Furthermore, a supine-avoidance alarm commencing 5-sec after a shift to the supine posture had no significant impact on the time taken for sleep to resume, and most of the time very effectively discouraged the user from remaining supine.

This study extends the previous findings of Bignold et al (2011), who found that time spent supine is markedly reduced with the same chest-worn posture recording and supine-avoidance alarm device. These new findings clarify that this reflects a combination of reduced supine sleep and wake time, and that posture shift behaviours and supine-avoidance effectiveness remain largely unchanged over 6-8 weeks of ongoing treatment with no major difference compared to baseline in the total number of shifts to supine by the end of 6-8 weeks of supine-avoidance treatment. This supports that patients do not simply learn to avoid the supine posture and appear to require ongoing alarms around 3-4 times per night in order to successfully avoid prolonged supine periods during sleep. Two other alternative vibration based alarm devices used by two separate groups in Belgium and the USA also show significantly reduced time spent supine during sleep (Van Maanen et al., 2013; Levendowski et al.,

2014). However, this is the first systematic study of the time taken to achieve sleep after supine posture shift events.

Unfortunately, it was not possible to accurately align supine-avoidance device versus PSG recorded device data to more comprehensively examine alarm or other physiological activation outcomes with supine-avoidance compared to baseline or PSG treatment nights. Nonetheless, and despite the strong device vibration stimulus and significant reduction in supine sleep, there were clearly occasional prolonged periods of residual supine time. These events are consistent with a rapid return to sleep followed by ongoing failure of the vibration alarm to effectively discourage supine positioning, thus negatively impacting on supine-avoidance efficacy. These are important observations and suggest that further technological improvements to increase supine-avoidance efficacy may be warranted. Nonetheless, overall supine-time remained markedly reduced. There was also no evidence of extended wake following device alarms and no significant difference between the cumulative time that patients spent supine during nights when they were equipped with an activated alarm device compared to baseline nights when the alarm was inactive.

The primary aim of this study was to examine temporal relationships between sustained posture shifts and EEG detected arousals. Analysis of baseline and treatment night sleep studies showed that most (~80%) sustained posture shifts from sleep were preceded by an arousal. These data are consistent with the hypothesis that most but perhaps not all body position shifts arising during sleep are preceded by an arousal. This could suggest that brief cortical activation is not a pre-requisite for all

body position changes in sleep. Alternatively, this could reflect an artefact of manual PSG scoring, which is well known to show considerable within- and between-scorer variability, particularly for arousal events (Ruehland et al., 2011). Consequently, some arousal events may well have been missed. Furthermore, independent manual identification of arousal onsets versus posture shifts may not be sufficiently accurate to reliably identify the onset times of arousal versus posture shift events and may systematically under-estimated the number of posture-shifts associated with cortical arousal. More comprehensive analysis of EEG activation responses, such as with quantitative EEG methods, along with examination of other physiological activation responses such as heart rate/pulse responses around the time of positions shifts may well be useful in future studies, but will need more accurate device synchronisation with sleep study recordings than was possible in this study.

Although how much "*consciousness*" is needed for a posture shift in sleep remains a matter of debate (Tassi and Muzet, 2001), frequent cortical activation preceding sustained posture shifts remains consistent with the concept that posture shifts require EEG activation to engage muscles needed for body movement, and that alarm events occurring shortly after posture shifts likely already coincide with a partial return to wake. Body movements during sleep could primarily be the result of accumulating discomfort associated with staying in one position for an extended period of time (De Koninck et al., 1983) in which case posture shifts might be expected to be distributed across most stages of sleep. In healthy young adults, De Koninck observed most body movements to occur during REM (De Koninck et al., 1983), although others have suggested a trend towards reduced movement with aging and no clear relation to

sleep stage/state (Gori et al., 2004). The current study found that the majority of posture shifts occurred during wake, with fewer initiated from N2, REM and N1 sleep respectively and very few from deep sleep. These findings appear to be largely consistent with the uneven distribution of sleep stages in combination with diminished sensory acuity during sleep, which is particularly prominent in deep sleep and to a lesser extent in REM, N2 and N1 (Tassi and Muzet, 2001; Pare and Llinas, 1995). Thus, a bias towards posture shifts occurring during wake and lighter stages of sleep remains consistent with a combination of accumulating posture discomfort and sleep depth dependence of sensory acuity. It is not clear whether patterns of movement across different sleep stages, or the control of the motor cortex and subcortical circuits are affected by sleep apnea. Further research is needed to better understand the main triggers for posture shifts in sleep and potential relationships between different stages of sleep and movement during sleep in relation to brain activity.

In addition to effectively reducing supine sleep, the supine-avoidance device used in this study did not prolong arousal/awakening responses associated with supine alarms. Most posture shifts from prior sleep were associated with a brief prior arousal followed by a rapid return to sleep within a few seconds to minutes of the position shifts, with no significant lengthening of arousals or sleep onset latency with an active supine alarm. Thus, there was no evidence to support sleep disturbance associated with the alarm *per se*. However, there was also no evidence to support improved sleep with the supine-avoidance alarm or CPAP, other than a previously reported reduction in AHI (Chapter 2). In contrast, when patients were already awake immediately before shifting posture it took considerably longer for sleep to resume, irrespective of the

alarm activation status or use of CPAP. Furthermore, there were clearly some differences in posture shift assessments between PSG and the supine-avoidance devices, with a substantially greater proportion of PSG versus supine-avoidance device recorded posture shifts to supine remaining supine after 10 min. The most likely explanation for these findings is that the normal central placement of the PSG recording device was displaced somewhat laterally due to the presence of the chest-worn supine-avoidance and recording device. Thus, posture assessment from the PSG device may be less reliable compared to the chest-worn device, which has previously been shown to reliably classify posture compared to video-confirmed posture assessments (Bignold et al., 2011). Despite clear posture classification differences, PSG device assessments of supine sleep were still shown to be significantly reduced during supine-avoidance treatment. In future studies, more accurate device synchronisation and positioning strategies would be very useful to more directly and reliably compare supine-avoidance treatment effectiveness and relationships with PSG determined sleep outcomes.

Other methodological considerations

The population represented in this study were specifically selected on the basis of a supine AHI twice that of the non-supine AHI along with a clinically abnormal supine AHI (>10 /h) and a normal non-supine AHI (<10 /h) (Bignold et al., 2011; Mador et al., 2005a). Whilst it is unclear if selecting patients with somewhat different characteristics would change the study outcomes, this patient group remains one of the most relevant to supine-avoidance treatment given that supine-avoidance should effectively normalise OSA according to these criteria. Nonetheless, further investigations to

establish outcomes in supine-predominant snorers and perhaps less sleepy patients who could potentially exhibit more sleep disturbance to alarm-based treatments may be warranted.

The supine-avoidance alarm used in this study did not possess an internal clock. Developing methods to synchronise device recorded posture shift and alarm data with PSG events was challenging and has not been attempted previously (Levendowski et al., 2014; Van Maanen et al., 2013). However, the accuracy of synchronization was inevitably more limited than within PSG recorded data but where temporal alignment with alarm events was uncertain. The lack of an internal clock in the supine avoidance device meant that accurate synchronization with sleep data (i.e. PSG data) was ultimately not possible. Thus, analysis of posture shift behaviours was conducted independently on supine-avoidance and PSG device data. Thus, future studies would benefit from more reliable methods to align alarm events with PSG recordings.

In summary, the findings of this study support that most posture shifts during sleep are preceded by either wake or EEG arousal / awakening such that most supine alarm events occur around the time of brief cortical activation potentially needed for co-ordinated body movements with minimal effect on the time taken for sleep to resume. Further studies examining posture shifts and other activation responses such as heart rate and airflow may provide further understanding of physiological disturbances associated with position changes in sleep.

CHAPTER 4. A comparison of snoring changes with a supine-avoidance alarm device compared to constant positive airway pressure treatment in patients with supine-predominant OSA

Key points

Question: This study investigated the effectiveness of supine-avoidance therapy via a vibratory alarm device and CPAP therapy in reducing objectively measured snoring frequency in patients with supine-dependent obstructive sleep apnea (SDOSA).

Findings: Whilst CPAP therapy successfully reduced snoring, supine-avoidance therapy did not systematically reduce snoring. However, in a sub-group of patients who suffered from supine snoring as well SDOSA supine-avoidance therapy significantly reduced snoring as well as CPAP.

Meaning: Reducing supine sleep via supine-avoidance therapy may help a subgroup of patients who suffer from supine-mainly snoring and SDOSA to significantly decrease both their snoring and OSA.

Abstract

Snoring is one of the primary complaints from patients or their partners when they seek sleep physician advice. Snoring can substantially disrupt patient and particularly bed-partner sleep contributing to insomnia, psychosocial problems and reduced quality of life for both snorers and their bed partners. Despite this, snoring is not routinely assessed and is often largely ignored in clinical settings where the main focus is typically on other measures such as the apnea hypopnea index, oxygen desaturation index and arousal index. This study aimed to analyse, for the first time, snoring data from the SUPA trial objectively (Chapter 2) to quantify how much snoring occurs in clinical patients with supine-dependent OSA and the comparative effectiveness of supine-avoidance therapy versus CPAP to reduce objective measures of snoring. Snoring and treatment outcomes were also examined in a subgroup of patients with supine dependent snoring (defined on the basis of snoring frequency at least 2x higher in supine than non-supine postures). Baseline measurements showed snoring frequency was 48.9 [95% CI 16.7 to 188.7], 26.3 [95% CI 17.0 to 106.3], 57.5 [95% CI 16.8 to 190.7] snores/h for the whole group, supine snorer and non-supine snorers respectively. Supine sleep as a percentage of total sleep was almost completely abolished with supine-avoidance treatment (Baseline= 30.7% [95% 15.4 to 46.4], supine-avoidance= 0.2% [95% 0.0 to 15.7]). In the whole group, CPAP significantly reduced overall snoring frequency although some residual snoring remained. However, no treatment effect was observed with supine-avoidance therapy in reducing snoring frequency (frequency of snores ≥50 dBA; baseline= 48.9 [95% 16.7 to 188.7], supine-avoidance= 36.8 [95% 6.3 to 233.7], CPAP= 4.2 [95% 2.1 to 29.5] snores /h). However, in the sub-group of supine-dependent snorers supineavoidance therapy was effective snoring frequency (Baseline= 26.3 [95% 17.0 to 106.3], supine-avoidance= 14.3 [95% 3.0 to 18.1], CPAP= 4.6 [95% 2.6 to 8.9] snores/h). In conclusion, supine-avoidance therapy may not necessarily lead to a systematic reduction of snoring frequency in patients with supine-predominant OSA, but does appear to be effective in the subgroup of patients with supine-predominant snoring. This study supports that in suitably selected patients, supine-avoidance is a simple, feasible, effective and affordable approach well-suited to patients with supine-predominant OSA and/or snoring.

4.1 Introduction

Snoring and obstructive sleep apnea (OSA) are part of a continuum of obstructed breathing in sleep and underpin the majority of referrals to a sleep physician for a sleep study (Larsson et al., 2003). Snoring reflects Starling resistor-like behaviour of the upper airway where dynamic partial airway collapse limits airflow irrespective of inspiratory driving pressures(Wellman et al., 2014). Consequently, a portion of often substantial inspiratory effort is converted into tissue vibration and snoring with reduced airflow (hypopnoea) instead of normal unimpeded airflow and ventilation. Although OSA can include apneas without snoring, and snoring can be sustained for long periods that fail to meet traditional hypopnea scoring criteria, most respiratory events in OSA reflect transient hypopneas (Ratnavadivel et al., 2009) likely to be directly associated with snoring. Thus, up to 95% of all OSA patients snore heavily most or every night (Viner et al., 1991) and snoring frequency and intensity are strongly associated with respiratory and arousal disturbances in sleep (Viner et al., 1991).

Habitual loud snoring without OSA affects somewhere between 15%-54% of all middle-aged adults, potentially making it a more common problem than OSA (Young et al., 1993; Enright et al., 1996). In clinical settings, snoring is a clear sign of a collapsible airway, but is considered less of a clinical concern than frequent transient obstruction, desaturation and arousal events that contribute to an elevated apnea hypopnoea index (AHI) in OSA. Nevertheless, habitual loud snoring without

necessarily a high AHI may have important negative effects on the snorer and bed partner sleep quality and health.

Snoring can substantially disrupt patient and particularly bed-partner sleep contributing to insomnia, psychosocial problems and reduced quality of life for both snorers and their partners. Snoring has been linked to marital problems (Jones and Swift, 2000), occasional acts of violence including homicide (Pelausa and Tarshis, 1989), and is one of the primary reasons patients seek sleep treatments from ear, nose and throat (ENT) specialists (Marshall et al., 2007; Mcardle et al., 2001). Snoring could also contribute to increased cardiovascular disease risk due to vibration-induced vascular injury and effects of chronic cardiac exposure to large negative intrathoracic pressures known to be associated with right ventricular hypertrophy (Amatoury et al., 2006; Curry et al., 2002; Puig et al., 2005). Substantial vibration energy and the associated tissue stress caused by snoring is likely to be transmitted to the nearby carotid arteries, potentially contributing to endothelial tissue damage, inflammation and potentially plaque rupture risk (Curry et al., 2002). A dose-dependent relationship between snoring severity and carotid atherosclerosis but not femoral artery atherosclerosis is consistent with this hypothesis (Lee et al., 2008). Thus, chronic obstructed breathing and snoring in sleep could contribute to increased adverse cardiovascular outcome risk, including stroke.

Upper airway collapsibility, and thus snoring, during sleep is strongly body position dependent and both are typically higher during supine compared to non-supine sleep. Although the prevalence of supine-predominant snoring is not well-established, approximately 50% of all patients diagnosed with OSA show strong supinepredominance of OSA severity, with an at least a 2-fold higher supine versus nonsupine AHI (Adams et al., 2017). The majority of snorers without OSA also snore less in lateral compared to supine sleep (Loord and Hultcrantz, 2007; Miyamoto et al., 1997). If the prevalence of supine-predominant snoring follows a similar pattern to OSA, then around 20-30% of all adults potentially exhibit supine-predominant snoring and could potentially benefit from supine-avoidance therapy. Given a growing body of evidence to support that simple supine-avoidance devices can effectively reduce supine sleep time, nearly one third of OSA patients who exhibit clinically significant OSA only when supine, and a large number of snorers without OSA (Oksenberg and Gadoth, 2019; Berry et al., 2019), could potentially be treated through simple supineavoidance treatment alone. Whilst promising, successful reduction of snoring through simple supine-avoidance device treatment remains to be demonstrated.

Despite the effectiveness of continuous positive airway pressure (CPAP) for reducing AHI, patient acceptance and adherence to treatment remains problematic (Rotenberg et al., 2016) and there remains a remarkable lack of objective data regarding effects on snoring, one of the key reasons for which patients may seek treatment. Chapter 2 reported that supine-avoidance via a simple non-discomfort vibration-based alarm device is effective in reducing sleepiness and showed substantially greater treatment adherence compared to CPAP over 6-8 weeks of treatment. Given a lack of data concerning OSA treatment effects on snoring, the primary purpose of this study was to compare anticipated snoring frequency, intensity and snoring-related complaint reductions between supine-avoidance therapy compared to CPAP. A secondary aim

was to examine the prevalence of supine-predominant snoring in patients selected on the basis of supine-predominant OSA, and to test for potential differences in snoring and treatment adherence outcomes in patients with versus without co-existing supinepredominant snoring.

4.2 Method

Participants were from a previously reported randomised-controlled cross-over trial (see Chapter 2 – SUPA trial) selected on the basis of polysomnography (PSG) confirmed supine-predominant OSA defined on the basis of a total apnea-hypopnoea index (AHI) > 10 events per hour and a supine AHI at least twice that of non-supine AHI measured from at least 4 hours of recording with at least 30 min in supine and non-supine postures. All available data from the full SUPA cohort were examined including a subgroup of patients who were also classified as supine-predominant snorers based on a snoring frequency of at least twice that in supine compared to nonsupine positions. A total of 56 patients completed the SUPA trial and were available for this analysis. Following screening and consent, each participant completed a 1week in-home assessment of sleep posture (inactive supine-alarm device) and a full in-home PSG study at baseline prior to randomisation to receive either CPAP or supine-avoidance therapy as their initial treatment. Patients then remained on each allocated treatment for a total of 6-8 weeks before crossing over to the other treatment for a further 6-8 weeks, with a full in-home PSG on allocated treatment repeated within the last week of each treatment arm of the study.

Baseline and follow up questionnaires including Snoring Severity Scale (SSS) (Lim and Curry, 1999), assessment of quality of life (AQoL-8D) (Richardson et al., 2014) and Epworth Sleepiness Score (ESS) (Johns, 1992) were used to assess the effects of snoring on quality of life and sleep quality.

4.2.1 Supine-avoidance treatment

As described in more detail in Chapter 2, supine-avoidance treatment was achieved using a simple battery-operated supine-alarm device (Buzzpod, Gorman ProMed Pty. Ltd).

4.2.2 Sleep and snoring assessments

PSG studies (Embletta MPR, Pleasanton, California, United States) included all standard recording cannels according to the AASM rules. For further detail see Chapter 2.

All PSG signal data were exported to EDF for sleep scoring and analysis, including snoring assessment using Compumedics software (Profusion4, Compumedics, Melbourne Australia). The primary study outcome was based on discrete snore events detected algorithmically (Profusion4, Compumedics) and categorised as soft (≥50 to <60dBA), medium (≥60 to <70 dBA) or loud (≥70 dBA) (Bignold et al., 2011) from which snoring frequency per hour of sleep was calculated based on sleep time in each stage and posture for each category of snoring and overall for all events ≥50 dBA.

Treatment adherence was determined using device recorded usage time per night during the course of each treatment. Blood pressure was measured before bedtime on PSG nights using a standard automatic digital sphygmomanometer.

4.2.3 Statistical analysis

Analysis was performed on all available data from a previously reported randomised controlled trial (RCT) of supine-avoidance therapy vs CPAP (SUPA trial). Given the count-based nature of the primary snoring frequency outcome, negative binomial regression analysis was used to test for effects of treatment condition (baseline, supine-avoidance and CPAP), including treatment order, on snoring frequency and intensity (mild, moderate and severe) in-line with recommendations for zero inflated count data (Yau et al., 2003; Alexander, 2012). All other comparisons between conditions in normally distributed continuous outcomes were examined using linear mixed effects model analysis using an auto-regressive covariance structure and subjects as a random effect, each with their own intercept, to account for the repeated measures design and expected inter-individual variability. IBM SPSS (Version 25) was used to perform all statistical analysis. All data are reported as mean ± SD or median and interquartile range as specified in the results. p-values <0.05 were considered statistically significant.

4.3 Results

The characteristics of the study population are shown in Table 4.1. A total of 10 patients dropped out of the study, 5 patients during supine-avoidance and 5 during CPAP and one further patient had missing snoring data due to technical failure (faulty

microphone) so was excluded from further analysis. Of the 66 participants in the SUPA-OSA trial, 56 completed both treatment arms, and 15 (26.7%) met the predefined supine-predominant snoring criteria of at least double the snoring frequency in supine compared to non-supine sleep. The study sample was comprised of predominantly middle-aged overweight men, as is typical in an OSA clinic sample and, by study selection criteria, exhibited at least mild through to severe supine-predominant OSA.

During PSG at baseline, patients slept on their back for almost 30% of sleep and showed supine-predominant OSA consistent with the study entry criteria applied to the pre-study diagnostic PSG. There were no significant differences in any baseline characteristic between the supine-predominant snorer sub-group compared to the remaining group.

Parameter	Whole Group	Supine snorer	Non-supine snorer
N Total	64	15 (23.4%)	49 (76.5%)
N Females:Males (%)	23:41 (36:64%)	4:11 (27:73%)	19:30 (39:61%)
Age (yr)	52.8 ± 11.9	52.7 ± 10.9	54.6 ± 12.3
Height (cm)	171.9 ± 9.1	172.1 ± 9.2	171.9 ± 9.2
Weight (kg)	93.8 ± 21.4	87.3 ± 17.1	95.8 ± 22.3
Body mass index (kg/m²)	31.9 ± 7.8	29.7 ± 7.3	32.5 ± 7.9
Supine sleep time (%)	33.0 ± 24.6	45.4 ± 22.0	25.9 ± 22.0
Total AHI (/h)	14.2 ± 13.2	10.9 ± 10.5	13.4 ± 11.1
Supine AHI (/h)	25.4 ± 24.4	19.1 ± 14.8	24.5 ± 24.9
ESS	10.3 ± 3.9	9.9 ± 2.9	10.5 ± 4.2
AQoL – 8D	69.6 ± 13.6	74.1 ± 10.2	68.3 ± 14.3
SSS	6.1 ± 2.6	6.4 ± 2.6	6.0 ± 2.6
Systolic blood pressure (mmHg)	127.8 ± 12.9	128.4 ± 14.6	127.2 ± 13.2
Diastolic blood pressure (mmHg)	79.0 ± 9.5	81.1 ± 8.0	78.1 ± 9.9
Total snoring frequency (snores	48.9 [16.7 to	26.3 [17.0 to	57.5 [16.8 to
≥50 dBA/h).	188.7]	106.3]	190.7]

Table 4.1 Baseline characteristics of the study population.

Values are mean±SD or median [interquartile range]. Apnea hypopnea index (AHI), Epworth sleepiness scale (ESS). Snore severity scale (SSS), Assessment of Quality of Life (AQoL-8D).

Table 4.2 shows snoring frequency in the full study sample at baseline and during repeat PSGs in the last week of each treatment. Compared to baseline, supine sleep time was significantly reduced with supine-avoidance treatment (p<0.001) but remained unchanged with CPAP. There were significant treatment by posture (p<0.001), posture by snoring intensity (p=0.017) and treatment by snore category (p<0.001) effects on snoring frequency. Over the whole night there was a reduction in snoring frequency with CPAP (p<0.001) but not supine-avoidance treatment (p=0.651). CPAP also significantly reduced snoring frequency and intensity compared to baseline and supine-avoidance therapy, in both supine and non-supine sleep.

Treatment	Baseline (N=56)	Supine-avoidance (N=56)	CPAP (N=56)
Supine Sleep time (%)	30.7 [15.4 to 46.4]	0.2 [0.0 to 15.7]*^	36.7 [12.2 to 70.2]
AQoL – 8D	69.5 [64.4 to 77.8]	71.3 [61.2 to 81.6]	73.4 [64.7 to 81.0]
SSS	6.0 [3.5 to 8.0]	5.5 [3.8 to 7.0]	6.0 [4.0 to 8.0]
Soft (≥50, <60 dBA)	27.0 [7.7 to 87.8]	14.7 [3.8 to 99.2]	1.9 [0.7 to 5.7]*
Medium (≥60, <70 dBA)	6.1 [1.4 to 40.9]	4.4 [0.3 to 55.4]	0.6 [0.3 to 2.6]*
Loud (≥70 dBA)	4.5 [1.6 to 26.0]	4.6 [0.8 to 42.5]	1.2 [0.5 to 6.2]*
Total (≥50 dBA)	48.9 [16.7 to 188.7]	36.8 [6.3 to 233.7]	4.2 [2.1 to 29.5]*^
During Supine Sleep			
Soft (≥50, <60 dBA)	38.1 [10.9 to 79.2]	2.2 [0.0 to 19.1]*	3.5 [0.9 to 8.5] *
Medium (≥60, <70 dBA)	9.2 [3.4 to 33.0]	0.0 [0.0 to 13.6]	1.0 [0.0 to 3.3] *
Loud (≥70 dBA)	5.3 [1.8 to 30.8]	1.3 [0.0 to 16.1]*	1.1 [0.0 to 12.1] *
Total (≥50 dBA)	66.3 [22.9 to 186.7]	13.2 [0.7 to 67.1]*	8.5 [1.6 to 44.3] *
During Non-Supine Sleep			
Soft (≥50, <60 dBA)	19.8 [2.3 to 71.3]	13.7 [1.1 to 100.3]	1.5 [0.6 to 4.5] *
Medium (≥60, <70 dBA)	3.4 [0.6 to 31.1]	2.6 [0.3 to 54.0]	0.5 [0.0 to 1.4] *
Loud (≥70 dBA)	3.4 [0.8 to 29.8]	2.4 [0.7 to 42.5]*	1.2 [0.5 to 4.8] *
Total (≥50 dBA)	33.8 [8.7 to 190.9]	36.8 [4.6 to 231.9]	4.0 [1.8 to 14.7] *^

Table 4.2 Snoring frequency (snores/hour) of snoring by treatment group and snoring intensity separated by sleep posture and whole group.

Values are median [interquartile range], N=56. * indicates p<0.05 vs baseline, ^ p<0.05 vs CPAP.Snore severity scale (SSS), Assessment of Quality of Life (AQoL-8D).

Figure 4.1 Average hours of use per night.



Median and inter-quartile ranges, 10-90% whiskers, X indicates group mean over 6-8 weeks of treatment with supine-avoidance compared to CPAP in the whole group (N=56) and in supine-predominant snorers (N=15) compared to the remainder (N=41) of the group.

Table 4.3 shows snoring frequency findings in the sub-group of participants with supine-predominant snoring expected to benefit most from supine-avoidance therapy. As in the full study sample this sub-group showed a reduction in supine time with supine-avoidance but not CPAP treatment. However, in contrast to the full group, these participants showed a significant reduction in total snoring frequency (treatment effect p<0.001) and a shift to less severe snoring with both CPAP and supine-avoidance therapy (posture effect p<0.001, snoring intensity p<0.001). However, snoring reductions were larger and more consistent with CPAP compared to supine-avoidance which, unlike CPAP (p=0.026), did not significantly reduce soft and medium snoring. In addition, compared to baseline, non-supine snoring was significantly reduced with CPAP (p<0.001) but not supine-avoidance (p=0.299). Treatment type had no significant effect on AQoL or SSS scores in either the whole group (p=0.294 and 0.327, respectively) or the supine snorer sub-group (p=0.345 and p=0.896, respectively).

Treatment	Baseline (N=15)	Supine-avoidance (N=15)	CPAP (N=15)
Supine Sleep time (%)	42.0 [35.3 to 54.3]	12.2 [0.0 to 22.6]*^	32.6 [27.5 to 40.6]
AQoL – 8D	75.2 [68.4 to 81.6]	70.2 [65.2 to 85.8]	78.0 [64.2 to 82.3]
SSS	6.5 [4.0 to 7.8]	6.0 [5.0 to 7.0]	7.0 [4.0 to 7.5]
Soft (≥50, <60 dBA)	22.9 [12.8 to 71.6]	11.4 [2.1 to 15.0]	1.4 [0.7 to 3.7]*
Medium (≥60, <70 dBA)	4.5 [2.4 to 15.9]	0.9 [0.2 to 2.6]	0.3 [0.2 to 0.9]*
Loud (≥70 dBA)	2.7 [1.5 to 4.6]	0.9 [0.3 to 3.0]	2.4 [0.7 to 3.7]
Total (≥50 dBA)	26.3 [17.0 to 106.3]	14.3 [3.0 to 18.1]*	4.6 [2.6 to 8.9]*
During Supine Sleep			
Soft (≥50, <60 dBA)	48.7 [19.5 to 139.2]	7.8 [0.7 to 13.3] *	4.8 [1.8 to 5.5] *
Medium (≥60, <70 dBA)	8.5 [5.2 to 43.9]	1.5 [0.0 to 3.0] *	0.9 [0.4 to 1.1] *
Loud (≥70 dBA)	3.9 [1.8 to 7.8]	2.1 [0.3 to 13.4] *	1.8 [0.4 to 3.5] *
Total (≥50 dBA)	57.8 [33.0 to 202.5]	11.1 [1.6 to 29.4] *	9.8 [4.4 to 10.2] *
During Non-Supine Sleep			
Soft (≥50, <60 dBA)	4.2 [2.0 to 11.5]	5.9 [0.6 to 11.6]	0.7 [0.4 to 1.2]
Medium (≥60, <70 dBA)	0.9 [0.3 to 1.7]	0.5 [0.2 to 1.4]	0.3 [0.2 to 0.4]
Loud (≥70 dBA)	0.9 [0.3 to 3.0]	0.9 [0.3 to 1.3] *	2.9 [0.8 to 5.1]
Total (≥50 dBA)	8.3 [3.3 to 17.1]	11.4 [3.3 to 17.1]	4.3 [2.4 to 5.7] *

Table 4.3 Snoring frequency at baseline and on supine-avoidance compared to CPAP treatment in patients with both supine-predominant OSA and supine-predominant snoring.

Values are median [interquartile range], N=15. * indicates p<0.05 vs baseline, ^ p<0.05 vs CPAP. Snore severity scale (SSS), Assessment of Quality of Life (AQoL-8D).

4.4 Discussion

This is the first study to assess supine-avoidance compared to CPAP treatment effects on objectively recorded snoring frequency and intensity in supine-predominant OSA patients, and in a sub-group of participants who also exhibited supine-predominant snoring. These are amongst the most relevant patient groups for targeting supineavoidance device treatments. This is also one of very few studies to have examined the effects of CPAP on snoring, one of the most common symptoms and complaints for which patients seek sleep physician assessment and treatment (Krieger, 1992; Guzman et al., 2017).

By study design patients were selected on the basis of supine-predominant OSA, but with a normal non-supine AHI; the OSA patient group most likely to benefit from supine-avoidance treatment. In this group, successful supine-avoidance should theoretically normalise the AHI and OSA specific symptomatology. However, although snoring and OSA are clear signs of a collapsible airway in sleep (Gleadhill et al., 1991), the frequency of transient partial or complete airway obstruction events, measured on the basis of AHI, may show little if any relationship with the frequency and intensity of snoring. Thus, problematic snoring could potentially occur throughout much of the sleep period despite a normal AHI and potentially despite treatments designed to improve airway function. In this study, only 23% of supine-predominant OSA participants also exhibited supine-predominant snoring, with markedly reduced (by a factor of at least 2) snoring in non-supine sleep. Thus, successful supine-avoidance in individuals with supine-predominant OSA clearly does not necessarily translate into

parallel improvements in snoring. This is an important finding and supports that more specific attention to snoring measurements and outcomes is clearly warranted in patients attending sleep clinic services, particularly when patients attend clinic on the basis of bed-partner reported problem snoring. Nevertheless, snoring may be a strong sign of OSA, and given the importance of gravitational effects on the upper airway (Bilston and Gandevia, 2014), supine-avoidance is clearly worth considering as a treatment option in patients with supine-predominant OSA or with a history of supinepredominant snoring as the primary complaint.

By pneumatically splinting the upper airway to render partial and complete airway collapse less likely, CPAP might be expected to largely abolish snoring altogether. Whilst CPAP clearly did markedly reduce snoring in this patient group, some residual snoring remained, including loud snoring, particularly in supine sleep. Residual snoring potentially reflects periods of mask leak, and more challenging periods during the night such as supine and REM sleep where higher pressures could potentially achieve further reductions in snoring. CPAP was clearly superior compared to supine-avoidance at reducing snoring. However, in the sub-group of patients with supine-predominant snoring, support the need for careful patient-selection and consideration of primary symptom complaints before selecting supine-avoidance versus CPAP treatment. Of further note is the finding of considerably greater treatment adherence with supine-avoidance compared to CPAP treatment, but not in the sub-group of patients with supine-predominant snoring. Whilst these findings are potentially confounded by multiple factors, including issues of statistical power given sub-group

analysis, these findings are strongly suggestive of important clinical trade-offs between patient acceptance, comfort and perceived symptom relief.

There are no widely accepted standards for assessing and defining problematic snoring. Given that classic definitions of supine-predominant OSA are based on an at least 2:1 ratio of supine vs non-supine respiratory events (Cartwright et al., 1985), and perhaps clinically more usefully include a normal AHI in non-supine sleep (Mador et al., 2005a), we used a similar 2:1 ratio basis to define supine-predominant snoring. This group could potentially benefit most from supine–avoidance, particularly when snoring is a primary complaint and given substantially poorly CPAP compared to supine-avoidance use. However, what constitutes problematic snoring remains very poorly defined and alternative definitions with a more specific focus around bed-partner complaints and what might constitute acceptable residual snoring appear to be needed. Furthermore, in supine-predominant snorers, treatment adherence was not different compared to CPAP despite significant reductions in snoring, supporting the need for further studies to clarify what problems are most concerning to patients and/or their partners, balanced against treatment choices most likely to be efficacious and to be accepted and used by patients.

Given that this study analysed snoring in patients with supine-predominant OSA, it is possible that outcomes may be different in participants with simple snoring alone without OSA. Alternative supine-avoidance device approaches might also achieve different outcomes (Ha et al., 2014; Levendowski et al., 2015; Levendowski et al., 2014), although similar outcomes appear likely as long as the selected approach
effectively achieves supine-avoidance with minimal discomfort and sleep disruption. Nevertheless, improved outcomes might be possible through smarter devices able to detect and more specifically respond to posture-dependent respiratory events or problem snoring without unnecessary alarm events during wake or quiet supine sleep without respiratory events that could potentially negatively impact patient acceptance and use.

In summary, this study showed that simple vibration alarm based supine-avoidance treatment in patients with supine-predominant OSA does not necessarily translate into systematic reductions in snoring in sleep clinic patients with supine-predominant OSA. Although CPAP clearly achieves superior OSA and snoring reduction outcomes, patient use over 6-8 weeks of treatment is substantially inferior, and some residual snoring remained even with CPAP. Yet in patients with both supine-predominant OSA and supine-predominant snoring, supine-avoidance not unexpectedly achieves relatively effective treatment outcomes comparable to CPAP, including comparable treatment usage. Thus, there appear to be important and potentially quite complex trade-offs between patient characteristics and treatment outcomes for which careful treatment selection is clearly warranted. These data support the principles of precision medicine and that in well-selected patients, supine-avoidance is a simple, viable and low-cost approach well-suited to patients with supine-predominant OSA and/or snoring. Thus, comfortable non-mask based treatments better targeted to specific abnormalities such as supine-dependant snoring could help to manage a significant fraction of the community burden of sleep breathing disorders, and help to achieve improved long-term treatment outcomes which are often poor.

The overall objective of this thesis was to advance new knowledge regarding a very simple but potentially clinically useful treatment option for patients with supinepredominant sleep apnea for which objective data on treatment efficacy, adherence, effectiveness and symptom relief were previously lacking. Although it is already well known that supine sleep significantly impacts OSA severity and that one of the simplest approaches to treat OSA is to avoid sleeping supine, very poor treatment acceptance and use of traditional discomfort-based supine-avoidance treatment approaches render this approach non-viable. Thus, avoiding supine sleep based on physician's recommendations, from almost a century ago, to attach a bulky object to person's back (e.g. tennis ball treatment) to discourage supine sleep is too inherently uncomfortable to achieve effective treatment outcomes for patients long-term. The introduction of more recent new generation non-discomfort based vibratory supine alarm devices show significant promise in this area and could potentially represent a clinically effective, acceptable, easier to use and lower cost treatment alternative to CPAP in appropriately selected patients. However, given much poorer treatment acceptance and use of traditional TBT therapy, the hypothesis that new generation devices can achieve non-inferior treatment outcomes compared to CPAP clearly requires sufficient high quality evidence before any new form of treatment can become a recommended treatment for any specific sub-group of patients. Thus, the systematic investigation of supine-avoidance treatment outcomes with direct comparison to usual

care with CPAP within this thesis work makes a major contribution to the field of sleep medicine.

Very few studies have addressed key questions surrounding supine-avoidance therapy via new generation vibratory devices (Chapter 1). The work presented in this thesis has provided objective evidence to help fill important knowledge gaps regarding the comparative effectiveness of a simple vibratory supine alarm device compared to current standard treatment for OSA using CPAP.

The primary focus was to investigate whether supine-avoidance therapy via a vibratory alarm device is non-inferior compared to CPAP in reducing daytime sleepiness, the primary symptom complaint of greatest clinical concern in patients with OSA (Chapter 2). Through a randomised controlled cross-over trial design it was shown that the lower bound of the reduction in sleepiness following 6-8 weeks of supine-avoidance alarm device treatment was within a conservative predetermined margin of non-inferiority (ESS difference < -1.5) when compared CPAP. Adherence to therapy was also shown to be significantly and substantially (almost 2 hours per night) greater with supine-avoidance therapy such that the estimated overall effectiveness at reducing AHI was similar with both treatments (Chapter 2).

The following chapter (Chapter 3) examined the temporal relationship between EEG and posture shifts during sleep, specifically seeking to examine the frequency and duration of posture shifts, if overnight posture shifts are initiated from wake or following brief arousal and to examine what impact supine-avoidance alarms have on postalarm sleep onset latency. Supine sleep time was markedly reduced supporting the efficacy of supine-avoidance alarms for discouraging supine sleep, although there were occasional "sleep through" events followed by prolonged periods of supine sleep. Around 50-60% of overnight posture shifts were initiated from prior wake and the majority of posture shifts initiated from prior sleep were preceded by a brief arousal or awakening event. Consequently, most device alarm events appeared to occur very shortly after a period of established wake or brief cortical activation potentially needed for coordinated body movements associated with posture shifts themselves. In combination with the finding of no difference in the time-course of sleep onset following alarm events, these data support that supine-alarms do not interfere with the attainment of sleep in the post-alarm period (Chapter 3). Sleep onset following posture shifts to supine that were initiated from wake were also not different compared to a baseline sleep study without treatment or another night with CPAP treatment. These findings further support that supine-avoidance alarms do not interfere with the attainment of sleep following posture shifts from either sleep or prior wake.

In the final study (Chapter 4), snoring frequency and loudness changes with CPAP compared to supine-avoidance therapy were investigated in patients with supine-predominant OSA and a subgroup of patients who also showed supine-predominant snoring. This study showed that a simple supine-avoidance alarm device is less effective than CPAP in reducing snoring frequency, and that even with CPAP there is substantial residual snoring. However, there was a significant and likely to be a clinically more useful reduction in snoring in a subgroup of patients with both supine-predominant OSA and supine-predominant snoring. These data support that simple

supine-avoidance treatments are likely to be clinically useful in appropriately selected patients, and that greater attention to snoring outcomes and their supine-dependence is likely warranted in clinical practice.

One of the main limitations of all three experimental studies presented in this thesis was the lack of an internal clock in the supine avoidance device, which logged posture at approximately 1 Hz, but with somewhat variable sampling rates and thus "clock drift" over time between devices. This made it particularly difficult to reliably and accurately synchronise posture shift data recorded by the supine-avoidance device to home PSG data from which sleep quality and respiratory disturbances were primarily assessed. Several approaches were attempted to minimise uncertainty surrounding time synchronisation between devices, including manual logging of device button-push events against accurate local time when posture devices are setup and returned. This allowed for direct assessment and adjustment for variable sampling rates between devices. Custom algorithms were also developed to minimise reliance on subjective bedtime/sleep onset reported by patients using a sleep diary, although these provided very useful confirmatory data when patients reliably used them. For the most part these methods demonstrated high levels of agreement between supine-avoidance device versus PSG recorded posture data. However, given some further inevitable uncertainty regarding correct patient placement of supine-avoidance and PSG devices in the home environment a degree of uncertainty remains regarding posture recording and classification accuracy. Further potential limitations that warrant consideration include some lost data associated with home PSG, potential interference associated with bed-partner snoring and the inevitable open-label nature of these studies for which study blinding of inherently different interventions was not possible.

The novel data and analysis presented in this thesis make an important contribution to sleep medicine towards establishing the evidence-base needed to support direct clinical translation and uptake of new generation supine-avoidance alarm devices to treat supine-predominant OSA and potentially snoring. Nevertheless, several unanswered questions remain for which further studies remain needed, especially given prior work clearly demonstrating that assumptions regarding treatments which in theory would be expected to benefit patients may not be sufficiently acceptable or beneficial to patients for long-term use. Although other similar devices might be expected to show similar performance, the treatment effectiveness and patient acceptance and use may not necessarily be comparable between devices and deserve cautious interpretation and sufficiently favourable comparative treatment outcomes for all specific new device treatments. This work showed substantially superior supine-avoidance compared to CPAP acceptance and use over 6-8 weeks. Nevertheless, longer-term effectiveness and adherence to therapy remains unknown and will require longer-term comparative effectiveness and outcome studies.

The current generation of vibration based supine-avoidance alarm devices rely only on posture signals alone to trigger an alarm. A substantial fraction of posture shifts to the supine position occurred during established wake and potentially include nuisance alarms when the wearer is not necessarily attempting to sleep and may prefer to lay supine and awake without the need to intervene to suppress alarms. Furthermore, supine sleep does not necessarily always lead to respiratory events or snoring. Consequently, future work appears likely to benefit from signal integration towards a new generation of "smarter" supine-avoidance alarms triggered not simply from posture shifts alone, but with additional consideration of respiratory effort, snoring and/or potentially oxygen desaturation signals. Despite a strong vibration stimulus, not all alarm events resulted in supine-avoidance. Thus, further device improvements to minimise alarms that fail to effectively discourage supine sleep may also be warranted.

References

- Abdelghani, A., Slama, S., Hayouni, A., Harrabi, I., Mezghanni, S., Garrouche, A., Klabi, N., Benzarti, M. & Jerray, M. 2009. [Acceptance and long-term compliance to continuous positive airway pressure in obstructive sleep apnea. A prospective study on 72 patients treated between 2004 and 2007]. *Rev Pneumol Clin*, 65, 147-52.
- Adams, R., Appleton, S., Taylor, A., Mcevoy, D. & Wittert, G. 2016. Are the ICSD-3 criteria for sleep apnoea syndrome too inclusive? *Lancet Respir Med*, 4, e19-20.
- Adams, R. J., Appleton, S. L., Taylor, A. W., Gill, T. K., Lang, C., Mcevoy, R. D. & Antic, N. A. 2017. Sleep health of Australian adults in 2016: results of the 2016 Sleep Health Foundation national survey. *Sleep Health*, 3, 35-42.
- Ahbab, S., Ataoglu, H. E., Tuna, M., Karasulu, L., Cetin, F., Temiz, L. U. & Yenigun,
 M. 2013. Neck circumference, metabolic syndrome and obstructive sleep apnea syndrome; evaluation of possible linkage. *Med Sci Monit*, 19, 111-7.
- Alexander, N. 2012. Review: analysis of parasite and other skewed counts. *Tropical Medicine & International Health*, 17, 684-693.
- Amatoury, J., Howitt, L., Wheatley, J. R., Avolio, A. P. & Amis, T. C. 2006. Snoringrelated energy transmission to the carotid artery in rabbits. *J Appl Physiol* (1985), 100, 1547-53.
- Appleton, S. L., Vakulin, A., Martin, S. A., Lang, C. J., Wittert, G. A., Taylor, A. W., Mcevoy, R. D., Antic, N. A., Catcheside, P. G. & Adams, R. J. 2016. Hypertension Is Associated With Undiagnosed OSA During Rapid Eye Movement Sleep. *Chest*, 150, 495-505.
- Arnardottir, E. S., Isleifsson, B., Agustsson, J. S., Sigurdsson, G. A., Sigurgunnarsdottir, M. O., Sigurđarson, G. T., Saevarsson, G., Sveinbjarnarson, A. T., Hoskuldsson, S. & Gislason, T. 2016. How to measure snoring? A comparison of the microphone, cannula and piezoelectric sensor. *Journal of Sleep Research*, 25, 158-168.
- Bahammam, A. S., Alassiri, S. S., Al-Adab, A. H., Alsadhan, I. M., Altheyab, A. M., Alrayes, A. H., Alkhawajah, M. M. & Olaish, A. H. 2015. Long-term compliance with continuous positive airway pressure in Saudi patients with obstructive sleep apnea. A prospective cohort study. *Saudi Med J*, 36, 911-9.
- Bakker, J. P., Weaver, T. E., Parthasarathy, S. & Aloia, M. S. 2019. Adherence to CPAP: What Should We Be Aiming For, and How Can We Get There?: What Should We Be Aiming For, and How Can We Get There? *Chest*, 155, 1272-1287.

- Basyuni, S., Barabas, M. & Quinnell, T. 2018. An update on mandibular advancement devices for the treatment of obstructive sleep apnoea hypopnoea syndrome. *Journal of thoracic disease,* 10, S48-S56.
- Benjafield, A., Valentine, K., Ayas, N., Eastwood, P., Heinzer, R., Ip, M., Patel, S., Peppard, P., Sinha, S. & Tufik, S. 2018. Global prevalence of obstructive sleep apnea in adults: estimation using currently available data. *B67. Risk and prevalence of sleep disordered breathing.* American Thoracic Society.
- Benjafield, A. V., Ayas, N. T., Eastwood, P. R., Heinzer, R., Ip, M. S. M., Morrell, M. J., Nunez, C. M., Patel, S. R., Penzel, T., Pepin, J. L., Peppard, P. E., Sinha, S., Tufik, S., Valentine, K. & Malhotra, A. 2019. Estimation of the global prevalence and burden of obstructive sleep apnoea: a literature-based analysis. *Lancet Respir Med*, *7*, 687-698.
- Benumof, J. L. 2016. Supine Position, Sleep, Wet Airways, and Wet Lungs. *Anesth Analg*, 123, 1333.
- Berry, R. B., Budhiraja, R., Gottlieb, D. J., Gozal, D., Iber, C., Kapur, V. K., Marcus, C. L., Mehra, R., Parthasarathy, S., Quan, S. F., Redline, S., Strohl, K. P., Davidson Ward, S. L., Tangredi, M. M. & American Academy of Sleep, M. 2012. Rules for scoring respiratory events in sleep: update of the 2007 AASM Manual for the Scoring of Sleep and Associated Events. Deliberations of the Sleep Apnea Definitions Task Force of the American Academy of Sleep Medicine. *J Clin Sleep Med*, 8, 597-619.
- Berry, R. B., Uhles, M. L., Abaluck, B. K., Winslow, D. H., Schweitzer, P. K., Gaskins, R. A., Jr., Doekel, R. C., Jr. & Emsellem, H. A. 2019. NightBalance Sleep Position Treatment Device Versus Auto-Adjusting Positive Airway Pressure for Treatment of Positional Obstructive Sleep Apnea. *J Clin Sleep Med*, 15, 947-956.
- Bignold, J. J., Deans-Costi, G., Goldsworthy, M. R., Robertson, C. A., Mcevoy, D., Catcheside, P. G. & Mercer, J. D. 2009. Poor long-term patient compliance with the tennis ball technique for treating positional obstructive sleep apnea. *J Clin Sleep Med*, 5, 428-30.
- Bignold, J. J., Mercer, J. D., Antic, N. A., Mcevoy, R. D. & Catcheside, P. G. 2011. Accurate position monitoring and improved supine-dependent obstructive sleep apnea with a new position recording and supine avoidance device. *J Clin Sleep Med*, 7, 376-83.
- Bilston, L. E. & Gandevia, S. C. 2014. Biomechanical properties of the human upper airway and their effect on its behavior during breathing and in obstructive sleep apnea. *J Appl Physiol (1985),* 116, 314-24.
- Caples, S. M., Rowley, J. A., Prinsell, J. R., Pallanch, J. F., Elamin, M. B., Katz, S. G. & Harwick, J. D. 2010. Surgical modifications of the upper airway for obstructive

sleep apnea in adults: a systematic review and meta-analysis. *Sleep*, 33, 1396-407.

- Carberry, J. C., Fisher, L. P., Grunstein, R. R., Gandevia, S. C., Mckenzie, D. K., Butler, J. E. & Eckert, D. J. 2017. Role of common hypnotics on the phenotypic causes of obstructive sleep apnoea: paradoxical effects of zolpidem. *Eur Respir J*, 50.
- Carter, S. G., Carberry, J. C., Cho, G., Fisher, L. P., Rollo, C. M., Stevens, D. J., D'rozario, A. L., Mckenzie, D. K., Grunstein, R. R. & Eckert, D. J. 2018. Effect of 1 month of zopiclone on obstructive sleep apnoea severity and symptoms: a randomised controlled trial. *Eur Respir J*, 52.
- Cartwright, R., Ristanovic, R., Diaz, F., Caldarelli, D. & Alder, G. 1991. A comparative study of treatments for positional sleep apnea. *Sleep*, 14, 546-52.
- Cartwright, R. D., Lloyd, S., Lilie, J. & Kravitz, H. 1985. Sleep position training as treatment for sleep apnea syndrome: a preliminary study. *Sleep*, 8, 87-94.
- Carvalho, B., Hsia, J. & Capasso, R. 2012. Surgical therapy of obstructive sleep apnea: a review. *Neurotherapeutics : the journal of the American Society for Experimental NeuroTherapeutics*, 9, 710-716.
- Casey, K. R. 2015. Positional therapy is worth a try in patients with mild obstructive sleep apnea. Point-counterpoint: is avoidance of supine sleep an adequate treatment for OSA? *J Clin Sleep Med*, 11, 89-90.
- Casey, K. R. & Teodorescu, M. 2015. Postoperative Complications in Patients with Obstructive Sleep Apnea: Where Do We Stand? *J Clin Sleep Med*, 11, 1081-2.
- Catcheside, P. G. 2010. Predictors of continuous positive airway pressure adherence. *F1000 Med Rep*, 2.
- Chaudhary, B. A., Chaudhary, T. K., Kolbeck, R. C., Harmon, J. D. & Speir, W. A. 1986. Therapeutic effect of posture in sleep apnea. *Southern medical journal,* 79, 1061-1063.
- Chen, Y. F., Hang, L. W., Huang, C. S., Liang, S. J. & Chung, W. S. 2015. Polysomnographic predictors of persistent continuous positive airway pressure adherence in patients with moderate and severe obstructive sleep apnea. *Kaohsiung J Med Sci*, 31, 83-9.
- Cho Yong, W., Kim Keun, T., Moon, H.-J., Korostyshevskiy Valeriy, R., Motamedi Gholam, K. & Yang Kwang, I. 2018. Comorbid Insomnia With Obstructive Sleep Apnea: Clinical Characteristics and Risk Factors. *Journal of Clinical Sleep Medicine*, 14, 409-417.
- Chopra, S., Rathore, A., Younas, H., Pham, L. V., Gu, C., Beselman, A., Kim, I.-Y., Wolfe, R. R., Perin, J., Polotsky, V. Y. & Jun, J. C. 2017. Obstructive Sleep

Apnea Dynamically Increases Nocturnal Plasma Free Fatty Acids, Glucose, and Cortisol During Sleep. *The Journal of Clinical Endocrinology & Metabolism*, 102, 3172-3181.

- Cistulli, P. A., Armitstead, J., Pepin, J.-L., Woehrle, H., Nunez, C. M., Benjafield, A. & Malhotra, A. 2019. Short-term CPAP adherence in obstructive sleep apnea: a big data analysis using real world data. *Sleep medicine*, *5*9, 114-116.
- Clemente, J., Valero, M., Li, F., Wang, C. & Song, W. Helena: Real-time Contact-free Monitoring of Sleep Activities and Events around the Bed. 2020 IEEE International Conference on Pervasive Computing and Communications (PerCom), 23-27 March 2020 2020. 1-10.
- Curry, B. D., Bain, J. L. W., Yan, J.-G., Zhang, L. L., Yamaguchi, M., Matloub, H. S. & Riley, D. A. 2002. Vibration injury damages arterial endothelial cells. *Muscle & Nerve*, 25, 527-534.
- Dafna, E., Tarasiuk, A. & Zigel, Y. 2014. Automatic Detection of Whole Night Snoring Events Using Non-Contact Microphone. *PLOS ONE,* 8, e84139.
- Danker-Hopfe, H., Anderer, P., Zeitlhofer, J., Boeck, M., Dorn, H., Gruber, G., Heller, E., Loretz, E., Moser, D., Parapatics, S., Saletu, B., Schmidt, A. & Dorffner, G. 2009. Interrater reliability for sleep scoring according to the Rechtschaffen & Kales and the new AASM standard. *Journal of Sleep Research*, 18, 74-84.
- De Geest, S. & Sabate, E. 2003. Adherence to long-term therapies: evidence for action. *Eur J Cardiovasc Nurs*, 2, 323.
- De Koninck, J., Gagnon, P. & Lallier, S. 1983. Sleep positions in the young adult and their relationship with the subjective quality of sleep. *Sleep*, 6, 52-9.
- Deacon, N. L., Jen, R., Li, Y. & Malhotra, A. 2016. Treatment of Obstructive Sleep Apnea. Prospects for Personalized Combined Modality Therapy. *Ann Am Thorac Soc,* 13, 101-8.
- Debacker, W. A., Verbraecken, J., Willemen, M., Wittesaele, W., Decock, W. & Van Deheyning, P. 1995. Central apnea index decreases after prolonged treatment with acetazolamide. *Am J Respir Crit Care Med*, 151, 87-91.
- Dempsey, J. A., Veasey, S. C., Morgan, B. J. & O'donnell, C. P. 2010. Pathophysiology of sleep apnea. *Physiol Rev*, 90, 47-112.
- Dieltjens, M., Vroegop, A. V., Verbruggen, A. E., Wouters, K., Willemen, M., De Backer, W. A., Verbraecken, J. A., Van De Heyning, P. H., Braem, M. J., De Vries, N. & Vanderveken, O. M. 2015. A promising concept of combination therapy for positional obstructive sleep apnea. *Sleep Breath*, 19, 637-44.

- Dingli, K., Coleman, E. L., Vennelle, M., Finch, S. P., Wraith, P. K., Mackay, T. W. & Douglas, N. J. 2003. Evaluation of a portable device for diagnosing the sleep apnoea/hypopnoea syndrome. *European Respiratory Journal*, 21, 253.
- Dopp, J. M., Reichmuth, K. J. & Morgan, B. J. 2007. Obstructive sleep apnea and hypertension: mechanisms, evaluation, and management. *Curr Hypertens Rep,* 9, 529-34.
- Drager, L. F., Malhotra, A., Yan, Y., Pépin, J. L., Armitstead, J. P., Woehrle, H., Nunez, C. M., Cistulli, P. A. & Benjafield, A. V. 2021. Adherence with positive airway pressure therapy for obstructive sleep apnea in developing vs. developed countries: a big data study. *J Clin Sleep Med*, 17, 703-709.
- Dredla, B. K. & Castillo, P. R. 2019. Cardiovascular Consequences of Obstructive Sleep Apnea. *Curr Cardiol Rep*, 21, 137.
- Eastwood, P. R., Barnes, M., Mackay, S. G., Wheatley, J. R., Hillman, D. R., Nguyen, X. L., Lewis, R., Campbell, M. C., Petelle, B., Walsh, J. H., Jones, A. C., Palme, C. E., Bizon, A., Meslier, N., Bertolus, C., Maddison, K. J., Laccourreye, L., Raux, G., Denoncin, K., Attali, V., Gagnadoux, F. & Launois, S. H. 2020. Bilateral hypoglossal nerve stimulation for treatment of adult obstructive sleep apnoea. *Eur Respir J*, 55.
- Eckert, D. J. 2016. Phenotypic approaches to obstructive sleep apnoea New pathways for targeted therapy. *Sleep Med Rev*.
- Eckert, D. J., Owens, R. L., Kehlmann, G. B., Wellman, A., Rahangdale, S., Yim-Yeh, S., White, D. P. & Malhotra, A. 2011. Eszopiclone increases the respiratory arousal threshold and lowers the apnoea/hypopnoea index in obstructive sleep apnoea patients with a low arousal threshold. *Clin Sci (Lond)*, 120, 505-14.

Economics, D. A. 2017. Asleep on the Job: Costs of inadequate sleep in Australia. Australia: Sleep Health Foundation.

- Edwards, B. A., Eckert, D. J. & Jordan, A. S. 2017. Obstructive sleep apnoea pathogenesis from mild to severe: Is it all the same? *Respirology*, 22, 33-42.
- Edwards, B. A., Sands, S. A., Eckert, D. J., White, D. P., Butler, J. P., Owens, R. L., Malhotra, A. & Wellman, A. 2012. Acetazolamide improves loop gain but not the other physiological traits causing obstructive sleep apnoea. *J Physiol*, 590, 1199-211.
- Edwards, B. A., Wellman, A., Sands, S. A., Owens, R. L., Eckert, D. J., White, D. P. & Malhotra, A. 2014. Obstructive sleep apnea in older adults is a distinctly different physiological phenotype. *Sleep*, *37*, 1227-36.
- Eijsvogel, M. M., Ubbink, R., Dekker, J., Oppersma, E., De Jongh, F. H., Van Der Palen, J. & Brusse-Keizer, M. G. 2015. Sleep position trainer versus tennis ball technique in positional obstructive sleep apnea syndrome. *Journal of clinical*

sleep medicine : JCSM : official publication of the American Academy of Sleep Medicine, 11, 139.

- Ejaz, S. M., Khawaja, I. S., Bhatia, S. & Hurwitz, T. D. 2011. Obstructive sleep apnea and depression: a review. *Innov Clin Neurosci,* 8, 17-25.
- Engleman, H. M. & Wild, M. R. 2003a. Improving CPAP use by patients with the sleep apnoea/hypopnoea syndrome (SAHS). *Sleep Med Rev*, 7, 81-99.
- Engleman, H. M. & Wild, M. R. 2003b. Improving CPAP use by patients with the sleep apnoea/hypopnoea syndrome (SANS). *Sleep Medicine Reviews*, 7, 81-99.
- Enright, P. L., Newman, A. B., Wahl, P. W., Manolio, T. A., Haponik, E. F. & Boyle, P. J. 1996. Prevalence and correlates of snoring and observed apneas in 5,201 older adults. *Sleep*, 19, 531-8.
- Fan, J., Wang, X., Ma, X., Somers, V. K., Nie, S. & Wei, Y. 2019. Association of Obstructive Sleep Apnea With Cardiovascular Outcomes in Patients With Acute Coronary Syndrome. J Am Heart Assoc, 8, e010826.
- Ferrer-Lluis, I., Castillo-Escario, Y., Montserrat, J. M. & Jané, R. 2021. Enhanced Monitoring of Sleep Position in Sleep Apnea Patients: Smartphone Triaxial Accelerometry Compared with Video-Validated Position from Polysomnography. Sensors (Basel), 21.
- Frank, M. H., Ravesloot, M. J., Van Maanen, J. P., Verhagen, E., De Lange, J. & De Vries, N. 2015. Positional OSA part 1: Towards a clinical classification system for position-dependent obstructive sleep apnoea. *Sleep Breath*, 19, 473-80.
- Franklin, K. A., Anttila, H., Axelsson, S., Gislason, T., Maasilta, P., Myhre, K. I. & Rehnqvist, N. 2009. Effects and side-effects of surgery for snoring and obstructive sleep apnea--a systematic review. *Sleep*, 32, 27-36.
- Friedman, M., Jacobowitz, O., Hwang, M. S., Bergler, W., Fietze, I., Rombaux, P., Mwenge, G. B., Yalamanchali, S., Campana, J. & Maurer, J. T. 2016. Targeted hypoglossal nerve stimulation for the treatment of obstructive sleep apnea: Sixmonth results. *Laryngoscope*, 126, 2618-2623.
- Gagnon, C., Belanger, L., Ivers, H. & Morin, C. M. 2013. Validation of the Insomnia Severity Index in primary care. *J Am Board Fam Med*, 26, 701-10.
- Giles, T. L., Lasserson, T. J., Smith, B. J., White, J., Wright, J. & Cates, C. J. 2006. Continuous positive airways pressure for obstructive sleep apnoea in adults. *Cochrane Database Syst Rev*, Cd001106.
- Gleadhill, I. C., Schwartz, A. R., Schubert, N., Wise, R. A., Permutt, S. & Smith, P. L. 1991. Upper airway collapsibility in snorers and in patients with obstructive hypopnea and apnea. *Am Rev Respir Dis*, 143, 1300-3.

- Gori, S., Ficca, G., Giganti, F., Di Nasso, I., Murri, L. & Salzarulo, P. 2004. Body movements during night sleep in healthy elderly subjects and their relationships with sleep stages. *Brain Res Bull*, 63, 393-7.
- Goyal, A., Agarwal, N. & Pakhare, A. 2017. Barriers to CPAP Use in India: An Exploratory Study. *Journal of clinical sleep medicine : JCSM : official publication of the American Academy of Sleep Medicine,* 13, 1385-1394.
- Greneche, J., Krieger, J., Bertrand, F., Erhardt, C., Maumy, M. & Tassi, P. 2013. Effect of continuous positive airway pressure treatment on short-term memory performance over 24 h of sustained wakefulness in patients with obstructive sleep apnea-hypopnea syndrome. *Sleep Med*, 14, 964-72.
- Guilleminault, C., Poyares, D., Rosa, A. & Huang, Y. S. 2005. Heart rate variability, sympathetic and vagal balance and EEG arousals in upper airway resistance and mild obstructive sleep apnea syndromes. *Sleep Med*, 6, 451-7.
- Guzman, M. A., Sgambati, F. P., Pho, H., Arias, R. S., Hawks, E. M., Wolfe, E. M., Ötvös, T., Rosenberg, R., Dakheel, R., Schneider, H., Kirkness, J. P., Smith, P. L. & Schwartz, A. R. 2017. The Efficacy of Low-Level Continuous Positive Airway Pressure for the Treatment of Snoring. *J Clin Sleep Med*, 13, 703-711.
- Ha, S. C., Hirai, H. W. & Tsoi, K. K. 2014. Comparison of positional therapy versus continuous positive airway pressure in patients with positional obstructive sleep apnea: a meta-analysis of randomized trials. *Sleep Med Rev,* 18, 19-24.
- Hawkins, R. B., Mehaffey, J. H., Mcmurry, T. L., Kirby, J., Malin, S. K., Schirmer, B. & Hallowell, P. T. 2017. Clinical significance of failure to lose weight 10 years after roux-en-y gastric bypass. *Surg Obes Relat Dis*, 13, 1710-1716.
- Heinzer, R., Petitpierre, N. J., Marti-Soler, H. & Haba-Rubio, J. 2018. Prevalence and characteristics of positional sleep apnea in the HypnoLaus population-based cohort. *Sleep Med*, 48, 157-162.
- Heinzer, R., Vat, S., Marques-Vidal, P., Marti-Soler, H., Andries, D., Tobback, N., Mooser, V., Preisig, M., Malhotra, A., Waeber, G., Vollenweider, P., Tafti, M. & Haba-Rubio, J. 2015. Prevalence of sleep-disordered breathing in the general population: the HypnoLaus study. *Lancet Respir Med*, 3, 310-8.
- Hidalgo Armas, L., Turino, C., Cordero-Guevara, J., Manjón, J. L., Durán-Carro, J., Barbé, F., Vaca, R. & Durán-Cantolla, J. 2019. A new postural device for the treatment of positional obstructive sleep apnea. A pilot study. *Respir Med*, 151, 111-117.
- Hillman, D. R., Murphy, A. S. & Pezzullo, L. 2006. The economic cost of sleep disorders. *Sleep*, 29, 299-305.

- Hofauer, B., Steffen, A., Knopf, A., Hasselbacher, K. & Heiser, C. 2019. Patient experience with upper airway stimulation in the treatment of obstructive sleep apnea. *Sleep Breath*, 23, 235-241.
- Hoffstein, V., Mateika, S. & Nash, S. 1996. Comparing perceptions and measurements of snoring. *Sleep*, 19, 783-9.
- Holty, J. E. & Guilleminault, C. 2010. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev,* 14, 287-97.
- Hudgel, D. W. 2016. Sleep Apnea Severity Classification Revisited. *Sleep*, 39, 1165-6.
- Iber, C., Ancoli-Israel, S., Chesson, A. & Quan, S., F. 2007. The AASM manual for the scoring of sleep and associated events: rules, terminology, and technical specification. *American Academy of Sleep Medicine*. Westchester, IL.
- Imayama, I. & Prasad, B. 2017. Role of Leptin in Obstructive Sleep Apnea. *Ann Am Thorac Soc,* 14, 1607-1621.
- Javaheri, S. 2006. Acetazolamide improves central sleep apnea in heart failure: a double-blind, prospective study. *Am J Respir Crit Care Med*, 173, 234-7.
- Jeng, P.-Y., Wang, L.-C., Hu, C.-J. & Wu, D. 2021. A Wrist Sensor Sleep Posture Monitoring System: An Automatic Labeling Approach. *Sensors (Basel, Switzerland)*, 21, 258.
- Johal, A. M. A., Patel, S. I. & Battagel, J. M. 2007. The relationship between craniofacial anatomy and obstructive sleep apnoea: a case-controlled study. *Journal of Sleep Research*, 16, 319-326.
- Johns, M. W. 1992. Reliability and factor analysis of the Epworth Sleepiness Scale. *Sleep*, 15, 376-81.
- Jokic, R., Klimaszewski, A., Crossley, M., Sridhar, G. & Fitzpatrick, M. F. 1999. Positional treatment vs continuous positive airway pressure in patients with positional obstructive sleep apnea syndrome. *Chest*, 115, 771-81.
- Jones, T. M. & Swift, A. C. 2000. Snoring: recent developments. *Hosp Med*, 61, 330-5.
- Joosten, S. A., Edwards, B. A., Wellman, A., Turton, A., Skuza, E. M., Berger, P. J. & Hamilton, G. S. 2015. The Effect of Body Position on Physiological Factors that Contribute to Obstructive Sleep Apnea. *Sleep*, 38, 1469-78.
- Joosten, S. A., Hamilton, G. S. & Naughton, M. T. 2017. Impact of Weight Loss Management in OSA. *Chest*, 152, 194-203.

- Joosten, S. A., O'donoghue, F. J., Rochford, P. D., Barnes, M., Hamza, K., Churchward, T. J., Berger, P. J. & Hamilton, G. S. 2014a. Night-to-night repeatability of supine-related obstructive sleep apnea. *Ann Am Thorac Soc*, 11, 761-9.
- Joosten, S. A., O'driscoll, D. M., Berger, P. J. & Hamilton, G. S. 2014b. Supine position related obstructive sleep apnea in adults: pathogenesis and treatment. *Sleep Med Rev*, 18, 7-17.
- Kavey, N. B., Blitzer, A., Gidro-Frank, S. & Korstanje, K. 1985. Sleeping position and sleep apnea syndrome. *Am J Otolaryngol, 6*, 373-7.
- Kent, D. T., Lee, J. J., Strollo, P. J., Jr. & Soose, R. J. 2016. Upper Airway Stimulation for OSA: Early Adherence and Outcome Results of One Center. *Otolaryngol Head Neck Surg*, 155, 188-93.
- Kezirian, E. J., Goding, G. S., Jr., Malhotra, A., O'donoghue, F. J., Zammit, G., Wheatley, J. R., Catcheside, P. G., Smith, P. L., Schwartz, A. R., Walsh, J. H., Maddison, K. J., Claman, D. M., Huntley, T., Park, S. Y., Campbell, M. C., Palme, C. E., Iber, C., Eastwood, P. R., Hillman, D. R. & Barnes, M. 2014. Hypoglossal nerve stimulation improves obstructive sleep apnea: 12-month outcomes. *J Sleep Res*, 23, 77-83.
- Kezirian, E. J. & Goldberg, A. N. 2006. Hypopharyngeal surgery in obstructive sleep apnea: an evidence-based medicine review. Arch Otolaryngol Head Neck Surg, 132, 206-13.
- Kim, A. M., Keenan, B. T., Jackson, N., Chan, E. L., Staley, B., Poptani, H., Torigian, D. A., Pack, A. I. & Schwab, R. J. 2014. Tongue fat and its relationship to obstructive sleep apnea. *Sleep*, 37, 1639-48.
- Kim, K. T., Cho, Y. W., Kim, D. E., Hwang, S. H., Song, M. L. & Motamedi, G. K. 2016. Two subtypes of positional obstructive sleep apnea: Supine-predominant and supine-isolated. *Clinical Neurophysiology*, 127, 565-570.
- Kneisley, L. W. 1998. Medical and nondental treatments of snoring and sleep apnea syndrome. *J Calif Dent Assoc,* 26, 572-8.
- Kompelli, A. R., Ni, J. S., Nguyen, S. A., Lentsch, E. J., Neskey, D. M. & Meyer, T. A. 2019. The outcomes of hypoglossal nerve stimulation in the management of OSA: A systematic review and meta-analysis. *World J Otorhinolaryngol Head Neck Surg*, 5, 41-48.
- Kribbs, N. B., Pack, A. I., Kline, L. R., Smith, P. L., Schwartz, A. R., Schubert, N. M., Redline, S., Henry, J. N., Getsy, J. E. & Dinges, D. F. 1993. Objective measurement of patterns of nasal CPAP use by patients with obstructive sleep apnea. *Am Rev Respir Dis*, 147, 887-95.

- Krieger, J. 1992. Long-term compliance with nasal continuous positive airway pressure (CPAP) in obstructive sleep apnea patients and nonapneic snorers. *Sleep*, 15, S42-6.
- Kubin, L. 2016. Neural Control of the Upper Airway: Respiratory and State-Dependent Mechanisms. *Comprehensive Physiology*, 6, 1801-1850.
- Larsson, L.-G., Lindberg, A., Franklin, K. A. & Lundba[¬]Ck, B. 2003. Gender Differences in Symptoms Related to Sleep Apnea in a General Population and in Relation to Referral to Sleep Clinic. *Chest*, 124, 204-211.
- Latshang, T. D., Nussbaumer-Ochsner, Y., Henn, R. M., Ulrich, S., Lo Cascio, C. M., Ledergerber, B., Kohler, M. & Bloch, K. E. 2012. Effect of acetazolamide and autoCPAP therapy on breathing disturbances among patients with obstructive sleep apnea syndrome who travel to altitude: a randomized controlled trial. *JAMA*, 308, 2390-8.
- Laub, R. R., Mikkelsen, K. L. & Tønnesen, P. 2015. Prevalence of positional obstructive sleep apnea and patients characteristics using various definitions. *European Respiratory Journal*, 46, PA2372.
- Laub, R. R., Tonnesen, P. & Jennum, P. J. 2017. A Sleep Position Trainer for positional sleep apnea: a randomized, controlled trial. *J Sleep Res*.
- Lee, C. H. K., Leow, L. C., Song, P. R., Li, H. & Ong, T. H. 2017. Acceptance and Adherence to Continuous Positive Airway Pressure Therapy in patients with Obstructive Sleep Apnea (OSA) in a Southeast Asian privately funded healthcare system. *Sleep science (Sao Paulo, Brazil)*, 10, 57-63.
- Lee, S. A., Amis, T. C., Byth, K., Larcos, G., Kairaitis, K., Robinson, T. D. & Wheatley, J. R. 2008. Heavy snoring as a cause of carotid artery atherosclerosis. *Sleep*, 31, 1207-13.
- Levendowski, D. J., Seagraves, S., Popovic, D. & Westbrook, P. R. 2014. Assessment of a neck- based treatment and monitoring device for positional obstructive sleep apnea. *Journal of clinical sleep medicine : JCSM : official publication of the American Academy of Sleep Medicine,* 10, 863.
- Levendowski, D. J., Veljkovic, B., Seagraves, S. & Westbrook, P. R. 2015. Capability of a neck worn device to measure sleep/wake, airway position, and differentiate benign snoring from obstructive sleep apnea. *J Clin Monit Comput*, 29, 53-64.
- Lim, P. V. & Curry, A. R. 1999. A new method for evaluating and reporting the severity of snoring. *J Laryngol Otol,* 113, 336-40.
- Liu, J. J., Xu, W., Huang, M., Alshurafa, N., Sarrafzadeh, M., Raut, N. & Yadegar, B. A dense pressure sensitive bedsheet design for unobtrusive sleep posture monitoring. 2013 IEEE International Conference on Pervasive Computing and Communications (PerCom), 18-22 March 2013 2013. 207-215.

- Loord, H. & Hultcrantz, E. 2007. Positioner--a method for preventing sleep apnea. *Acta Otolaryngol,* 127, 861-8.
- Lundin, A., Hallgren, M., Theobald, H., Hellgren, C. & Torgén, M. 2016. Validity of the 12-item version of the General Health Questionnaire in detecting depression in the general population. *Public Health*, 136, 66-74.
- Mackay, S. G., Carney, A. S., Woods, C., Antic, N., Mcevoy, R. D., Chia, M., Sands, T., Jones, A., Hobson, J. & Robinson, S. 2013. Modified uvulopalatopharyngoplasty and coblation channeling of the tongue for obstructive sleep apnea: a multi-centre Australian trial. *J Clin Sleep Med*, 9, 117-24.
- Mador, M. J., Kufel, T., Magalang, U., Rajesh, S., Watwe, V. & Grant, B. 2005a. Prevalence of positional sleep apnea in patients undergoing polysomnography. *Chest*, 128, 2130-7.
- Mador, M. J., Kufel, T. J., Magalang, U. J., Rajesh, S. K., Watwe, V. & Grant, B. J. 2005b. Prevalence of positional sleep apnea in patients undergoing polysomnography. *Chest*, 128, 2130-7.
- Malhotra, A., Trinder, J., Fogel, R., Stanchina, M., Patel, S. R., Schory, K., Kleverlaan, D. & White, D. P. 2004. Postural effects on pharyngeal protective reflex mechanisms. *Sleep*, 27, 1105-12.
- Marques, M., Genta, P. R., Sands, S. A., Azarbazin, A., De Melo, C., Taranto-Montemurro, L., White, D. P. & Wellman, A. 2017. Effect of Sleeping Position on Upper Airway Patency in Obstructive Sleep Apnea is Determined by the Pharyngeal Structure Causing Collapse. *Sleep*.
- Marshall, N. S., Bartlett, D. J., Matharu, K. S., Williams, A. & Grunstein, R. R. 2007. Prevalence of treatment choices for snoring and sleep apnea in an Australian population. *Journal of clinical sleep medicine : JCSM : official publication of the American Academy of Sleep Medicine,* 3, 695-699.
- Marshall, N. S., Wong, K. K., Liu, P. Y., Cullen, S. R., Knuiman, M. W. & Grunstein, R. R. 2008. Sleep apnea as an independent risk factor for all-cause mortality: the Busselton Health Study. *Sleep*, 31, 1079-85.
- Martin, S. S., Qasim, A. & Reilly, M. P. 2008. Leptin Resistance. *Journal of the American College of Cardiology*, 52, 1201.
- Maspero, C., Giannini, L., Galbiati, G., Rosso, G. & Farronato, G. 2015. Obstructive sleep apnea syndrome: a literature review. *Minerva Stomatol,* 64, 97-109.
- Mcardle, N., Kingshott, R., Engleman, H. M., Mackay, T. W. & Douglas, N. J. 2001. Partners of patients with sleep apnoea/hypopnoea syndrome: effect of CPAP treatment on sleep quality and quality of life. *Thorax*, 56, 513-8.

- Mcevoy, R. D., Antic, N. A., Heeley, E., Luo, Y., Ou, Q., Zhang, X., Mediano, O., Chen, R., Drager, L. F., Liu, Z., Chen, G., Du, B., Mcardle, N., Mukherjee, S., Tripathi, M., Billot, L., Li, Q., Lorenzi-Filho, G., Barbe, F., Redline, S., Wang, J., Arima, H., Neal, B., White, D. P., Grunstein, R. R., Zhong, N., Anderson, C. S., Investigators, S. & Coordinators 2016. CPAP for Prevention of Cardiovascular Events in Obstructive Sleep Apnea. *N Engl J Med*, 375, 919-31.
- Mcevoy, R. D., Sharp, D. J. & Thornton, A. T. 1986. The effects of posture on obstructive sleep apnea. *Am Rev Respir Dis,* 133, 662-6.
- Miyamoto, K., Ozbek, M. M., Lowe, A. A. & Fleetham, J. A. 1997. Effect of body position on tongue posture in awake patients with obstructive sleep apnoea. *Thorax*, 52, 255-9.
- Mo, J.-H., Lee, C. H., Rhee, C.-S., Yoon, I.-Y. & Kim, J.-W. 2011. Positional Dependency in Asian Patients With Obstructive Sleep Apnea and Its Implication for Hypertension. *Archives of Otolaryngology–Head & Neck Surgery*, 137, 786-790.
- Mok, Y., Tan, A., Hsu, P. P., Seow, A., Chan, Y. H., Wong, H. S., Poh, Y. & Wong, K.
 K. H. 2020. Comparing treatment effects of a convenient vibratory positional device to CPAP in positional OSA: a crossover randomised controlled trial. *Thorax*, 75, 331.
- Mollayeva, T., Thurairajah, P., Burton, K., Mollayeva, S., Shapiro, C. M. & Colantonio, A. 2016. The Pittsburgh sleep quality index as a screening tool for sleep dysfunction in clinical and non-clinical samples: A systematic review and metaanalysis. *Sleep Med Rev*, 25, 52-73.
- Neill, A. M., Angus, S. M., Sajkov, D. & Mcevoy, R. D. 1997. Effects of sleep posture on upper airway stability in patients with obstructive sleep apnea. *Am J Respir Crit Care Med*, 155, 199-204.
- Nelson, L. M. & Barrera, J. E. 2007. High energy single session radiofrequency tongue treatment in obstructive sleep apnea surgery. *Otolaryngol Head Neck Surg*, 137, 883-8.
- Oksenberg, A. & Gadoth, N. 2019. Positional Therapy is a Valuable Treatment Alternative for Obstructive Sleep Apnea. *Journal of translational internal medicine* [Online], 7.
- Oksenberg, A., Goizman, V., Eitan, E., Nasser, K., Gadoth, N. & Leppänen, T. 2019. Obstructive Sleep Apnea: Do Positional Patients Become Nonpositional Patients With Time? *The Laryngoscope*, 130, 2263-2268.
- Oksenberg, A., Silverberg, D., Offenbach, D. & Arons, E. 2006. Positional therapy for obstructive sleep apnea patients: A 6-month follow-up study. *Laryngoscope*, 116, 1995-2000.

- Oksenberg, A. & Silverberg, D. S. 1998. The effect of body posture on sleep-related breathing disorders: facts and therapeutic implications. *Sleep Med Rev,* 2, 139-62.
- Oksenberg, A. & Silverberg, D. S. 2009. Avoiding the supine posture during sleep for patients with mild obstructive sleep apnea. *Am J Respir Crit Care Med*, 180, 101; author reply 101-2.
- Omobomi, O. & Quan, S. F. 2018. Positional therapy in the management of positional obstructive sleep apnea-a review of the current literature. *Sleep Breath*, 22, 297-304.
- Otsuka, R., Ono, T., Ishiwata, Y. & Kuroda, T. 2000. Respiratory-Related Genioglossus Electromyographic Activity in Response to Head Rotation and Changes in Body Position. *The Angle Orthodontist*, 70, 63-69.
- Pare, D. & Llinas, R. 1995. Conscious and pre-conscious processes as seen from the standpoint of sleep-waking cycle neurophysiology. *Neuropsychologia*, 33, 1155-68.
- Parish, J. M. & Lyng, P. J. 2003. Quality of life in bed partners of patients with obstructive sleep apnea or hypopnea after treatment with continuous positive airway pressure. *Chest*, 124, 942-7.
- Parish, J. M. & Somers, V. K. 2004. Obstructive sleep apnea and cardiovascular disease. *Mayo Clin Proc*, 79, 1036-46.
- Patel, S., Kon, S. S. C., Nolan, C. M., Barker, R. E., Simonds, A. K., Morrell, M. J. & Man, W. D. 2018. The Epworth Sleepiness Scale: Minimum Clinically Important Difference in Obstructive Sleep Apnea. *Am J Respir Crit Care Med*, 197, 961-963.
- Pelausa, E. O. & Tarshis, L. M. 1989. Surgery for snoring. *The Laryngoscope*, 99, 1006-1010.
- Peppard, P. E., Young, T., Palta, M., Dempsey, J. & Skatrud, J. 2000. Longitudinal study of moderate weight change and sleep-disordered breathing. *Jama*, 284, 3015-21.
- Popescu, G., Latham, M., Allgar, V. & Elliott, M. W. 2001. Continuous positive airway pressure for sleep apnoea/hypopnoea syndrome: usefulness of a 2 week trial to identify factors associated with long term use. *Thorax*, 56, 727.
- Poyares, D., Guilleminault, C., Rosa, A., Ohayon, M. & Koester, U. 2002. Arousal, EEG spectral power and pulse transit time in UARS and mild OSAS subjects. *Clinical Neurophysiology*, 113, 1598-1606.

Prisk, G. K. 1998. Sleep and respiration in microgravity. *Neurosci News*, 1, 39-45.

- Puig, F., Rico, F., Almendros, I., Montserrat, J. M., Navajas, D. & Farre, R. 2005. Vibration enhances interleukin-8 release in a cell model of snoring-induced airway inflammation. *Sleep*, 28, 1312-6.
- Ratnavadivel, R., Chau, N., Stadler, D., Yeo, A., Mcevoy, R. D. & Catcheside, P. G. 2009. Marked reduction in obstructive sleep apnea severity in slow wave sleep. *J Clin Sleep Med*, 5, 519-24.
- Ren, A., Dong, B., Lv, X., Zhu, T., Hu, F. & Yang, X. A non-contact sleep posture sensing strategy considering three dimensional human body models. 2016 2nd IEEE International Conference on Computer and Communications (ICCC), 14-17 Oct. 2016 2016. 414-417.
- Richardson, J., Iezzi, A., Khan, M. A. & Maxwell, A. 2014. Validity and reliability of the Assessment of Quality of Life (AQoL)-8D multi-attribute utility instrument. *Patient*, 7, 85-96.
- Roche, J., Isacco, L., Masurier, J., Pereira, B., Mougin, F., Chaput, J. P. & Thivel, D. 2020. Are obstructive sleep apnea and sleep improved in response to multidisciplinary weight loss interventions in youth with obesity? A systematic review and meta-analysis. *Int J Obes (Lond)*.
- Rotenberg, B. W., Murariu, D. & Pang, K. P. 2016. Trends in CPAP adherence over twenty years of data collection: a flattened curve. *J Otolaryngol Head Neck Surg*, 45, 43.
- Roth, T., Walsh, J. K., Krystal, A., Wessel, T. & Roehrs, T. A. 2005. An evaluation of the efficacy and safety of eszopiclone over 12 months in patients with chronic primary insomnia. *Sleep Med*, 6, 487-95.
- Ruehland, W. R., O'donoghue, F. J., Pierce, R. J., Thornton, A. T., Singh, P., Copland, J. M., Stevens, B. & Rochford, P. D. 2011. The 2007 AASM recommendations for EEG electrode placement in polysomnography: impact on sleep and cortical arousal scoring. *Sleep*, 34, 73-81.
- Saghaei, M. & Saghaei, S. 2011. Implementation of an open-source customizable minimization program for allocation of patients to parallel groups in clinical trials. *Journal of Biomedical Science and Engineering*, *4*, 734-739.
- Sakamoto, T., Nakazawa, Y., Hashizume, Y., Tsutsumi, Y., Mizuma, H., Hirano, T., Mukai, M. & Kotorii, T. 1995. Effects of acetazolamide on the sleep apnea syndrome and its therapeutic mechanism. *Psychiatry Clin Neurosci*, 49, 59-64.
- Salejee, I., Tarasiuk, A., Reder, I. & Scharf, S. M. 1993. Chronic upper airway obstruction produces right but not left ventricular hypertrophy in rats. *Am Rev Respir Dis*, 148, 1346-50.

- Schmickl, C. N., Owens, R. L., Orr, J. E., Edwards, B. A. & Malhotra, A. 2020. Side effects of acetazolamide: a systematic review and meta-analysis assessing overall risk and dose dependence. *BMJ Open Respir Res,* 7.
- Senaratna, C. V., Perret, J. L., Lodge, C. J., Lowe, A. J., Campbell, B. E., Matheson, M. C., Hamilton, G. S. & Dharmage, S. C. 2017. Prevalence of obstructive sleep apnea in the general population: A systematic review. *Sleep Med Rev*, 34, 70-81.
- Skinner, M. A., Kingshott, R. N., Filsell, S. & Taylor, D. R. 2008. Efficacy of the 'tennis ball technique' versus nCPAP in the management of position-dependent obstructive sleep apnoea syndrome. *Respirology*, 13, 708-15.
- Smoots, J. R. & Pisani, R. J. 2004. Reduction in Body Weight Produces a Significant Decrease in the Apnea Hypopnea Index [AHI] in Patients with Obstructive Sleep Apnea [OSA]. *Chest*, 126, 782S.
- Srijithesh, P. R., Aghoram, R., Goel, A. & Dhanya, J. 2019. Positional therapy for obstructive sleep apnoea. *Cochrane Database of Systematic Reviews*.
- Stadler, D. L., Mcevoy, R. D., Bradley, J., Paul, D. & Catcheside, P. G. 2010. Changes in lung volume and diaphragm muscle activity at sleep onset in obese obstructive sleep apnea patients vs. healthy-weight controls. J Appl Physiol (1985), 109, 1027-36.
- Stadler, D. L., Mcevoy, R. D., Sprecher, K. E., Thomson, K. J., Ryan, M. K., Thompson, C. C. & Catcheside, P. G. 2009. Abdominal compression increases upper airway collapsibility during sleep in obese male obstructive sleep apnea patients. *Sleep*, 32, 1579-1587.
- Steffen, A., Sommer, J. U., Hofauer, B., Maurer, J. T., Hasselbacher, K. & Heiser, C. 2018. Outcome after one year of upper airway stimulation for obstructive sleep apnea in a multicenter German post-market study. *Laryngoscope*, 128, 509-515.
- Stein, P. K. & Pu, Y. 2012. Heart rate variability, sleep and sleep disorders. *Sleep Med Rev*, 16, 47-66.
- Stöberl, A. S., Schwarz, E. I., Haile, S. R., Turnbull, C. D., Rossi, V. A., Stradling, J. R. & Kohler, M. 2017. Night-to-night variability of obstructive sleep apnea. *Journal of Sleep Research*, 26, 782-788.
- Strollo, P. J., Jr., Soose, R. J., Maurer, J. T., De Vries, N., Cornelius, J., Froymovich, O., Hanson, R. D., Padhya, T. A., Steward, D. L., Gillespie, M. B., Woodson, B. T., Van De Heyning, P. H., Goetting, M. G., Vanderveken, O. M., Feldman, N., Knaack, L., Strohl, K. P. & Group, S. T. 2014. Upper-airway stimulation for obstructive sleep apnea. *N Engl J Med*, 370, 139-49.

- Sulivan, F. 2015. Hidden Health Crisis Costing America Billions. Mountain View, CA: The American Academy of Sleep Medicine.
- Sullivan, C. E., Issa, F. G., Berthon-Jones, M. & Eves, L. 1981. Reversal of obstructive sleep apnoea by continuous positive airway pressure applied through the nares. *Lancet*, 1, 862-5.
- Supriyatno, B., Said, M., Hermani, B., Sjarir, D. R. & Sastroasmoro, S. 2010. Risk factors of obstructive sleep apnea syndrome in obese early adolescents: a prediction model using scoring system. *Acta Med Indones*, 42, 152-7.
- Sutherland, K., Takaya, H., Qian, J., Petocz, P., Ng, A. T. & Cistulli, P. A. 2015. Oral appliance treatment response and polysomnographic phenotypes of obstructive sleep apnea. *Journal of Clinical Sleep Medicine*, 11, 861-868.
- Tanahashi, T., Nagano, J., Yamaguchi, Y., Kubo, C. & Sudo, N. 2012. Factors that predict adherence to continuous positive airway pressure treatment in obstructive sleep apnea patients: A prospective study in Japan. *Sleep and Biological Rhythms*, 10, 126-135.
- Taranto-Montemurro, L., Messineo, L., Sands, S. A., Azarbarzin, A., Marques, M., Edwards, B. A., Eckert, D. J., White, D. P. & Wellman, A. 2019. The Combination of Atomoxetine and Oxybutynin Greatly Reduces Obstructive Sleep Apnea Severity. A Randomized, Placebo-controlled, Double-Blind Crossover Trial. Am J Respir Crit Care Med, 199, 1267-1276.
- Tarasiuk, A., Reznor, G., Greenberg-Dotan, S. & Reuveni, H. 2012. Financial incentive increases CPAP acceptance in patients from low socioeconomic background. *PLoS One*, *7*, e33178.
- Tassi, P. & Muzet, A. 2001. Defining the states of consciousness. *Neurosci Biobehav Rev,* 25, 175-91.
- Tregear, S., Reston, J., Schoelles, K. & Phillips, B. 2009. Obstructive sleep apnea and risk of motor vehicle crash: systematic review and meta-analysis. *J Clin Sleep Med*, *5*, 573-81.
- Ulrich, S., Keusch, S., Hildenbrand, F. F., Lo Cascio, C., Huber, L. C., Tanner, F. C., Speich, R. & Bloch, K. E. 2015. Effect of nocturnal oxygen and acetazolamide on exercise performance in patients with pre-capillary pulmonary hypertension and sleep-disturbed breathing: randomized, double-blind, cross-over trial. *Eur Heart J*, 36, 615-23.
- Van De Graaff, W. B. 1991. Thoracic traction on the trachea: mechanisms and magnitude. *J Appl Physiol (1985)*, 70, 1328-36.
- Van Maanen, J. P. & De Vries, N. 2014. Long-term effectiveness and compliance of positional therapy with the sleep position trainer in the treatment of positional obstructive sleep apnea syndrome. *Sleep*, *37*, 1209-15.

- Van Maanen, J. P., Meester, K. A., Dun, L. N., Koutsourelakis, I., Witte, B. I., Laman, D. M., Hilgevoord, A. A. & De Vries, N. 2013. The sleep position trainer: a new treatment for positional obstructive sleep apnoea. *Sleep Breath*, 17, 771-9.
- Viner, S., Szalai, J. P. & Hoffstein, V. 1991. Are history and physical examination a good screening test for sleep apnea? *Ann Intern Med*, 115, 356-9.
- Walsh, J. H., Leigh, M. S., Paduch, A., Maddison, K. J., Armstrong, J. J., Sampson, D. D., Hillman, D. R. & Eastwood, P. R. 2008. Effect of body posture on pharyngeal shape and size in adults with and without obstructive sleep apnea. *Sleep*, 31, 1543-9.
- Wannamethee, S. G., Tchernova, J., Whincup, P., Lowe, G. D. O., Kelly, A., Rumley, A., Wallace, A. M. & Sattar, N. 2007. Plasma leptin: Associations with metabolic, inflammatory and haemostatic risk factors for cardiovascular disease. *Atherosclerosis*, 191, 418-426.
- Weaver, T. E. 2006. Adherence to positive airway pressure therapy. *Curr Opin Pulm Med*, 12, 409-13.
- Weaver, T. E. & Grunstein, R. R. 2008. Adherence to continuous positive airway pressure therapy: the challenge to effective treatment. *Proceedings of the American Thoracic Society*, 5, 173-178.
- Weaver, T. E., Maislin, G., Dinges, D. F., Bloxham, T., George, C. F., Greenberg, H., Kader, G., Mahowald, M., Younger, J. & Pack, A. I. 2007. Relationship between hours of CPAP use and achieving normal levels of sleepiness and daily functioning. *Sleep*, 30, 711-9.
- Weaver, T. E., Maislin, G., Dinges, D. F., Younger, J., Cantor, C., Mccloskey, S. & Pack, A. I. 2003. Self-Efficacy in Sleep Apnea: Instrument Development and Patient Perceptions of Obstructive Sleep Apnea Risk, Treatment Benefit, and Volition to Use Continuous Positive Airway Pressure. *Sleep*, 26, 727-732.
- Wellman, A., Genta, P. R., Owens, R. L., Edwards, B. A., Sands, S. A., Loring, S. H., White, D. P., Jackson, A. C., Pedersen, O. F. & Butler, J. P. 2014. Test of the Starling resistor model in the human upper airway during sleep. *Journal of applied physiology (Bethesda, Md. : 1985),* 117, 1478-1485.
- White, D. P. & Younes, M. K. 2012. Obstructive sleep apnea. *Compr Physiol*, 2, 2541-94.
- Wilkinson, V., Malhotra, A., Nicholas, C. L., Worsnop, C., Jordan, A. S., Butler, J. E., Saboisky, J. P., Gandevia, S. C., White, D. P. & Trinder, J. 2008. Discharge patterns of human genioglossus motor units during sleep onset. *Sleep*, 31, 525-33.

- Wilson, K., Stoohs, R. A., Mulrooney, T. F., Johnson, L. J., Guilleminault, C. & Huang,
 Z. 1999. The snoring spectrum: acoustic assessment of snoring sound intensity
 in 1,139 individuals undergoing polysomnography. *Chest*, 115, 762-70.
- Wolkove, N., Baltzan, M., Kamel, H., Dabrusin, R. & Palayew, M. 2008. Long-term compliance with continuous positive airway pressure in patients with obstructive sleep apnea. *Can Respir J*, 15, 365-9.
- Yau, K. W., Wang, K. & Lee, A. H. 2003. Zero-Inflated Negative Binomial Mixed Regression Modeling of Over-Dispersed Count Data with Extra Zeros. *Biometrical Journal*, 45, 437-452.
- Young, T., Evans, L., Finn, L. & Palta, M. 1997. Estimation of the Clinically Diagnosed Proportion of Sleep Apnea Syndrome in Middle-aged Men and Women. *Sleep*, 20, 705-706.
- Young, T., Palta, M., Dempsey, J., Skatrud, J., Weber, S. & Badr, S. 1993. The occurrence of sleep-disordered breathing among middle-aged adults. *N Engl J Med*, 328, 1230-5.
- Young, T., Peppard, P. E. & Gottlieb, D. J. 2002. Epidemiology of obstructive sleep apnea: a population health perspective. *Am J Respir Crit Care Med*, 165, 1217-39.
- Zinchuk, A. V., Gentry, M. J., Concato, J. & Yaggi, H. K. 2016. Phenotypes in obstructive sleep apnea: A definition, examples and evolution of approaches. *Sleep Med Rev*.
- Zuberi, N. A., Rekab, K. & Nguyen, H. V. 2004. Sleep apnea avoidance pillow effects on obstructive sleep apnea syndrome and snoring. *Sleep Breath*, 8, 201-7.